## Constant Proportion Portfolio Insurance and Related Topics WITH EMPIRICAL STUDY

A Dissertation presented to the Faculty of the Graduate School at the University of Missouri

In Partial Fulfillment of the Requirements for the Degree Doctor of Philosophy

by

Wang, Mingming Professor Allanus H. Tsoi, Dissertation Supervisor MAY 2012

©Copyright by Mingming Wang 2012 All Rights Reserved The undersigned, appointed by the Dean of the Graduate School, have examined the dissertation entitled:

## Constant Proportion Portfolio Insurance and Related Topics WITH EMPIRICAL STUDY

presented by Wang, Mingming,

a candidate for the degree of Doctor of Philosophy and hereby certify that, in their opinion, it is worthy of acceptance.

Prof. Allanus Tsoi

Prof. Carmen Chicone

Prof. Stephen Montgomery-Smith

Prof. Michael Pang

Prof. X.H. Wang

### ACKNOWLEDGMENTS

Looking back, I am surprised and very grateful for all I have received from the university throughout the past four years. It has certainly shaped me as a person and has led me to where I am now. All these years of PhD studies are full of such gifts. Firstly, I wish to express my sincere gratitude to my advisor Professor Allanus H. Tsoi for his help and guidance during the past four years.

Secondly, am I thankful the financial support from department of mathematics.

I am also grateful to my doctoral committee members Professor Carmen Chicone, Professor Michael Pang, Professor Stephen Montgomery-Smith and Professor X.H. Wang for their interest and help.

## TABLE OF CONTENTS

A	CKN	OWLI	EDGMENTS	ii
$\mathbf{LI}$	ST C	OF FIC	GURES	vii
A	BSTI	RACT		viii
C	HAP	TER		
1	Intr	oducti	on	1
<b>2</b>	CPI	PI and	EPPI in Diffusion model	7
	2.1	CPPI	in the Black-Scholes model	7
		2.1.1	The continuous trading time case	7
		2.1.2	The discrete trading time case	13
	2.2	EPPI	in Black-Scholes model	18
		2.2.1	The discrete trading time case	18
		2.2.2	The continuous trading time case	22
	2.3	CPPI	in GARCH model	24
		2.3.1	The Continuous trading time cases	24
		2.3.2	The Discrete trading time case $GARCH(1,1)$ model	28
	2.4	EPPI	in $GARCH(1,1)$ model	30
3	CPI tinu	PI in t ious .	he Jump-diffusion model when the trading time is con-	35
	3.1	Jump-	diffusion model	35
		3.1.1	Set up the model:	35
		3.1.2	Two special Jump-diffusion models	37
		3.1.3	Martingale Measure	42

	3.2	The C	PPI strategies	45
		3.2.1	The constant multiple case	45
		3.2.2	The case when the multiple is a function of time	51
	3.3	The C	PPI portfolio as a hedging tool	53
		3.3.1	PIDE Approach	54
		3.3.2	Fourier Transformation Approach	61
		3.3.3	Martingale Approach	62
	3.4	Mean-	variance Hedging	71
		3.4.1	Introduction	71
		3.4.2	Our Problem	73
4	Gap	o risks		77
	4.1	Introd	uction $\ldots$	77
	4.2	Gap ri	sk Measure for CPPI strategies in Jump-diffusion model	78
		4.2.1	Probability of Loss	78
		4.2.2	Expected Loss	80
		4.2.3	Loss Distribution	84
	4.3	Condit diffusio	tional Floor and Conditional Multiple of CPPI in the Jump- on Model	86
		4.3.1	Introduction	86
		4.3.2	Probability of Loss	87
		4.3.3	Expected Loss	88
		4.3.4	Loss Distribution	89
		4.3.5	Conclusion	90
<b>5</b>	CPI	PI in tl	he jump-diffusion model when the trading time is discrete	91
	5.1	Introd	uction $\ldots$	91

	5.2	The st	prategy	92
	5.3	Measu the dis	are the Gap risk for CPPI strategies in the jump-diffusion model- screte time case	96
		5.3.1	Probability of Loss	96
		5.3.2	Expected Loss	99
		5.3.3	Loss Distribution	102
		5.3.4	Conclusion	104
	5.4	Condi diffusi	tional Floor and Conditional Multiple of CPPI under Jump- on Model in Discrete Trading Time	105
		5.4.1	Introduction	105
		5.4.2	Probability of Loss	105
		5.4.3	Expected Loss	107
		5.4.4	Loss Distribution	108
	5.5	Conve	rgence	109
6	Sto	chastic	and dynamic floors	111
	6.1	Introd	uction	111
	6.2	When percen	the floor equals to the maximum of its past value and a given stage of the portfolio value	112
		6.2.1	Discrete-time case with fixed multiple	112
		6.2.2	Continuous-time case with a fixed multiple	117
		6.2.3	Capped CPPI	120
	6.3	CPPI	with a floor indexed on a given portfolio performance $\ldots$ .	122
		6.3.1	Discrete-time with a fixed multiple	122
		6.3.2	Continuous-time case	125
	6.4	CPPI	with a floor indexed on the exposition variance $\ldots$	128
		6.4.1	The "Ratchet" CPPI	128

		6.4.2 CPPI with margin	130
7	CP	PI in the Fractional Brownian Markets	134
	7.1	Fractional Brownian Markets	134
	7.2	CPPI in the Fractional Black-Scholes market	137
	7.3	CPPI Option	140
	7.4	PDE Approach	141
8	CP	PI in Fractional Brownian Markets with Jumps	145
	0.1		
	8.1	Fractional Brownian Markets with Jumps	145
	8.1	Fractional Brownian Markets with Jumps	145 $153$
	8.1	Fractional Brownian Markets with Jumps       8.1.1         Esscher transform       8.1.1         CPPI in fractional Black-Scholes market with jumps       8.1.1	145 153 154
B	8.1 8.2 IBLI	Fractional Brownian Markets with Jumps       8.1.1         Esscher transform       8.1.1         CPPI in fractional Black-Scholes market with jumps       8.1.1         OGRAPHY       9.1.1	<ol> <li>145</li> <li>153</li> <li>154</li> <li>160</li> </ol>

## LIST OF FIGURES

# Figure

# Page

2.1	EPPI vs CPPI in Black-Scholes Model	22
2.2	CPPI in the $GARCH(1,1)$ model $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	31
2.3	EPPI in GARCH	33
2.4	EPPI vs CPPI in GARCH	34

## ABSTRACT

The concept of Constant Proportion Portfolio Insurance (CPPI) in terms of jumpdiffusion, as well as the associated mean-variance hedging problem, has been studied. Three types of risk related to: the probability of loss, the expected loss, and the loss distribution are being analyzed. Both the discrete trading time case and the continuous trading time case have been studied. Next, CPPI with stochastic dynamic floors are being discussed. The concept of exponential proportion portfolio insurance is being introduced. Finally CPPI associated with the fractional Brownian market is being studied.

# Chapter 1

# Introduction

Constant Proportion Portfolio Insurance (CPPI) was introduced by [61] for equity instruments, and has been further analyzed by many scholars (such as [10]). An investor invests in a portfolio and wants to protect the portfolio value from falling below a pre-assigned value. The investor shifts his asset allocation over the investment period among a risk-free asset plus a collection of risky assets. The CPPI strategy is based on the dynamic portfolio allocation of two basic assets: a riskless asset (usually a treasury bill) and a risky asset (a stock index for example). This strategy relies crucially on the concept of a *cushion* C, which is defined as the difference between the *portfolio value* V and the *floor* F. This later one corresponds to a guaranteed amount at any time t of the management period [0, T]. The key assumption is that the amount e invested on the risky asset, called the *exposure*, is equal to the cushion multiplied by a fixed coefficient m, called the *multiple*. The floor and the multiple can be chosen according to the investor's risk tolerance.

#### In chapter 2

In this chapter, we introduce the background and concept of the CPPI and EPPI modeling by a diffusion process. In section 2.1, we consider the simplest CPPI and its background. In this case the risky asset model is the classical Black-Scholes model. Both continuous and discrete are considered. We introduce the concept of EPPI (Exponential Proportion Portfolio Insurance). In section 2.3, we consider the case when the stock model satisfies a GARCH model. We also consider both the discrete and continuous trading cases. In section 2.4, we consider the EPPI in GARCH.

#### In chapter 3

In this chapter, we discuss the CPPI-jump-diffusion model when the trading time is continuous. The jump-diffusion model was introduced and widely studied by [58] and [65].

Let  $Y_n > -1$  be the percentage of the size of *n*-th jump, and  $S_t$  be the process who represent the stock price at time *t*. Thus,  $S_{T_n} = S_{T_n}(1 + Y_n)$ . Between two jumps, we assume the risky asset model satisfies Black-Scholes. The number of jumps upper to time *t* is a Poisson processes  $N_t$  with intensity  $\lambda_t$ . Then our model becomes

$$S_{t} = S_{0} \exp\left[\int_{0}^{t} \left(\mu_{s} - \frac{\sigma_{s}^{2}}{2}\right) ds + \int_{0}^{t} \sigma_{s} dW_{s} + \sum_{n=1}^{N_{t}} \ln(1 + Y_{n})\right].$$

We usually assume  $\ln(1 + Y_n)$  is i.i.d. and has density function  $f_Q$ .

Our outline of this section is following.

In section 3.1, we set up the jump-diffusion model, calculate the density function and discuss the martingale measure. In section 3.2, we describe the CPPI strategy and then calculate the CPPI portfolio value, its expectation and variance. In section 3.3, we consider the CPPI portfolio as a hedging tool. [16] considers the situation in Black-scholes model. Our discussion is a generalization of it. Both the PDE/PIDE approach and the martingale approach are studied there. However, because of the introduction of the jump term in the model, the calculation is much more complex. In section 3.2 and subsection 3.3, both short-sell and negative exposure are allowed. In section 3.4, we consider the mean-variance hedging for a given contingent claim H. In our jump-diffusion model, the market is not complete and then H is not attainable. Thus, we consider the mean-variance hedging which is a kind of quadratic hedging. [67] is a review paper about quadratic hedging, we adopt the symbol and definition from it. We consider H as the function of portfolio value  $V_T$  and measure the risk in probability  $\mathbb{Q}$ . Our optimal problem is following

$$\min_{(Z_0,\vartheta)\in\mathbb{R}\times\Theta}\mathbb{E}^{\mathbb{Q}}\left(\tilde{H}-Z_0-\int_0^T\vartheta_u d\tilde{V}_u\right)^2.$$

We adopt the method in Chapter 10 in [18] and give the explicit form optimal solution of  $Z_0$  and  $\vartheta_t$ . In section 3.2, section 3.3 and section 3.4, both short-sell and negative exposure are allowed.

The main contribution of this chapter is in section 3.3 and section 3.4.

#### In chapter 4

In this chapter, we continue to discuss the CPPI in the jump-diffusion model.

In section 4.2, we discuss the **Gap risk** which is defined as the amount which represents how much that the portfolio value is below the floor at the terminal time. In this case, we do not allow short-sell and negative exposure. It is deduced that the gap happens only when the jump is negatively large enough such that  $1 + mY_i \leq 0$ . The probability of loss, expected loss and loss distribution are introduced to measure the gap. [17] has discuss the case in a more general model as

$$\frac{dS_t}{S_{t-}} = dZ_t,$$

where  $Z_t$  is a Levy process. Our jump-diffusion model could be treated as a special case. Thus, the conclusion in this section is a special case of [17]. However, we deduce more explicit expression as compare with [17], which is more appreciated in simulation. We will show that the conclusion of the probability of loss is consistant the conclusion in [17]; our conclusion for the expected loss is more explicit and the method is similar as [17]; our conclusion for the loss distribution is explicit and our method is different from [17]. In section 4.3, we consider the conditional multiple from the view of Probability of loss. Its idea is similar as the Value-at-Risk([27]). Four kinds of conditional floor are also discussed from the view of expected loss and loss distribution.

#### In chapter 5

In this chapter, we will study the jump-diffusion model when the trading time is discrete.

The risky asset model is similar as that in chapter 3 and 4.

In section 5.2, we calculate the CPPI portfolio value and its expectation and variance. Gap risks exist because the risky model has jumps and also the trading time is discrete. In section 5.3, we measure the gap risk with respect to three aspects: probability of loss, expected loss and loss distribution. We give out their explicit forms.

In section 5.4, we define the conditional multiples associated with the probability of loss, conditional floors associated with expected loss and loss distribution.

In section 5.5, we prove that as the interval length of the trading times tends to zero, the CPPI strategies in discrete trading time will convergent to the CPPI strategies in continuous time.

#### In chapter 6

In this chapter, we investigate several types of stochastic floors and dynamic floors. In [59], they have considered the cases of diffusion models without jumps. Here we generalize it to the jump-diffusion case.

In section 6.2, we consider the case when the stochastic floor is equal to the maximum of its past value and a given percentage of the portfolio value. The idea is that when the portfolio value is large enough, the level of the floor rises. Both the continuous trading and discrete trading time cases will be analyzed. We will calculate the distribution of the time when the floor is increased.

In section 6.3, we consider the case when stochastic floor is indexed with respect to the given portfolio performance. The idea is similar as section 6.2. Both the continuous trading and discrete trading time cases will also be analyzed. We will also calculate the distribution of the first-time-change of the floor.

In section 6.4, we will deal with the Ratchet and Margin CPPI strategies with time change related to the exposition variance. We will show in discrete trading time case, the Ratchet CPPI is equivalent to the stochastic floor index on the given portfolio performance. The idea of CPPI with margin is that when the floor is close to the portfolio value, the exposure will be very small and we will reduce the floor. We will discuss the distribution of the first-change-time of the floor in the continuous trading time case.

#### In chapter 7

In this chapter, we consider the CPPI in a fractional Brownian Market.

Fractional Black-Scholes market was introduced by [35] where they utilize the wick product and thus redefined many market concepts such as **portfolio**, **value process**, **self-financing**, **admissible**, **arbitrage** and **complete**. In Section 7.1, we adopt the fractional Brownian markets and new markets concepts as in [35]. Under this new market, we calculate the CPPI portfolio value, its expectation and variance in Section 7.2. In Section 7.3, we calculate the CPPI option. Moreover, we consider the associate hedging problem by PDE approach in Section 7.4.

#### In Chapter 8

In this chapter, we consider the CPPI in a fractional Brownian Markets with jumps. This chapter could be treated as an extension of Chapter 7. In Section 8.1, we setup the fractional Brownian markets with jumps and redifined many market concepts as in Chapter 7. We also deduce the Girsanov Formula in fractional Black-Scholes model with jumps. In Section 8.2, we calculate the CPPI portfolio value, its expectation and variance.

# Chapter 2

# **CPPI** and **EPPI** in Diffusion model

### 2.1 CPPI in the Black-Scholes model

### 2.1.1 The continuous trading time case

The CPPI (Constant Proportion Portfolio Insurance) strategy is based on a dynamic portfolio allocation on two basic assets: a riskless asset (usually a treasury bill) and a risky asset (a stock index for example).

This strategy depends crucially on the cushion C, which is defined as the difference between the portfolio value V and the floor F. This later one corresponds to a guaranteed amount at any time t of the management period [0, T]. The key assumption is that the amount e invested on the risky asset, called the *exposure*, is equal to the cushion multiplied by a fixed coefficient m, called the *multiple*. The floor and the multiple can be chosen according to the investors risk tolerance. The risk-aversion investor will choose a small multiple or/and a high floor and vice versa. The higher the multiple, the more the investor will benefit from increases in stock prices. Nevertheless, the higher the multiple, the higher the risk that the portfolio value becomes smaller than the floor if the risky asset price drops suddenly. As the cushion value is approximately equal to zero, exposure is near zero too. In the continuous-time case, if the asset dynamics has no jump, then the portfolio value does not fall below the floor. We define:

> interest rate: r; time: t; time period: [0,T]; floor: F; floor at time t:  $F_t$ ; portfolio value: V; portfolio value at time t:  $V_t$ ; cushion C; cushion at time t:  $C_t$ ; multiple m; exposure e; exposure at time t:  $e_t$ ; riskless asset at time t:  $B_t$ .

where

$$C = V - F \qquad e = mC.$$

Let  $(\Omega, \mathfrak{F}, \mathfrak{F}_t, \mathbb{P})$  be a probability space satisfying the "usual assumption". In the simple CPPI continuous time case we assume that the risky asset satisfies the Black-Scholes model, i.e.

$$dS_t = S_t(\mu dt + \sigma dW_t), \quad S_0 = s, \tag{2.1}$$

where  $V_t = mC_t + (V_t - mC_t)$ .

Let the interest rate be r and floor at time t be  $F_t = F_0 e^{rt} = F_T e^{-r(T-t)}$ . We denote  $F_T = G$ .

Here are a list of their relation:

$$C_t = V_t - F_t;$$
  

$$e_t = mC_t;$$
  

$$B_t = V_t - e_t.$$

**Proposition 2.1.** The portfolio value of CPPI under the Black-Scholes model in continuous time trading is

$$V_t = (V_0 - Ge^{-rT}) \exp\left\{ (r + m(\mu - r))t - \frac{m^2 \sigma^2 t}{2} + m\sigma W_t \right\} + G \times \exp\{-r(T - t)\}$$
(2.2)

where  $G = F_T$ .

*Proof.* We have

$$V_t = mC_t + (V_t - mC_t)$$
  
=  $V_t \left( \frac{mC_t}{V_t} + \left( 1 - \frac{mC_t}{V_t} \right) \right),$ 

and by the assumption of self-financing, we have

$$dV_t = V_t \left(\frac{mC_t}{V_t}\frac{dS_t}{S_t} + \left(1 - \frac{mC_t}{V_t}\right)\frac{dB_t}{B_t}\right),$$

thus

$$dC_t = d(V_t - F_t)$$

$$= V_t \left( \frac{mC_t}{V_t} \frac{dS_t}{S_t} + \left( 1 - \frac{mC_t}{V_t} \right) \frac{dB_t}{B_t} \right) - F_t \frac{dB_t}{B_t}$$

$$= C_t \left( \frac{mdS_t}{S_t} - (m-1)rdt \right)$$

$$= C_t (m(\mu dt + \sigma dW_t) - (m-1)rdt)$$
9

$$= C_t((r+m(\mu-r))dt+m\sigma dW_t).$$

Then

$$C_t = C_0 \exp\left\{ (r + m(\mu - r))t - \frac{m^2 \sigma^2 t}{2} + m\sigma W_t \right\},$$

therefore, we have

$$V_{t} = C_{t} + F_{t}$$

$$= C_{0} \exp\left\{ (r + m(\mu - r))t - \frac{m^{2}\sigma^{2}t}{2} + m\sigma W_{t} \right\} + G \times \exp\left\{ -r(T - t) \right\}$$

$$= \left( V_{0} - Ge^{-r(T - t)} \right) \exp\left\{ (r + m(\mu - r))t - \frac{m^{2}\sigma^{2}t}{2} + m\sigma W_{t} \right\} + G \times \exp\{-rT\}.$$

The expectation and variance of the CPPI portfolio value are obviously two important values to describe the strategies.

We know that  $\exp\left(m\sigma W_t - \frac{1}{2}m^2\sigma^2 t\right)$  is an exponential martingale. Thus, we get the expectation of the CPPI portfolio value in the following proposition.

**Proposition 2.2.** The expectation of CPPI portfolio value under the Black-Scholes model in continuous time trading is

$$Ge^{-rT} + (V_0 - Ge^{-rT}) \exp\{(r + m(\mu - r))t\}$$

Proof.

$$\mathbb{E}[V_t] = Ge^{-r(T-t)} + C_0 \exp\{(r+m(\mu-r))t\}\mathbb{E}\left[\exp\left(m\sigma W_t - \frac{1}{2}m^2\sigma^2 t\right)\right] \\ = Ge^{-r(T-t)} + (V_0 - Ge^{-rT})\exp\{(r+m(\mu-r))t\}.$$

In order to calculate the variance, we will use the following lemma.

Lemma 2.3. Let  $h_t = \exp\left(m\sigma W_t - \frac{1}{2}m^2\sigma^2 t\right)$ , then  $\mathbb{E}[h_t] = 1$  and  $Var(h_t) = \exp(b^2 t) - 1$ .

*Proof.* By Ito formula,  $dh_t = bh_t dW_t$ , then

$$h_t - h_0 = \int_0^t bh_s dW_s,$$

then  $h_t$  is a martingale and then  $\mathbb{E}[h_t] = \mathbb{E}[h_0] = 1$ . We have

$$\begin{aligned} \operatorname{Var}(h_t) &= \mathbb{E}(h_t - \mathbb{E}(h_t))^2 = \mathbb{E}(h_t - h_0)^2 \\ &= \mathbb{E}\left(\int_0^t bh_s dW_s\right)^2 = \mathbb{E}\left(\int_0^t b^2 h_s ds\right) \\ &= b^2 \left(\int_0^t \mathbb{E}\left(h_s^2\right) ds\right) = b^2 \left(\int_0^t \mathbb{E}\left(\exp\left(2m\sigma W_t - m^2\sigma^2 t\right)\right) ds\right) \\ &= b^2 \left(\int_0^t \exp\left(b^2 s\right) \mathbb{E}\left(\exp\left(2m\sigma W_t - \frac{(2b)^2}{2}m^2\sigma^2 t\right)\right) ds\right) \\ &= b^2 \left(\int_0^t \exp\left(b^2 s\right) ds\right) = \exp\left(b^2 t\right) - 1. \end{aligned}$$

Using the above lemma, we could calculate the variance of the CPPI portfolio value in the following proposition. (Referent [16].)

**Proposition 2.4.** The variance of the CPPI portfolio value under the Black-Scholes model in continuous time trading is

$$\left(V_0 - Ge^{-rT}\right)^2 \exp(2(r + m(\mu - r)t) \left(\exp\left(m^2 \sigma^2 t\right) - 1\right)$$

Proof.

$$\operatorname{Var}[V_t] = \operatorname{Var}[C_t]$$
 11

$$= C_0^2 \exp(2(r+m(\mu-r)t) \operatorname{Var}\left[\exp\left(m\sigma W_t - \frac{1}{2}m^2\sigma^2 t\right)\right] \\ = C_0^2 \exp(2(r+m(\mu-r)t) \operatorname{Var}[h_t] \\ = \left(V_0 - Ge^{-rT}\right)^2 \exp(2(r+m(\mu-r)t)\left(\exp\left(m^2\sigma^2 t\right) - 1\right).$$

It is interesting at this point to wonder how the leverage regime modifies the return/risk profile of the product. As our intuition suggests, an increase in the gearing constant (multiple) which determines the leverage regime amplifies heavily the volatility.

**Proposition 2.5.** The expected portfolio value and the variance of the CPPI portfolio's value, increase with the multiple m. In particular it is true that for any  $t \in [0, T]$ ,

$$\lim_{m \to \infty} \mathbb{E}[V_t] = +\infty;$$
$$\lim_{m \to \infty} Var[V_t] = +\infty;$$

and

$$\lim_{m \to \infty} \frac{\mathbb{E}[V_t]}{Var[V_t]} = 0,$$

with an order of  $o\left(\exp((r+m(\mu-r)t)\frac{1}{\exp(m^2\sigma^2t)-1}\right)$ .

*Proof.* The first two equations is obviously. and for the third one

$$\lim_{m \to \infty} \frac{\mathbb{E}[V_t]}{\operatorname{Var}[V_t]} \sim \frac{\exp\{(r+m(\mu-r))t}{\exp(2(r+m(\mu-r)t)(\exp{(m^2\sigma^2 t)}-1))}$$
$$\sim \frac{1}{\exp((r+m(\mu-r)t)(\exp{(m^2\sigma^2 t)}-1))}$$
$$\sim 0. \quad (m \to \infty)$$
$$12$$

The following proposition shows there is no fallen risk for the continuous trading time CPPI defined on the continuous model.

**Proposition 2.6.** Let the risky asset model  $S_t$  be  $\mathbb{P}$  almost sure continuous and the trading time be continuous. If the CPPI defined on this model, then the portfolio value  $V_t$  is almost sure greater than the floor  $F_t$ .

*Proof.* In the proof of Proposition 2.1, we have got

$$dC_t = C_t \left(\frac{mdS_t}{S_t} - (m-1)rdt\right).$$

Then

$$\ln(C_t) - \ln(C_0) = m(\ln(S_t) - \ln(S_0)) - (m-1)rt.$$

We have

$$C_t = C_0 \exp\left(\ln\frac{S_t}{S_0} - (m-1)rt\right)$$

and it is  $\mathbb{P}$  almost sure positive.

### 2.1.2 The discrete trading time case

Here we continue to assume that our risky asset satisfies the Black-Scholes model. In addition, let  $\tau^N = \{t_0 = 0 < t_1 < t_2 < ... < t_n = T\}$  denote a sequence of equidistant refinements of the interval [0, T], where  $t_{k+1} - t_k = \frac{T}{n}$  for k = 0, ..., n - 1. We assume now that trading is restricted to the discrete set  $\tau^n$ . We have

$$C_{t_{k+1}} = C_{t_k} \left( m \frac{S_{t_{k+1}}}{S_{t_k}} - (m-1)e^{rT/n} \right),$$

then

$$C_T = C_{t_n} = C_0 \prod_{k=0}^{n-1} \left( m \frac{S_{t_{k+1}}}{S_{t_k}} - (m-1)e^{rT/n} \right),$$

thus

$$V_T = C_T + G.$$

Since  $\frac{S_{t_{k+1}}}{S_{t_k}}$ , k = 0.1.2...n-1 are mutually independent and also they have the identity distribution. Then we have

$$\mathbb{E}\left[\frac{S_{t_{k+1}}}{S_{t_k}}\right] = \mathbb{E}\left[\exp\left(\mu\frac{T}{n} + \sigma W_{T/n} - \frac{1}{2}\sigma^2\frac{T}{n}\right)\right] = \exp\left(\mu\frac{T}{n}\right);$$

and

$$\mathbb{E}\left[\frac{S_{t_{k+1}}}{S_{t_k}}\right]^2 = \mathbb{E}\left[\exp\left(2\mu\frac{T}{n} + 2\sigma W_{T/n} - \sigma^2\frac{T}{n}\right)\right]$$
$$= \mathbb{E}\left[\exp\left(2\mu\frac{T}{n} + \sigma^2\frac{T}{n} + 2\sigma W_{T/n} - \frac{1}{2}(2\sigma)^2\frac{T}{n}\right)\right]$$
$$= \exp\left(2\mu\frac{T}{n} + \sigma^2\frac{T}{n}\right).$$

In the discrete case, it is possible that  $V_{t_i} \leq F_{t_i}$  for some  $t_i$ . We generally allow the possibility of short-sell and negative cushion. However, this also means that the CPPI-insured portfolio would incur a loss.

**Proposition 2.7.** The expected terminal CPPI portfolio value under Black-Scholes model in the discrete trading is

$$\left(V_0 - Ge^{-rT}\right) \left(m \exp\left(\mu \frac{T}{n}\right) - (m-1)e^{rT/n}\right)^n + G.$$
(2.3)

Proof.

$$\mathbb{E}[V_T] = \mathbb{E}[C_T] + G = \mathbb{E}\left[C_0 \prod_{k=0}^{n-1} \left(m \frac{S_{t_{k+1}}}{S_{t_k}} - (m-1)e^{rT/n}\right)\right] + G$$
14

$$= C_0 \prod_{k=0}^{n-1} \left( m \mathbb{E} \left[ \frac{S_{t_{k+1}}}{S_{t_k}} \right] - (m-1)e^{rT/n} \right) + G$$
  
$$= C_0 \prod_{k=0}^{n-1} \left( m \exp \left( \mu \frac{T}{n} \right) - (m-1)e^{rT/n} \right) + G$$
  
$$= \left( V_0 - Ge^{-rT} \right) \left( m \exp \left( \mu \frac{T}{n} \right) - (m-1)e^{rT/n} \right)^n + G.$$

In order to calculate the variance of the terminal CPPI portfolio value, we need the following lemma.

**Lemma 2.8.** Let  $A_i$ , i=1,2,...n be independent random variables, then we have

$$Var\left[\prod_{k=1}^{n} A_{i}\right] = \prod_{k=1}^{n} \left(\mathbb{E}A_{i}^{2}\right) - \prod_{k=1}^{n} \left(\mathbb{E}A_{i}\right)^{2}.$$

*Proof.* We have

$$\operatorname{Var}\left[\prod_{k=1}^{n} A_{i}\right] = \mathbb{E}\left(\prod_{k=1}^{n} A_{i}\right)^{2} - \left(\mathbb{E}\prod_{k=1}^{n} A_{i}\right)^{2}$$
$$= \mathbb{E}\left(\prod_{k=1}^{n} A_{i}^{2}\right) - \left(\prod_{k=1}^{n} \mathbb{E}A_{i}\right)^{2}$$
$$= \prod_{k=1}^{n} \left(\mathbb{E}A_{i}^{2}\right) - \prod_{k=1}^{n} \left(\mathbb{E}A_{i}\right)^{2}.$$

By the above lemma, we could calculate the variance of the CPPI terminal portfolio value in the following proposition.

Proposition 2.9. The variance of the CPPI terminal portfolio value under Black-

Scholes model in the discrete trading is

$$(V_0 - Ge^{-rT})^2 \left( \left( m^2 \exp\left(2\mu \frac{T}{n} + \sigma^2 \frac{T}{n}\right) + (m-1)^2 e^{2rT/n} - 2m(m-1) \exp\left(\mu \frac{T}{n}\right) e^{rT/n} \right)^n - \left(m \exp\left(\mu \frac{T}{n}\right) - (m-1)e^{rT/n} \right)^{2n} \right).$$

Proof. Since

$$\mathbb{E}\left(m\frac{S_{t_{k+1}}}{S_{t_k}} - (m-1)e^{rT/n}\right)^2$$

$$= \mathbb{E}\left(m^2\left(\frac{S_{t_{k+1}}}{S_{t_k}}\right)^2 + (m-1)^2e^{2rT/n} - 2m(m-1)\frac{S_{t_{k+1}}}{S_{t_k}}e^{rT/n}\right)$$

$$= m^2\left(\mathbb{E}\frac{S_{t_{k+1}}}{S_{t_k}}\right)^2 + (m-1)^2e^{2rT/n} - 2m(m-1)\mathbb{E}\frac{S_{t_{k+1}}}{S_{t_k}}e^{rT/n}$$

$$= m^2\exp\left(2\mu\frac{T}{n} + \sigma^2\frac{T}{n}\right) + (m-1)^2e^{2rT/n} - 2m(m-1)\exp\left(\mu\frac{T}{n}\right)e^{rT/n},$$

we have

$$\begin{aligned} \operatorname{Var}[V_{T}] &= \operatorname{Var}[C_{T}] \\ &= \operatorname{Var}\left[C_{0}\prod_{k=0}^{n-1}\left(m\frac{S_{t_{k+1}}}{S_{t_{k}}} - (m-1)e^{rT/n}\right)\right] \\ &= C_{0}^{2}\left[\prod_{k=0}^{n-1}\mathbb{E}\left(m\frac{S_{t_{k+1}}}{S_{t_{k}}} - (m-1)e^{rT/n}\right)^{2} \\ &- \prod_{k=0}^{n-1}\left(\mathbb{E}\left(m\frac{S_{t_{k+1}}}{S_{t_{k}}} - (m-1)e^{rT/n}\right)\right)^{2}\right] \\ &= (V_{0} - Ge^{-rT})^{2}\left(\left(m^{2}\exp\left(2\mu\frac{T}{n} + \sigma^{2}\frac{T}{n}\right) + (m-1)^{2}e^{2rT/n} \\ &- 2m(m-1)\exp\left(\mu\frac{T}{n}\right)e^{rT/n}\right)^{n} - \left(m\exp\left(\mu\frac{T}{n}\right) - (m-1)e^{rT/n}\right)^{2n}\right). \end{aligned}$$

#### Probability of Loss

In the case of discrete-time trading, it is possible that the portfolio value falls below the floor. i.e.  $V_t \leq F_t$  which is equivalent to  $C_t \leq 0$ , happens only at time  $t_i$ . We call it the **Probability of Loss**.

There are two possible causes for gap risks. One is the existence of jumps in the risky asset model and the other is because of the trading time is not continuous. In this section, we consider the case when the gap risk happens at discontinuous trading time. In section 4.2, we will consider the presence of jumps and the trading time is continuous. In section 5.2, we will consider the co-existence of the above two situations.

**Proposition 2.10.** The probability of the CPPI portfolio value under Black-Scholes model in the discrete trading going below the floor taking happen is given by

$$\mathbb{P}[\exists t_i : V_{t_i} \le F_{t_i}] = 1 - \Psi^n \left( -\frac{1}{\sigma} \left( \sqrt{\frac{n}{T}} \ln \left( \frac{m-1}{m} \right) + \left( r - \mu + \frac{\sigma^2}{2} \right) \sqrt{\frac{T}{n}} \right) \right)$$
(2.4)

where

$$\Psi(y) = \int_{-\infty}^{y} \frac{1}{\sqrt{2\pi}} e^{-\frac{x^2}{2}} dx.$$

Proof.

$$\begin{split} & \mathbb{P}[\exists t_i : V_{t_i} \le F_{t_i}] = \mathbb{P}[\forall \quad t_i : C_{t_i} \le 0] \\ &= 1 - \mathbb{P}[\exists \quad t_i : C_{t_i} > 0] = 1 - \mathbb{P}\left[\bigcap_{i=1}^n \{C_{t_i} > 0\}\right] = 1 - \prod_{i=1}^n \mathbb{P}[\{C_{t_i} > 0\}] \\ &= 1 - \prod_{i=1}^n \mathbb{P}\left[\left\{m\frac{S_{t_i}}{S_{t_{i-1}}} - (m-1)e^{rT/n} > 0\right\}\right] \\ &= 1 - \prod_{i=1}^n \mathbb{P}\left[\left\{m\exp\left\{\left(\mu - \frac{\sigma^2}{2}\right)\frac{T}{n} + \sigma W_{\frac{T}{n}}\right\} - (m-1)e^{rT/n} > 0\right\}\right] \end{split}$$

$$= 1 - \left( \mathbb{P}\left[ \sigma W_{\frac{T}{n}} \ge \ln\left(\frac{m-1}{m}\right) + \left(r - \mu + \frac{\sigma^2}{2}\right) \frac{T}{n} \right] \right)^n$$
$$= 1 - \Psi^n \left( -\frac{1}{\sigma} \left( \sqrt{\frac{n}{T}} \ln\left(\frac{m-1}{m}\right) + \left(r - \mu + \frac{\sigma^2}{2}\right) \sqrt{\frac{T}{n}} \right) \right)$$

where

$$\Psi(y) = \int_{-\infty}^{y} \frac{1}{\sqrt{2\pi}} e^{-\frac{x^2}{2}} dx.$$

**Proposition 2.11.** The probability of the CPPI portfolio value under the Black-Scholes model in the discrete trading given by (2.4) is monotone increased function as the multiple m.

Proof. We have

$$m \uparrow \Rightarrow \frac{m-1}{m} = 1 - \frac{1}{m} \uparrow \Rightarrow$$
$$-\frac{1}{\sigma} \left( \sqrt{\frac{n}{T}} \ln \left( \frac{m-1}{m} \right) + \left( r - \mu + \frac{\sigma^2}{2} \right) \sqrt{\frac{T}{n}} \right) \downarrow \Rightarrow (2.4) \uparrow .$$

More description of the gap risk and their applications will be discussed in chapter 4 and chapter 5.

## 2.2 EPPI in Black-Scholes model

### 2.2.1 The discrete trading time case

It is sometimes not practical to assume that the multiple m is a constant. We consider the case when the multiple is a function of time. Let  $m_{t_k} = \eta + e^{a \ln(S_{t_k}/S_{t_{k-1}})}$  where a > 1. i.e. at time  $t_k$ , we employ the multiple  $m_{t_k}$ , where  $\eta \ge 0$  is a constant. We may as well assume  $\eta = m - 1 \ e^{a \ln(S_{t_k}/S_{t_{k-1}})} = (S_{t_k}/S_{t_{k-1}})^a$ . When  $S_{t_k} > S_{t_{k-1}}$ , the stock price increases. Then

$$(S_{t_k}/S_{t_{k-1}}) > 1, \quad (S_{t_k}/S_{t_{k-1}})^a > 1.$$

Thus,  $m_{t_k} > m$ . This means that we will invest more money into the stock market. When  $S_{t_k} < S_{t_{k-1}}$ , the stock price is decreases. Then

$$(S_{t_k}/S_{t_{k-1}}) < 1, \quad (S_{t_k}/S_{t_{k-1}})^a < 1.$$

Thus,  $m_{t_k} < m$ . This means that we will invest less money into the stock market. When  $S_{t_k} = S_{t_{k-1}}$ , the stock price is not changed. Then

$$(S_{t_k}/S_{t_{k-1}}) = 1, \quad (S_{t_k}/S_{t_{k-1}})^a = 1.$$

Thus,  $m_{t_k} = m$ . This means that we keep the strategy as before.

We call the new strategy an Exponential Proportion Portfolio Insurance (EPPI). This is practical in real markets, the investor would like to invest more money when the stock is increasing and less money when the stock is decreasing. Here the a > 1 is just like a multiplier of the effect of the change of stock market. When we assume a = 0, then  $m_{t_k} = m$  everywhere. The EPPI becomes CPPI. Thus, we can treat EPPI as an extension of CPPI.

**Proposition 2.12.** The cushion of EPPI under the Black-Scholes model in the discrete trading satisfies

$$C_{t_{k+1}} = C_{t_k} \left( m_{t_k} \frac{S_{t_{k+1}}}{S_{t_k}} - (m_{t_k} - 1)e^{rT/n} \right).$$

*Proof.* We have

$$\begin{split} V_{t_{k+1}} &= \frac{m_{t_k} C_{t_k}}{S_{t_k}} S_{t_{k+1}} + (V_{t_k} - m_{t_k} C_{t_k}) \frac{B_{t_{k+1}}}{B_{t_k}} \\ &= m_{t_k} C_{t_k} \frac{S_{t_{k+1}}}{S_{t_k}} + (V_{t_k} - m_{t_k} C_{t_k}) \frac{B_{t_{k+1}}}{B_{t_k}} \\ &= (V_{t_k} - C_{t_k}) \frac{B_{t_{k+1}}}{B_{t_k}} - (m_{t_k} - 1) C_{t_k} \frac{B_{t_{k+1}}}{B_{t_k}} + m_{t_k} C_{t_k} \frac{S_{t_{k+1}}}{S_{t_k}} \\ &= F_{t_k} \frac{B_{t_{k+1}}}{B_{t_k}} + C_{t_k} \left( m_{t_k} \frac{S_{t_{k+1}}}{S_{t_k}} - (m_{t_k} - 1) \frac{B_{t_{k+1}}}{B_{t_k}} \right) \\ &= F_{t_{k+1}} + C_{t_k} \left( m_{t_k} \frac{S_{t_{k+1}}}{S_{t_k}} - (m_{t_k} - 1) \frac{B_{t_{k+1}}}{B_{t_k}} \right). \end{split}$$

Since

$$V_{t_{k+1}} = F_{t_{k+1}} + C_{t_{k+1}},$$

then we have

$$C_{t_{k+1}} = C_{t_k} \left( m_{t_k} \frac{S_{t_{k+1}}}{S_{t_k}} - (m_{t_k} - 1)e^{rT/n} \right).$$

Therefore, we have

$$C_T = C_{t_n} = C_0 \prod_{k=0}^{n-1} \left( m_{t_k} \frac{S_{t_{k+1}}}{S_{t_k}} - (m_{t_k} - 1)e^{rT/n} \right)$$
  
=  $\left( V_0 - Ge^{-rT} \right) \prod_{k=0}^{n-1} \left( m_{t_k} \frac{S_{t_{k+1}}}{S_{t_k}} - (m_{t_k} - 1)e^{rT/n} \right),$ 

and since

$$V_T = C_T + G,$$

thus we get

Proposition 2.13. The EPPI terminal portfolio value under the Black-Scholes model

in the discrete trading is

$$(V_0 - Ge^{-rT}) \prod_{k=0}^{n-1} \left( m_{t_k} \frac{S_{t_{k+1}}}{S_{t_k}} - (m_{t_k} - 1)e^{rT/n} \right) + G.$$

Monte Carlo simulation techniques We want to simulate both the CPPI strategy and EPPI strategy under the Black-scholes model. In the discrete case,

$$\ln \frac{S_{k+1}}{S_k} \sim N\left(\mu \frac{T}{n} - \frac{1}{2}\sigma^2 \frac{T}{n}, \sigma^2 \frac{T}{n}\right).$$

The algorithm could be

Generate 
$$(Z_0, ..., Z_{n-1}) \sim N(0, I)$$
;  
for  $i = 0, 1 ... n - 1$ ;  
 $A_i \leftarrow \exp \mu \frac{T}{n} - \frac{1}{2}\sigma^2 \frac{T}{n} + \sqrt{\sigma^2 \frac{T}{n}} Z_i$ ;  
 $B_i \leftarrow m A_i - (m - 1)e^{rT/n}$ ;  
for  $j = 0, 1, ... n - 1$   
 $V_j^{\text{CPPI}} \leftarrow (V_0 - Ge^{-rT})B_0 * B_1 * ... * B_{j-1} + Ge^{(n-j)T/n}$ ;  
 $m_0 = \eta + 1$ ; for  $i = 1, 2, ... n - 1$   
 $m_{t_k} = \eta + e^{a \ln(A_i)}$ ;  
 $B_i \leftarrow m_{t_k} A_i - (m_{t_k} - 1)e^{rT/n}$ ;  
for  $j = 0, 1, ... n - 1 V_j^{\text{EPPI}} \leftarrow (V_0 - Ge^{-rT})B_0 * B_1 *$   
 $... * B_{j-1} + Ge^{(n-j)T/n}$ ;  
 $\text{plot}(V_j^{\text{CPPI}}, V_j^{\text{EPPI}})$ ;

We use matlab to implement the algorithm. (Figure 2.1)



Figure 2.1: We design the function [Vn, Vn2]=EPPIBS (r, mu, sigma, T, n, s0, m, a, v0, G) with arguments in Matlab to implement the simulation. When in EPPIBS (0.01, 0.02, 0.1, 1, 20, 10, 4, 5, 5000, 4500), this is in particular, here we assume r = 0.01,  $\mu = 0.02$ ,  $\sigma = 0.1$ , T = 1, n = 20 m = 4, a = 5, V(0) = 5000 and floor G = 4500.

### 2.2.2 The continuous trading time case

We still assume the stock price satisfies the Black-Scholes model. Let

$$0 = t_0 < t_1 < t_2 < \dots < t_{n-1} < t_n = T$$

where  $t_{k+1} - t_k = \frac{T}{n}$  for k = 0, ..., n - 1. We reconsider the multiple only at time  $t_i$  which i = 0, 1, ..., n. Let

$$m_0 = \eta + 1;$$
  

$$m_{t_k} = \eta + e^{aln(S_{t_k}/S_{t_{k-1}})} \text{ when } k \ge 1;$$
  

$$m_t = m_{t_k} \text{ when } t \in [t_k, t_{k+1}).$$

In this case, in every interval  $[t_k, t_{k+1})$ , the strategy is standard CPPI. If we let a = 0, then it is same as standard CPPI. Therefore, we can treat this strategy also as an extension of standard CPPI. We deduced

$$C_{t} = C_{t_{k}} \exp\left\{\left(r + m_{t_{k}}(\mu - r) - \frac{1}{2}m_{t_{k}}^{2}\right)(t - t_{k}) + \sigma m_{t_{k}}(W_{t} - W_{t_{k}})\right\}$$

when  $t \in [t_k, t_{k+1})$ , and thus

$$C_{t_{k+1}} = C_{t_k} \left( m_{t_k} \frac{S_{t_{k+1}}}{S_{t_k}} - (m_{t_k} - 1)e^{rT/n} \right).$$

Therefore, we have

$$C_{T} = C_{0} \exp\left\{rT + \frac{m_{0} + \ldots + m_{t_{k-1}}(\mu - r)}{n}T - \frac{\sigma^{2}}{2}\frac{m_{0}^{2} + \ldots + m_{t_{k-1}}^{2}}{n}T + \sigma\left(\sum_{i=0}^{n-1} m_{t_{i}}(W_{t_{i+1}} - W_{t_{i}})\right)\right\}$$
  
$$= \left(V_{0} - Ge^{-rT}\right) \exp\left\{rT + \frac{m_{0} + \ldots + m_{t_{k-1}}}{n}(\mu - r)T - \frac{\sigma^{2}}{2}\frac{m_{0}^{2} + \ldots + m_{t_{k-1}}^{2}}{n}T + \sigma\left(\sum_{i=0}^{n-1} m_{t_{i}}(W_{t_{i+1}} - W_{t_{i}})\right)\right\}.$$

Since

$$V_T = C_T + G,$$

thus we get

**Proposition 2.14.** The terminal EPPI portfolio value under the Black-Scholes model in the discrete trading is

$$\left( V_0 - Ge^{-rT} \right) \exp\left\{ rT + \frac{m_0 + \dots + m_{t_{k-1}}}{n} (\mu - r)T - \frac{\sigma^2}{2} \frac{m_0^2 + \dots + m_{t_{k-1}}^2}{n} T + \sigma \left( \sum_{i=0}^{n-1} m_{t_i} (W_{t_{i+1}} - W_{t_i}) \right) \right\} + G.$$

Monte Carlo simulation techniques In this case, the algorithm could be

Generate  $(Z_1, ..., Z_n) \sim N(0, I);$ for i = 0, 1 ... n - 1;  $A_i \leftarrow \exp \mu \frac{T}{n} - \frac{1}{2}\sigma^2 \frac{T}{n} + \sqrt{\sigma^2 \frac{T}{n}} Z_i;$   $m_0 = \eta + 1; C_{t_0} = V_{t_0} - Ge^{-rT};$ for i = 1, 2, ... n - 1  $m_{t_k} = \eta + e^{a \ln(A_i)};$   $C_{t_{k+1}} = C_{t_k} \left( m_{t_k} A_i - (m_{t_k} - 1)e^{rT/n} \right)$   $V_{t_{k+1}} = C_{t_{k+1}} + Ge^{(n-k-1)rT/n};$ plot(V);

## 2.3 CPPI in GARCH model

### 2.3.1 The Continuous trading time cases

Here instead of treating the volatility as a constant, we consider the following model. The ARCH/GARCH model considers the volatility which depend on the past history. In particular consider the **GARCH(p,q)** model:

$$\ln \frac{S_t}{S_{t-1}} = \mu(\sigma_t) - \frac{\sigma_t^2}{2} + \sigma_t \epsilon_t,$$

where  $\mu$  is a given function,  $\epsilon_1$ ,  $\epsilon_2$ ,...is a sequence of i.i.d. standard normal random variables, and  $\sigma_t$  satisfies:

$$\sigma_t^2 = \omega + \sum_{i=1}^p \alpha_i (\sigma_{t-i} \epsilon_{t-i})^2 + \sum_{j=1}^q \beta_j \sigma_{t-j}^2$$

 $\omega, \alpha_1, ... \alpha_p, \beta_1, ..., \beta_q$  being fixed constants.

By an embedding methodology, we can recover the continuous-time GARCH(1, 1) model. (Refer to [44])

$$F(X)_t = \begin{cases} \sigma_0 & \text{for } 0 \le t < 1\\ \left(\omega + \alpha \left(X_{[t]-X_{[t]-1}}\right)^2 + \beta F(x)_{[t]-1}^2\right)^{1/2} & \text{for } t \ge 1 \end{cases};$$
  

$$S_t = S_0 \exp\left\{\int_0^t (\mu(\sigma_{s-}) - \frac{\sigma_{s-}^2}{2})ds + X_t\right\};$$
  

$$X_t = \int_0^t \sigma_{s-}dB_s;$$
  

$$\sigma := F(x).$$

We have

$$dS_t = S_t \left( \mu(\sigma_{t-})dt + \sigma_{t-}dB_t \right).$$

Next we have,

**Proposition 2.15.** The CPPI cushion under GARCH(1, 1) model in the continuous trading time case above satisfies

$$C_t = C_0 \exp\left(\int_0^t \left(\mu(\sigma_{s-}) - \frac{m^2 \sigma_{s-}^2}{2}\right) ds + m \int_0^t \sigma_{s-} dB_s - (m-1)rt\right).$$
 (2.5)

*Proof.* Since the strategy is self-financing, thus, we have

$$V_t = V_t \left(\frac{mC_t}{V_t} + \left(1 - \frac{mC_t}{V_t}\right)\right)$$

and

$$dV_t = V_t \left(\frac{mC_t}{V_t}\frac{dS_t}{S_t} + \left(1 - \frac{mC_t}{V_t}\right)\frac{dB_t}{B_t}\right).$$

Then

$$\begin{split} dC_t &= d(V_t - F_t) \\ &= V_t \left( \frac{mC_t}{V_t} \frac{dS_t}{S_t} + \left( 1 - \frac{mC_t}{V_t} \right) \frac{dB_t}{B_t} \right) - F_t \frac{dB_t}{B_t} \\ &= C_t \left( \frac{mdS_t}{S_t} - (m-1)rdt \right) \\ &= C_t((\mu(\sigma_{t-})dt + \sigma_{t-}dB_t) - (m-1)rdt). \end{split}$$

Hence, we get

$$C_{t} = C_{0} \exp\left(\int_{0}^{t} \left(\mu(\sigma_{s-}) - \frac{m^{2}\sigma_{s-}^{2}}{2}\right) ds + m \int_{0}^{t} \sigma_{s-} dB_{s} - (m-1)rt\right).$$

We then got the *portfolio value* is

$$V_t = C_t + F_t = C_t + G \exp\{-r(T-t)\}.$$

The following proposition is the property of GARCH(1, 1) model.

**Proposition 2.16.** Let  $n \in \mathbb{N}$ , we have

$$\mathbb{E}[\sigma_n^2] = (\alpha + \beta)^{n-1} \left(\sigma_0^2 + \frac{\omega}{\alpha + \beta - 1}\right) - \frac{\omega}{\alpha + \beta - 1}.$$
(2.6)

*Proof.* By definition

$$F(X)_{t} = \begin{cases} \sigma_{0} \text{ for } 0 \leq t < 1\\ \left(\omega + \alpha \left(X_{[t]-X_{[t]-1}}\right)^{2} + \beta F(x)_{[t]-1}^{2}\right)^{1/2} \text{ for } t \geq 1 \end{cases}$$

$$S_{t} = S_{0} \exp\left\{\int_{0}^{t} (\mu(\sigma_{s-}) - \frac{\sigma_{s-}^{2}}{2})ds + X_{t}\right\};$$

$$X_{t} = \int_{0}^{t} \sigma_{s-}dB_{s};$$
26
$$\sigma := F(x).$$

then

$$\begin{cases} \sigma_t = \sigma_0 \text{ for } 0 \le t < 1\\ \sigma_t^2 = \sigma_{[t]}^2 \omega + \alpha \left( \int_{[t]-1}^{[t]} \sigma_{s-} dB_s \right)^2 + \beta \sigma_{[t]-1}^2 \text{ for } t \ge 1 \end{cases},$$

thus,

$$\sigma_n^2 = \omega + \alpha \left( \int_{n-1}^n \sigma_{s-} dB_s \right)^2 + \beta \sigma_{n-1}^2$$
  
=  $\omega + \alpha \left( \sigma_{n-1} (B_n - B_{n-1}) \right)^2 + \beta \sigma_{n-1}^2$   
=  $\omega + \sigma_{n-1}^2 (\alpha (B_n - B_{n-1})^2 + \beta),$ 

then

$$\mathbb{E}\sigma_n^2 = \omega + E\sigma_{n-1}^2(\alpha + \beta),$$

and hence

$$\mathbb{E}[\sigma_n^2] + \frac{\omega}{\alpha + \beta - 1} = (\alpha + \beta) \left( \mathbb{E}[\sigma_{n-1}^2] + \frac{\omega}{\alpha + \beta - 1} \right).$$

Thus, we get

$$\mathbb{E}[\sigma_n^2] = (\alpha + \beta)^{n-1} \left( \sigma_0^2 + \frac{\omega}{\alpha + \beta - 1} \right) - \frac{\omega}{\alpha + \beta - 1}.$$

In order to calculate the expectation of  $V_t$  explicitly, in the following, we assume  $\mu(z) = \mu$  be constant function.

**Lemma 2.17.** Let 
$$h_t = \exp\left(\int_0^t \left(m\sigma_{s-}dB_s - \int_0^t \frac{m^2\sigma_{s-}^2}{2}\right)ds\right)$$
, then  $\mathbb{E}[h_t] = 1$ .

Proof. By Ito formula

$$dh_t = m\sigma_{t-}h_t dB_t,$$

thus

$$h_t - h_0 = \int_0^t m\sigma_{s-} h_s dB_s.$$

Hence  $h_t$  is a martingale and thus  $\mathbb{E}[h_t] = \mathbb{E}[h_0] = 1$ .

**Proposition 2.18.** Let  $\mu(z) = \mu$  be constant function. Then the expectation of the CPPI portfolio value  $V_t$  under the GARCH(1,1) model in the continuous trading time case is

$$\mathbb{E}(V_t) = Ge^{-r(T-t)} + (V_0 - Ge^{-rT})e^{m\mu t - (m-1)rt}.$$

Proof.

$$\mathbb{E}(V_t) = Ge^{-r(T-t)} + C_0 e^{m \int_0^t \mu(\sigma_{s-})ds - (m-1)rt)} \\
\times \mathbb{E}\left[\exp\left(\int_0^t \left(m\sigma_{s-}dB_s - \int_0^t \frac{m^2\sigma_{s-}^2}{2}\right)ds\right)\right] \\
= Ge^{-r(T-t)} + (V_0 - Ge^{-rT})e^{m\mu t - (m-1)rt}.$$

#### 2.3.2 The Discrete trading time case GARCH(1,1) model

In this case, the model is

$$\ln \frac{S_t}{S_{t-1}} = \mu(\sigma_t) - \frac{\sigma_t^2}{2} + \sigma_t \epsilon_t;$$
$$\sigma_t^2 = \omega + \alpha (\sigma_{t-1} \epsilon_{t-1})^2 + \beta \sigma_{t-1}^2;$$

where  $\omega$ ,  $\alpha$ ,  $\beta$  are fixed constants. We only consider the time on the integer-value, i.e. integer times unit time.

**Proposition 2.19.** The CPPI cushion under GARCH(1,1) model in the discrete trading time case model satisfies

$$C_{t+1} = C_t \left( m \frac{S_{t+1}}{S_t} - (m-1)e^r \right).$$
(2.7)

*Proof.* Since the strategy is self-financing, we have

$$V_{t+1} = (V_t - mC_t) \frac{B_{t+1}}{B_t} + mC_t(S_{t+1}/S_t)$$
  
=  $(V_t - C_t) \frac{B_{t+1}}{B_t} - (m-1)C_t \frac{B_{t+1}}{B_t} + mC_t(S_{t+1}/S_t)$   
=  $F_t \frac{B_{t+1}}{B_t} + C_t \left( m \frac{S_{t+1}}{S_t} - (m-1) \frac{B_{t+1}}{B_t} \right)$   
=  $F_{t+1} + C_t \left( m \frac{S_{t+1}}{S_t} - (m-1) \frac{B_{t+1}}{B_t} \right)$ ,

and since

$$V_{t+1} = F_{t+1} + C_{t+1},$$

then

$$C_{t+1} = C_t \left( m \frac{S_{t+1}}{S_t} - (m-1)e^r \right).$$

-	_

We then have

$$C_n = C_0 \prod_{k=0}^{n-1} \left( m \frac{S_{k+1}}{S_k} - (m-1)e^r \right),$$

and

$$V_n = C_n + F_n = C_n + G.$$

#### Monte Carlo simulation techniques Our algorithm could be

Generate 
$$(\epsilon_1, \epsilon_2, \dots \epsilon_t) \sim N(0, I)$$
  
for  $i = 1, \dots t$   
 $\sigma_t \leftarrow \sqrt{\omega + \alpha(\sigma_{t-1}\epsilon_{t-1})^2 + \beta\sigma_{t-1}^2};$   
 $A_i \leftarrow \exp\left(\mu(\sigma_t) - \frac{\sigma_t^2}{2} + \sigma_t\epsilon_t\right);$   
 $C_0 = Ge^{-rt};$   
for  $i = 0, \dots t - 1$   
 $C_{i+1} \leftarrow C_t((mA_i - (m-1)e^r));$   
 $V_{i+1} = C_{i+1} + Ge^{-(t-i)};$   
plot $(V);$ 

We use Matlab to implement the strategy according to the above algorithm.(Figure 2.2)

# **2.4 EPPI in GARCH**(1, 1) **model**

We consider the EPPI in GARCH(1, 1) model. We assume the stock price satisfy:

$$\ln \frac{S_t}{S_{t-1}} = \mu(\sigma_t) - \frac{\sigma_t^2}{s} + \sigma_t \epsilon_t;$$
$$\sigma_t^2 = \omega + \alpha (\sigma_{t-1} \epsilon_{t-1})^2 + \beta \sigma_{t-1}^2,$$

where  $\omega$ ,  $\alpha$ ,  $\beta$  are fixed constants and the multiple is

$$m_t = \eta + \exp\{a \ln(S_t / S_{t-1})\}.$$

**Proposition 2.20.** The EPPI cushion in the GARCH(1,1) model satisfies

$$C_{t+1} = C_t \left( m_t \frac{S_{t+1}}{S_t} - (m_t - 1)e^r \right)$$
  
30



Figure 2.2: We design the function GARCHCPPI(r, mu, sigma0, alpha1, beta1,omega, n, v0, G, m) with arguments to implement the simulation. When in GARCHCPPI(0.0001, 0.00015, 0.0003, 0.05, 0.05,0.0002, 100, 5000, 4500, 4), this is in particular, here we set  $\mu(\sigma_{s-}$  is constant, r = 0.0001,  $\mu = 0.00015$ ,  $\sigma_0 = 0.0003$ ,  $\alpha = 0.05$ ,  $\beta = 0.05$ ,  $\omega = 0.0002$ , n = 100, m = 4, V(0) = 5000 and floor G = 4500.

*Proof.* Since the strategy is self-financing, we have

$$\begin{aligned} V_{t+1} &= (V_t - m_t C_t) \frac{B_{t+1}}{B_t} + m_t C_t (S_{t+1}/S_t) \\ &= (V_t - C_t) \frac{B_{t+1}}{B_t} - (m_t - 1) C_t \frac{B_{t+1}}{B_t} + m_t C_t (S_{t+1}/S_t) \\ &= F_t \frac{B_{t+1}}{B_t} + C_t \left( m_t \frac{S_{t+1}}{S_t} - (m_t - 1) \frac{B_{t+1}}{B_t} \right) \\ &= F_{t+1} + C_t \left( m_t \frac{S_{t+1}}{S_t} - (m_t - 1) \frac{B_{t+1}}{B_t} \right), \end{aligned}$$

and since

$$V_{t+1} = F_{t+1} + C_{t+1},$$

then

$$C_{t+1} = C_t \left( m_t \frac{S_{t+1}}{S_t} - (m_t - 1)e^r \right).$$

_		-
L		н
L		н
L		н

Therefore we have

$$C_n = C_0 \prod_{k=0}^{n-1} \left( m_k \frac{S_{k+1}}{S_k} - (m_k - 1)e^r \right),$$

and

$$V_n = C_n + F_n = C_n + G.$$

#### Monte Carlo simulation techniques Our algorithm could be

Generate 
$$(\epsilon_1, \epsilon_2, \dots \epsilon_t) \sim N(0, I)$$
  
for  $i = 1, \dots t$   
 $\sigma_t \leftarrow \sqrt{\omega + \alpha(\sigma_{t-1}\epsilon_{t-1})^2 + \beta\sigma_{t-1}^2};$   
 $A_i \leftarrow \exp\left(\mu(\sigma_t) - \frac{\sigma_t^2}{2} + \sigma_t\epsilon_t\right);$   
 $C_0 = Ge^{-rt};$   
 $m_0 = m;$   
for  $i = 0, \dots t - 1$   
 $m_{i+1} = m - 1 + \exp\left(a\ln(A_i)\right);$   
 $C_{i+1} \leftarrow C_t((m_iA_i - (m_i - 1)e^r));$   
 $V_{i+1} = C_{i+1} + Ge^{-(t-i)};$   
plot(V);

We use Matlab to implement the strategy according the above algorithm. (Figure 2.3)



Figure 2.3: We design the function GARCHEPPI(r, mu, sigma0, alpha1, beta1, omega, n, v0, G, m,a) with arguments to implement the simulation. When in GARCHEPPI(0.0001, 0.00015, 0.0003, 0.05, 0.05,0.0002, 100, 5000, 4500, 4,2), this is particular, here we set  $\mu(\sigma_{s-})$  is constant, r = 0.0001,  $\mu = 0.00015$ ,  $\sigma_0 = 0.0003$ ,  $\alpha = 0.05$ ,  $\beta = 0.05$ ,  $\omega = 0.0002$ , n = 100, m = 4, a = 2, V(0) = 5000 and floor G = 4500.

The next figure draws the EPPI versus CPPI in GARCH. (Figure 2.4)



Figure 2.4: We design the function GARCHEPPIvsCPPI(r, mu, sigma0, alpha1, beta1, omega, n, v0, G, m,a)to implement the simulation. When in [y1,y2]=GARCHEPPIvsCPPI(0.0001, 0.00015, 0.0003, 0.05, 0.05,0.0002, 100, 5000, 4500, 4,2), this is particular, here we set  $\mu(\sigma_{s-})$  is constant, r = 0.0001,  $\mu = 0.00015$ ,  $\sigma_0 = 0.0003$ ,  $\alpha = 0.05$ ,  $\beta = 0.05$ ,  $\omega = 0.0002$ , n = 100, m = 4, a = 2, V(0) = 5000 and floor G = 4500.

# Chapter 3

# CPPI in the Jump-diffusion model when the trading time is continuous

#### 3.1 Jump-diffusion model

#### 3.1.1 Set up the model:

In this section, we consider the jump-diffusion model. It has been studied by many researchers since the Merton's Paper [58]. The model in our paper is described in section 3.1.1 of [65]. [53] is another survey paper about jump-diffusion model, which gives four reasons for choosing the jump-diffusion models. [53] also gives the short-coming of the jump-diffusion model. We also want to mention [18], [71], [33], [57], [50] among others, for further information.

Let  $(\Omega, \mathfrak{F}, \mathfrak{F}_t, \mathbb{P})$  be a probability space satisfying the "usual assumption". Let the price  $S_t$  of a risky asset (usually stocks or their benchmark) be a right continuous with left limits stochastic process on this probability space which jumps at the random times  $T_1, T_2,...$  and suppose that the relative/proportional change in its value at a jump time is given by  $Y_1, Y_2,...$  respectively. We assume  $\ln(1 + Y_n)$ s be i.i.d., and denote the density of  $\ln(1+Y_n)$ s by  $f_Q$ . We assume that, between any two consecutive jump times, the price  $S_t$  follows the Black-Scholes model. These  $T_n$ s are the jump times of a Poisson process  $N_t$  with intensity  $\lambda_t$  and the  $Y_n$ s are a sequence of random variables with values in  $(-1, +\infty)$ . We have

$$N_t = \sum_{n \ge 1} \chi_{t \ge T_n}$$

and

$$\mathbb{P}[N_t = n] = \frac{e^{-\int_0^t \lambda_s ds} \left(\int_0^t \lambda_s ds\right)^n}{n!}.$$

Then on the intervals  $[T_n, T_{n+1})$ , the description of the model can be formalized by letting,

$$dS_t = S_t(\mu_t dt + \sigma_t dW_t),$$

and in exponential form:

$$S_t = S_{T_n} \exp\left[\int_{T_n}^t \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds + \int_0^t \sigma_s dW_s\right].$$

At  $t = T_n$ , the jump size is given by  $\Delta S_n = S_{T_n} - S_{T_n} = S_{T_n-}Y_n$ , i.e.

$$S_{T_n} = S_{T_n^-}(1+Y_n)$$

which, by the assumption that  $Y_n > -1$ , leads to always positive values of the prices. At the generic time t,  $S_t$  can be expressed by the following equivalent representations

$$S_t = S_0 \exp\left[\int_0^t \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds + \int_0^t \sigma_s dW_s\right] \left[\prod_{n=1}^{N_t} (1+Y_n)\right]$$
(3.1)

$$= S_0 \exp\left[\int_0^t \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds + \int_0^t \sigma_s dW_s + \sum_{n=1}^{N_t} \ln(1+Y_n)\right]$$
(3.2)

$$= S_0 \exp\left[\int_0^t \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds + \int_0^t \sigma_s dW_s + \int_0^t \ln(1+Y_s) dN_s\right]$$
(3.3)

where  $Y_t$  is obtained from  $Y_n$  by a piecewise constant and left continuous time interpolation, i.e.

$$Y_t = Y_n \text{ if } T_n < t \le T_{n+1},$$

here we let  $T_0 = 0$ . The term  $\sum_{n=1}^{N_t} \ln(1 + Y_n)$ ] in (3.2) is a compound Poisson process. It has independent and stationary increments. Also because of (3.2), our jump-diffusion model is an exponential levy model. Moreover, by the generalized Ito formula, the processes  $S_t$  is the solution of

$$dS_t = S_{t-}[\mu_t dt + \sigma_t dW_t + Y_t dN_t], \qquad (3.4)$$

with initial value  $S_0 = s$ .

#### 3.1.2 Two special Jump-diffusion models

Two important special jump-diffusion models will be considered and we introduce them here. The Merton's Model When we assume  $\ln(1 + Y_n) \sim N(\alpha, \delta^2)$ . This is the Merton's model ([58]). The following Proposition considers the density of  $\ln(\frac{S_t}{S_0})$ .

**Proposition 3.1.** Let  $\phi(x, m, v^2)$  be a density function for a normally distributed random variable with mean m and variance  $v^2$ , i.e.  $\phi(x, m, v^2) = \frac{1}{\sqrt{2\pi v^2}} e^{-\frac{(x-m)^2}{2v^2}}$ . Then, the density function of

$$\ln\left(\frac{S_t}{S_0}\right) = \int_0^t \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds + \int_0^t \sigma_s dW_s + \sum_{n=1}^{N_t} \ln(1+Y_n)$$

is:

$$p(x) = \sum_{j=0}^{\infty} \frac{e^{-\int_0^t \lambda_s ds} (\int_0^t \lambda_s ds)^j}{j!} \phi\left(x; \int_0^t \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds + j\alpha, \int_0^t \sigma_s^2 ds + j\delta^2\right). \quad (3.5)$$

*Proof.* Let  $L = \int_0^t \left( \mu_s - \frac{\sigma_s^2}{2} \right) ds + \int_0^t \sigma_s dW_s$  and  $M = \sum_{n=1}^{N_t} \ln(1+Y_n)$ . Then we have,

$$L \sim N\left(\int_0^t \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds, \int_0^t \sigma_s^2 ds\right).$$

When  $N_t = j$ , by the properties of normal distribution, we have

$$L + M \sim N\left(\int_0^t \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds + j\alpha, \int_0^t \sigma_s^2 ds + j\delta^2\right).$$

In general,

$$\forall x \in \mathbb{R},$$
$$\mathbb{P}(L+M \le x) = \mathbb{P}\left(\bigcup_{j=0}^{\infty} (L+M \le x, N_t = j)\right)$$
$$= \sum_{j=0}^{\infty} \mathbb{P}(L+M \le x, N_t = j) = \sum_{j=0}^{\infty} \mathbb{P}(L+M \le x | N_t = j) \mathbb{P}(N_t = j)$$

$$= \sum_{j=0}^{\infty} \mathbb{P}(L + \Sigma_{n=1}^{j} \ln(1 + Y_{n}) \le x | N_{t} = j) \mathbb{P}(N_{t} = j)$$

$$= \sum_{j=0}^{\infty} \frac{\mathbb{P}((L + \Sigma_{n=1}^{j} \ln(1 + Y_{n}) \le x, N_{t} = j))}{\mathbb{P}(N_{t} = j)} P(N_{t} = j)$$

$$= \sum_{j=0}^{\infty} \frac{\mathbb{P}((L + \Sigma_{n=1}^{j} \ln(1 + Y_{n}) \le x) P(N_{t} = j))}{\mathbb{P}(N_{t} = j)} P(N_{t} = j)$$

$$= \sum_{j=0}^{\infty} \mathbb{P}(L + \Sigma_{n=1}^{j} \ln(1 + Y_{n}) \le x) \mathbb{P}(N_{t} = j)$$

$$= \sum_{j=0}^{\infty} \int_{-\infty}^{x} \phi\left(y; \int_{0}^{t} \left(\mu_{s} - \frac{\sigma_{s}^{2}}{2}\right) ds + j\alpha, \int_{0}^{t} \sigma_{s}^{2} ds + j\delta^{2}\right) dy \frac{e^{-\int_{0}^{t} \lambda_{s} ds} (\int_{0}^{t} \lambda_{s} ds)^{j}}{j!}$$

When j = 0, we take  $\sum_{n=1}^{j} \ln(1+Y_n) = 0$ . Each item in the above equations is positive, thus the series is absolute convergence. Thus, the density function is

$$= \frac{p(x)}{d\left(\sum_{j=0}^{\infty}\int_{-\infty}^{x}\phi\left(y;\int_{0}^{t}\left(\mu_{s}-\frac{\sigma_{s}^{2}}{2}\right)ds+j\alpha,\int_{0}^{t}\sigma_{s}^{2}ds+j\delta^{2}\right)dy\frac{e^{-\int_{0}^{t}\lambda_{s}ds}(\int_{0}^{t}\lambda_{s}ds)^{j}}{j!}\right)}{dx}$$

$$= \sum_{j=0}^{\infty}\frac{e^{-\int_{0}^{t}\lambda_{s}ds}(\int_{0}^{t}\lambda_{s}ds)^{j}}{j!}\phi\left(x;\int_{0}^{t}\left(\mu_{s}-\frac{\sigma_{s}^{2}}{2}\right)ds+j\alpha,\int_{0}^{t}\sigma_{s}^{2}ds+j\delta^{2}\right).$$

**The Kou's Model** When we assume  $Q = \ln(1 + Y_n)$  has an asymmetric double exponential distribution with the density

$$f_Q(y) = p \cdot \eta_1 e^{-\eta_1 y} \chi_{y \ge 0} + q \cdot \eta_2 e^{-\eta_2 y} \chi_{y < 0}$$

where  $\eta_1 > 1$ ,  $\eta_2 > 0$ ,  $p, q \ge 0$  and p + q = 1.

This is called the **Kou's** model([51]). We have:

Proposition 3.2. The density function of

$$\ln\left(\frac{S_t}{S_0}\right) = \int_0^t \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds + \int_0^t \sigma_s dW_s + \sum_{n=1}^{N_t} \ln(1+Y_n)$$

is:

$$p(x) = \sum_{j=0}^{\infty} \frac{e^{-\int_{0}^{t} \lambda_{s} ds} (\int_{0}^{t} \lambda_{s} ds)^{j}}{j!} \int_{-\infty}^{\infty} \phi\left(x - y; \int_{0}^{t} \left(\mu_{s} - \frac{\sigma_{s}^{2}}{2}\right) ds, \int_{0}^{t} \sigma_{s}^{2} ds\right) f_{Q}^{(j)}(y) dy,$$

where  $f_Q^{(j)}(y)$  is the density function of  $\sum_{n=1}^j \ln(1+Y_n)$ .

*Proof.* Let  $L = \int_0^t \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds + \int_0^t \sigma_s dW_s$  and  $M = \sum_{n=1}^{N_t} \ln(1+Y_n)$ . Then,

$$L \sim N\left(\int_0^t \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds, \int_0^t \sigma_s^2 ds\right).$$

When  $N_t = j$ , we have the distribution of the sum of two random variables is

$$\mathbb{P}(L+M \le x) = \int_{-\infty}^x \int_{-\infty}^\infty \phi\left(y-y_2; \int_0^t \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds, \int_0^t \sigma_s^2 ds\right) f_Q^{(j)}(y_2) dy_2 dy.$$

We calculate the distribution of L + M in general.

$$\begin{aligned} \forall x \in \mathbb{R} \\ \mathbb{P}\left(L + M \le x\right) &= \mathbb{P}\left(\bigcup_{j=0}^{\infty} (L + M \le x, N_t = j)\right) \\ &= \sum_{j=0}^{\infty} \mathbb{P}\left((L + M \le x, N_t = j)\right) = \sum_{j=0}^{\infty} \mathbb{P}\left((L + M \le x, N_t = j)\right) \\ &= \sum_{j=0}^{\infty} \mathbb{P}\left(L + M \le x | N_t = j\right) P(N_t = j) \\ &= \sum_{j=0}^{\infty} \mathbb{P}\left(L + \sum_{n=1}^{j} \ln(1 + Y_n) \le x | N_t = j\right) \mathbb{P}(N_t = j) \\ &= \sum_{j=0}^{\infty} \frac{\mathbb{P}\left(L + \sum_{n=1}^{j} \ln(1 + Y_n) \le x, N_t = j\right)}{\mathbb{P}(N_t = j)} \mathbb{P}(N_t = j) \end{aligned}$$

$$= \sum_{j=0}^{\infty} \frac{\mathbb{P}(L + \Sigma_{n=1}^{j} \ln(1+Y_{n}) \leq x) \mathbb{P}(N_{t}=j)}{\mathbb{P}(N_{t}=j)} \mathbb{P}(N_{t}=j)$$

$$= \sum_{j=0}^{\infty} \mathbb{P}((L + \Sigma_{n=1}^{j} \ln(1+Y_{n}) \leq x)) \mathbb{P}(N_{t}=j)$$

$$= \sum_{j=0}^{\infty} \int_{-\infty}^{x} \int_{-\infty}^{\infty} \phi\left(y - y_{2}; \int_{0}^{t} \left(\mu_{s} - \frac{\sigma_{s}^{2}}{2}\right) ds, \int_{0}^{t} \sigma_{s}^{2} ds\right) f_{Q}^{(j)}(y_{2}) dy_{2} dy$$

$$\times \frac{e^{-\int_{0}^{t} \lambda_{s} ds} (\int_{0}^{t} \lambda_{s} ds)^{j}}{j!}.$$

Each item in the above equations is positive, thus the series is absolute convergence. Hence, the density function is

$$= \frac{p(x)}{d\left(\sum_{j=0}^{\infty}\int_{-\infty}^{x}\int_{-\infty}^{\infty}\phi\left(y-y_{2};\int_{0}^{t}\left(\mu_{s}-\frac{\sigma_{s}^{2}}{2}\right)ds,\int_{0}^{t}\sigma_{s}^{2}ds\right)f_{Q}^{(j)}(y_{2})dy_{2}dy\frac{e^{-\int_{0}^{t}\lambda_{s}ds}(\int_{0}^{t}\lambda_{s}ds)^{j}}{j!}\right)}{dx}$$

$$= \sum_{j=0}^{\infty}\frac{e^{-\int_{0}^{t}\lambda_{s}ds}(\int_{0}^{t}\lambda_{s}ds)^{j}}{j!}\int_{-\infty}^{\infty}\phi\left(x-y;\int_{0}^{t}\left(\mu_{s}-\frac{\sigma_{s}^{2}}{2}\right)ds,\int_{0}^{t}\sigma_{s}^{2}ds\right)f_{Q}^{(j)}(y)dy.$$

 $f_Q^{(j)}(y)$  can be calculated by the convolution of  $k \ f_Q(y)$ s. i.e.

$$f_Q^{(j)}(y) = \underbrace{f_Q(y) * f_Q(y) * \dots f_Q(y)}_{j \text{ terms}}.$$
(3.6)

Thus, the density function could be calculated explicitly. In generally, when we assume  $Q_n = \ln(1+Y_n)$  have i.i.d. with density  $f_Q$ , then the density of  $\sum_{n=1}^{j} \ln(1+Y_n)$  is  $f_Q^{(j)}$ . We have the following proposition:

**Proposition 3.3.** Let  $Q_n = \ln(1+Y_n)$  be *i.i.d.* random variables with density function  $f_Q$ . The density function of

$$\ln\left(\frac{S_t}{S_0}\right) = \int_0^t \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds + \int_{4\frac{1}{2}0}^t \sigma_s dW_s + \sum_{n=1}^{N_t} \ln(1+Y_n)$$

$$p(x) = \sum_{j=0}^{\infty} \frac{e^{-\int_0^t \lambda_s ds} (\int_0^t \lambda_s ds)^j}{j!} \int_{-\infty}^\infty \phi\left(x - y; \int_0^t \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds, \int_0^t \sigma_s^2 ds\right) f_Q^{(j)}(y) dy.$$

#### 3.1.3 Martingale Measure

For our jump-diffusion model defined by (3.2), consider a predictable  $\mathfrak{F}_t$ -process  $\psi_t$ , such that  $\int_0^t \psi_t \lambda_s ds < \infty$ . Choose  $\theta_t$  and  $\psi_t$  such that

$$\mu_t + \sigma_t \theta_t + Y_t \psi_t \lambda_t = r_t \tag{3.7}$$

and

 $\psi_t \ge 0.$ 

From here we see that

$$\theta_t = \sigma_t^{-1} (r_t - \mu_t - Y_t \psi_t \lambda_t) \tag{3.8}$$

where  $\psi_t$  is arbitrary. Define

$$L_t = \exp\left\{\int_0^t \left[(1-\psi_s)\lambda_s - \frac{1}{2}\theta_s^2\right]ds + \int_0^t \theta_s dW_s + \int_0^t \ln\psi_s dN_s\right\}$$
(3.9)

for  $t \in [0, T]$  and the Radon-Nykodymn derivative

$$\frac{d\mathbb{Q}}{d\mathbb{P}} = L_T. \tag{3.10}$$

Then the  $\mathbb{Q}$  is a risk neutral measure or martingale measure, i.e. a measure under which  $\tilde{S}_t = \exp\{-\int_0^t r_s ds\}S_t$  is a martingale (see [65]).

is:

Define

$$dW_t^{\mathbb{Q}} = dW_t - \theta_t dt; \tag{3.11}$$

$$dM_t^{\mathbb{Q}} = dN_t - \psi_t \lambda_t dt. \tag{3.12}$$

Then  $W_t^{\mathbb{Q}}$  and  $M_t^{\mathbb{Q}}$  are  $\mathbb{Q}$ -martingales. Also under the measure  $\mathbb{Q}$ ,  $S_t$  satisfies

$$dS_t = S_{t-} \left[ (\mu_t + \sigma_t \theta_t + Y_t \psi_t \lambda_t) dt + \sigma_t dW_t^{\mathbb{Q}} + Y_t dM_t^{\mathbb{Q}} \right].$$
(3.13)

Under the measure  $\mathbb{Q}$ ,  $N_t$  is a Poisson Processes with intensity  $\lambda_t \psi_t$ .

There are many risk-neutral measures  $\mathbb{Q} \sim \mathbb{P}$ . A special case of a risk-neutral measure, reflecting the case of a risk-neutral world, it should satisfy

$$\mathbb{E}(S(t)) = S_0 e^{rt}.$$

(See page 312 on [33], page 248-250 on [38], page 19 on [57].)

For Merton Model, since its density function has explicit expression, we will deduce it. We have deduce the density function of Merton's model in (3.5).

$$p(x) = \sum_{j=0}^{\infty} \frac{e^{-\int_0^t \lambda_s ds} (\int_0^t \lambda_s ds)^j}{j!} \phi\left(x; \int_0^t \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds + j\alpha, \int_0^t \sigma_s^2 ds + j\delta^2\right).$$

Then

$$\mathbb{E}(S(t)) = S_0 \mathbb{E}\left(e^{\ln S_t/S_0}\right) = S_0 \int_{\mathbb{R}} e^x p(x) dx$$

$$= S_0 \int_{\mathbb{R}} e^x \sum_{j=0}^{\infty} \frac{e^{-\int_0^t \lambda_s ds} (\int_0^t \lambda_s ds)^j}{j!} \phi\left(x; \int_0^t \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds + j\alpha, \int_0^t \sigma_s^2 ds + j\delta^2\right) dx$$

$$= S_0 \sum_{j=0}^{\infty} \frac{e^{-\int_0^t \lambda_s ds} (\int_0^t \lambda_s ds)^j}{j!} \int_{\mathbb{R}} e^x \phi\left(x; \int_0^t \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds + j\alpha, \int_0^t \sigma_s^2 ds + j\delta^2\right) dx$$

$$= 43$$

$$= S_0 \sum_{j=0}^{\infty} \frac{e^{-\int_0^t \lambda_s ds} (\int_0^t \lambda_s ds)^j}{j!} \exp\left\{\int_0^t \mu_s ds + j\alpha + j\frac{\delta^2}{2}\right\}$$
$$= S_0 \exp\left[\int_0^t \left(\mu_s - \lambda_s + e^{\alpha + \frac{\delta^2}{2}} \lambda_s\right) ds.$$

In case of

$$\mathbb{E}(S(t)) = S_0 e^{rt},$$

then we have

$$\mu_s - \lambda_s + e^{\alpha + \frac{\delta^2}{2}} \lambda_s = r.$$

Thus under our new risk-neutral measure  $\mathbb{P}^{rn}$ , we can use  $r + \lambda_s - e^{\alpha + \frac{\delta^2}{2}} \lambda_s$  to substitute  $\mu_s$ . The model then becomes

$$S_{t} = S_{0} \exp\left[\int_{0}^{t} \left(r + \lambda_{s} - e^{\alpha + \frac{\delta^{2}}{2}}\lambda_{s} - \frac{\sigma_{s}^{2}}{2}\right) ds + \int_{0}^{t} \sigma_{s} dW_{s}^{rn} + \sum_{n=1}^{N_{t}^{(rn)}} \ln(1 + Y_{n})\right].$$

 $W_s^{(rn)}$  is a Brownian motion and  $N_t^{(rn)}$  is a Poisson process whose intensity is  $\lambda_s$ under the probability measure  $\mathbb{P}^{rn}$ . For convenient, we still denote them as  $W_s$  and  $N_t$ . Then, under the probability measure  $\mathbb{P}^{rn}$ , the model is

$$S_t = S_0 \exp\left[\int_0^t \left(r + \lambda_s - e^{\alpha + \frac{\delta^2}{2}}\lambda_s - \frac{\sigma_s^2}{2}\right) ds + \int_0^t \sigma_s dW_s + \sum_{n=1}^{N_t} \ln(1+Y_n)\right].$$

### 3.2 The CPPI strategies

#### 3.2.1 The constant multiple case

Recall that in the jump-diffusion model, the exposure  $e_t$  is equal to the cushion  $C_t$ multiplied by m. The cushion  $C_t$  is the difference between the portfolio value  $V_t$  and the floor  $F_t$  and  $F_t = G \times \exp\{-r(T-t)\}$ . It is possible to have the portfolio value less than the floor, which means that the cushion will be negative and so will be the exposure. Thus short-sell should be allowed. The following proposition describes the portfolio value under this strategy. CPPI would fail if the value of the portfolio falls below the floor. We will measure the failure.

In our strategy the portfolio value  $V_t$  consists of a riskless asset  $V_t - mC_t$  and risky asset  $mC_t$ . i.e.  $V_t = mC_t + (V_t - mC_t)$ 

**Proposition 3.4.** The CPPI portfolio value under the jump-diffusion model defined by (3.2) is

$$V_t = C_0 \exp\left\{\int_0^t \left(r + m(\mu_s - r) - \frac{m\sigma_s^2}{2}\right) ds + \int_0^t m\sigma_s dW_s\right\} \left[\prod_{n=1}^{N_t} (1 + mY_n)\right] + F_t,$$

where

$$C_0 = (V_0 - Ge^{-rT});$$
  
$$F_t = G \times \exp\{-r(T-t)\}.$$

*Proof.* We have

$$V_t = mC_t + (V_t - mC_t)$$
$$= V_t \left(\frac{mC_t}{V_t} + \left(1 - \frac{mC_t}{V_t}\right)\right)$$

and

$$dV_t = V_t \left(\frac{mC_t}{V_{t-}}\frac{dS_t}{S_{t-}} + \left(1 - \frac{mC_t}{V_{t-}}\right)\frac{dB_t}{B_t}\right).$$

Since  $B_s$  is continuous, then  $B_{s-} = B_s$ , we have

$$dC_{t} = d(V_{t} - F_{t})$$

$$= V_{t} \left( \frac{mC_{t-}}{V_{t}} \frac{dS_{t}}{S_{t-}} + \left( 1 - \frac{mC_{t-}}{V_{t}} \right) \frac{dB_{t}}{B_{t}} \right) - F_{t} \frac{dB_{t}}{B_{t}}$$

$$= C_{t-} \left( \frac{mdS_{t}}{S_{t-}} - (m-1)rdt \right)$$

$$= C_{t-} (m(\mu_{t}dt + \sigma_{t}dW_{t} + Y_{t}dN_{t}) - (m-1)rdt)$$

$$= C_{t-} ((r + m(\mu_{t} - r))dt + m\sigma_{t}dW_{t} + mY_{t}dN_{t}).$$
(3.14)

Then

$$C_{t} = C_{0} \exp\left\{\int_{0}^{t} \left(r + m(\mu_{s} - r) - \frac{m^{2}\sigma_{s}^{2}}{2}\right) ds + \int_{0}^{t} m\sigma_{s} dW_{s}\right\} \left[\prod_{n=1}^{N_{t}} (1 + mY_{n})\right].$$

Hence

$$V_{t} = C_{t} + F_{t}$$
  
=  $C_{0} \exp\left\{\int_{0}^{t} \left(r + m(\mu_{s} - r) - \frac{m\sigma_{s}^{2}}{2}\right) ds + \int_{0}^{t} m\sigma_{s} dW_{s}\right\} \left[\prod_{n=1}^{N_{t}} (1 + mY_{n})\right] + F_{t}.$ 

If we substitute  $\mu_s$  by  $r + \lambda_s - e^{\alpha + \frac{\delta^2}{2}} \lambda_s$ , under the probability measure  $\mathbb{P}^{rn}$ , we get the following corollary.

**Corollary 3.5.** In the Merton's model, under the probability measure  $\mathbb{P}^{rn}$ , the CPPI

portfolio value  $V_t$  under our jump-diffusion model is

$$C_0 \exp\left\{\int_0^t \left(r + m(\lambda_s - e^{\alpha + \frac{\delta^2}{2}}\lambda_s) - \frac{m\sigma_s^2}{2}\right) ds + \int_0^t m\sigma_s dW_s\right\} \left[\prod_{n=1}^{N_t} (1 + mY_n)\right] + F_t,$$

where

$$C_0 = (V_0 - Ge^{-rT});$$
  
$$F_t = G \times \exp\{-r(T-t)\}.$$

The expectation and variance of the CPPI portfolio value are deduced in the following two propositions.

**Proposition 3.6.** The expected CPPI portfolio value at time t under the jumpdiffusion model is

$$\mathbb{E}[V_t] = C_0 \exp\left\{\int_0^t (r+m(\mu_s-r))ds\right\} \sum_{k=1}^\infty \frac{e^{-\int_0^t \lambda_s ds} (\int_0^t \lambda_s ds)^k}{k!} \mathbb{E}\left[\prod_{n=1}^k (1+mY_n)\right] + F_t.$$

*Proof.* Because

$$\begin{split} \mathbb{P}\left[\prod_{n=1}^{N_t} (1+mY_n) \le x\right] &= \mathbb{P}\left[\bigcup_{k=1}^{\infty} \left[\prod_{n=1}^{N_t} (1+mY_n) \le x, N_t = k\right]\right] \\ &= \sum_{k=1}^{\infty} \mathbb{P}\left[\prod_{n=1}^{N_t} (1+mY_n) \le x | N_t = k\right] \mathbb{P}[N_t = k] \\ &= \sum_{k=1}^{\infty} \mathbb{P}\left[\prod_{n=1}^{k} (1+mY_n) \le x, N_t = k\right] \mathbb{P}[N_t = k] \\ &= \sum_{k=1}^{\infty} \mathbb{P}\frac{\left[\prod_{n=1}^{k} (1+mY_n) \le x, N_t = k\right]}{\mathbb{P}[N_t = k]} \mathbb{P}[N_t = k] \\ &= \sum_{k=1}^{\infty} \mathbb{P}\frac{\left[\prod_{n=1}^{k} (1+mY_n) \le x\right] \mathbb{P}[N_t = k]}{\mathbb{P}[N_t = k]} \\ &= \sum_{k=1}^{\infty} \mathbb{P}\left[\prod_{n=1}^{k} (1+mY_n) \le x\right] \mathbb{P}[N_t = k] \end{split}$$

$$= \sum_{k=1}^{\infty} \frac{e^{-\int_0^t \lambda_s ds} (\int_0^t \lambda_s ds)^k}{k!} \mathbb{P}\left[\prod_{n=1}^k (1+mY_n) \le x\right],$$

we get

$$\mathbb{E}\left[\prod_{n=1}^{N_t} (1+mY_n)\right] = \sum_{k=1}^{\infty} \frac{e^{-\int_0^t \lambda_s ds} (\int_0^t \lambda_s ds)^k}{k!} \mathbb{E}\left[\prod_{n=1}^k (1+mY_n)\right]$$

and then

$$\mathbb{E}[V_t]$$

$$= C_0 \mathbb{E}\left[\exp\left\{\int_0^t \left(r + m(\mu_s - r) - \frac{m^2 \sigma_s^2}{2}\right) ds + \int_0^t m \sigma_s dW_s\right\}\right]$$

$$\times \mathbb{E}\left[\prod_{n=1}^{N_t} (1 + mY_n)\right] + F_t$$

$$= C_0 \exp\left\{\int_0^t (r + m(\mu_s - r)) ds\right\} \mathbb{E}\left[\prod_{n=1}^{N_t} (1 + mY_n)\right] + F_t$$

$$= C_0 \exp\left\{\int_0^t (r + m(\mu_s - r)) ds\right\} \sum_{k=1}^{\infty} \frac{e^{-\int_0^t \lambda_s ds} (\int_0^t \lambda_s ds)^k}{k!} \mathbb{E}\left[\prod_{n=1}^k (1 + mY_n)\right] + F_t.$$

**Proposition 3.7.** The variance of the CPPI portfolio value at time t under jumpdiffusion model is

$$C_{0}^{2} \exp\left\{\int_{0}^{t} 2(r+m(\mu_{s}-r)+m^{2}\sigma_{s}^{2})ds\right\} \sum_{k=1}^{\infty} \mathbb{E}\left[\prod_{n=1}^{k} (1+mY_{n})\right]^{2} \frac{e^{-\int_{0}^{t} \lambda_{s} ds} (\int_{0}^{t} \lambda_{s} ds)^{k}}{k!} - \left[\exp\left\{\int_{0}^{t} (r+m(\mu_{s}-r))\right\} ds \sum_{k=1}^{\infty} \frac{e^{-\int_{0}^{t} \lambda_{s} ds} (\int_{0}^{t} \lambda_{s} ds)^{k}}{k!} \mathbb{E}\left[\prod_{n=1}^{k} (1+mY_{n})\right]\right]^{2}.$$

*Proof.* Similar to the proof of Prop. 3.6, we have

$$\mathbb{E}\left[\left[\prod_{n=1}^{N_t} (1+mY_n)\right]^2\right] = \frac{e^{-\int_0^t \lambda_s ds} (\int_0^t \lambda_s ds)^k}{k!} \mathbb{E}\left[\left[\prod_{n=1}^k (1+mY_n)\right]^2\right].$$

Thus,

$$\begin{split} &\operatorname{Var}[V_{l}] = \operatorname{Var}[C_{l}] \\ &= C_{0}^{2} \operatorname{Var}\left[ \exp\left\{ \int_{0}^{t} \left( r + m(\mu_{s} - r) - \frac{m^{2}\sigma_{s}^{2}}{2} \right) ds + \int_{0}^{t} m\sigma_{s} dW_{s} \right\} \left[ \prod_{n=1}^{N_{t}} (1 + mY_{n}) \right] \right] \\ &= C_{0}^{2} \mathbb{E}\left[ \exp\left\{ \int_{0}^{t} \left( r + m(\mu_{s} - r) - \frac{m^{2}\sigma_{s}^{2}}{2} \right) ds + \int_{0}^{t} m\sigma_{s} dW_{s} \right\} \left[ \prod_{n=1}^{N_{t}} (1 + mY_{n}) \right] \right]^{2} \\ &- C_{0}^{2} \left( \mathbb{E}\left[ \exp\left\{ \int_{0}^{t} \left( r + m(\mu_{s} - r) - \frac{m^{2}\sigma_{s}^{2}}{2} \right) ds + \int_{0}^{t} m\sigma_{s} dW_{s} \right\} \left[ \prod_{n=1}^{N_{t}} (1 + mY_{n}) \right] \right] \right)^{2} \\ &= C_{0}^{2} \mathbb{E}\left[ \exp\left\{ \int_{0}^{t} \left( r + m(\mu_{s} - r) - \frac{m^{2}\sigma_{s}^{2}}{2} \right) ds + \int_{0}^{t} m\sigma_{s} dW_{s} \right\} \left[ \prod_{n=1}^{N_{t}} (1 + mY_{n}) \right] \right]^{2} \\ &- C_{0}^{2} \mathbb{E}\left[ \exp\left\{ \int_{0}^{t} \left( r + m(\mu_{s} - r) - \frac{m^{2}\sigma_{s}^{2}}{2} \right) ds + \int_{0}^{t} m\sigma_{s} dW_{s} \right\} \left[ \prod_{n=1}^{N_{t}} (1 + mY_{n}) \right] \right]^{2} \\ &- C_{0}^{2} \mathbb{E}\left[ \exp\left\{ \int_{0}^{t} (r + m(\mu_{s} - r) - \frac{m^{2}\sigma_{s}^{2}}{2} \right) ds + 2 \int_{0}^{t} m\sigma_{s} dW_{s} \right\} \mathbb{E}\left[ \prod_{n=1}^{N_{t}} (1 + mY_{n}) \right] \right]^{2} \\ &= C_{0}^{2} \mathbb{E}\left[ \exp\left\{ \int_{0}^{t} 2(r + m(\mu_{s} - r) - m^{2}\sigma_{s}^{2} \right) ds + 2 \int_{0}^{t} m\sigma_{s} dW_{s} \right\} \mathbb{E}\left[ \prod_{n=1}^{N_{t}} (1 + mY_{n}) \right] \right]^{2} \\ &- C_{0}^{2} \mathbb{E}\left[ \exp\left\{ \int_{0}^{t} (r + m(\mu_{s} - r) \right) ds \right\} \sum_{k=1}^{\infty} \frac{e^{-\int_{0}^{t} \lambda_{s} ds} (\int_{0}^{t} \lambda_{s} ds)^{k}}{k!} \mathbb{E}\left[ \prod_{n=1}^{k} (1 + mY_{n}) \right] \right]^{2} \\ &= C_{0}^{2} \exp\left\{ \int_{0}^{t} 2(r + m(\mu_{s} - r) + m^{2}\sigma_{s}^{2}) ds \right\} \sum_{k=1}^{\infty} \mathbb{E}\left[ \prod_{n=1}^{k} (1 + mY_{n}) \right]^{2} \\ &\times \frac{e^{-\int_{0}^{t} \lambda_{s} ds} (\int_{0}^{t} \lambda_{s} ds)^{k}}{k!} \\ &- \left[ \exp\left\{ \int_{0}^{t} (r + m(\mu_{s} - r)) \right\} ds \sum_{k=1}^{\infty} \frac{e^{-\int_{0}^{t} \lambda_{s} ds} (\int_{0}^{t} \lambda_{s} ds)^{k}}{k!} \mathbb{E}\left[ \prod_{n=1}^{k} (1 + mY_{n}) \right]^{2} . \\ & \prod_{n=1}^{k} (1 + mY_{n}) \right]^{2} . \\ \end{bmatrix}$$

Remarks. (1) Another method to calculate the expectation of the portfolio value is

through calculating the characteristic function of

$$\int_{0}^{t} \left( r + m(\mu_{s} - r) - \frac{m\sigma_{s}^{2}}{2} \right) ds + \int_{0}^{t} m\sigma_{s} dW_{s} + \prod_{n=1}^{N_{t}} (1 + mY_{n})$$

In subsection 3.5.2, we will use this method to calculate a similar expectation. (2) For the Merton's and Kou's model,  $\mathbb{E}\left[\prod_{n=1}^{k}(1+mY_n)\right]$  and  $\mathbb{E}\left[\prod_{n=1}^{k}(1+mY_n)\right]^2$ can be calculated and thus expected portfolio can be calculated explicitly. In general, if we assume  $Q_n = \ln(1+Y_n)$  have i.i.d. with density  $f_Q$ ,  $\mathbb{E}\left[\prod_{n=1}^{k}(1+mY_n)\right]$  and  $\mathbb{E}\left[\prod_{n=1}^{k}(1+mY_n)\right]^2$  still can be calculated in terms of the function  $f_Q$ .

The following lemma gives the density function of  $1 + mY_i$ .

**Lemma 3.8.** Let the density function of  $\ln(1+Y_n)$  be  $f_Q(y)$ , then the density function  $f'_Q$  of the random variable  $1 + mY_i$  is

$$f'_Q(z) = f_Q\left(\ln\left(1 + \frac{z-1}{m}\right)\right) \frac{1}{m+z-1}.$$

*Proof.* Since

$$\mathbb{P}(1+mY_i \le z) = \mathbb{P}\left(\ln(1+Y_i) \le \ln\left(1+\frac{z-1}{m}\right)\right)$$
$$= \int_{-\infty}^{\ln(1+\frac{z-1}{m})} f_Q(y) dy,$$

the density  $f'_Q$  of the random variable  $1 + mY_i$  is

$$f'_Q(z) = \frac{d\left(\mathbb{P}(1+mY_i \le z)\right)}{dz} = f_Q\left(\ln\left(1+\frac{z-1}{m}\right)\right)\frac{1}{m+z-1}.$$

	-	-	

Now we can calculate

$$\mathbb{E}\left[\prod_{n=1}^{k} (1+mY_n)\right] = \mathbb{E}\left[\exp\left\{\sum_{n=1}^{k} \ln(1+mY_n)\right\}\right]$$
$$= \int_{\mathbb{R}} \exp\left\{\underbrace{f'_Q * f'_Q * \dots * f'_Q(x)}_{k \text{ items}}\right\} dx$$

and

$$\mathbb{E}\left[\prod_{n=1}^{k} (1+mY_n)^2\right] = \mathbb{E}\left[\exp\left\{\sum_{n=1}^{k} 2\ln(1+mY_n)\right\}\right]$$
$$= \int_{\mathbb{R}} \exp\left\{2\underbrace{f'_Q * f'_Q * \dots * f'_Q(x)}_{k \text{ items}}\right\} dx.$$

#### 3.2.2 The case when the multiple is a function of time

Let  $m_t$  be the multiple at time t. We have similar results:

**Proposition 3.9.** When the multiple is a function of time the CPPI portfolio value under the jump-diffusion model is

$$V_t = C_0 \exp\left\{\int_0^t \left(r + m_s(\mu_s - r) - \frac{m_s^2 \sigma_s^2}{2}\right) ds + \int_0^t m_s \sigma_s dW_s\right\} \left[\prod_{n=1}^{N_t} (1 + m_n Y_n)\right] + F_t,$$

where  $m_n$  is obtained from  $m_t$  by the formula

$$m_n = m_{T_n},$$

where  $T_0 = 0$ .

Proposition 3.10. When the multiple is a function of time the expected CPPI port-

folio value under jump-diffusion model is

$$C_0 \exp\left\{\int_0^t (r+m_s(\mu_s-r))ds \sum_{k=1}^\infty \frac{e^{-\int_0^t \lambda_s ds} (\int_0^t \lambda_s ds)^k}{k!} \mathbb{E}\left[\prod_{n=1}^k (1+m_n Y_n)\right]\right\} + F_t.$$

**Proposition 3.11.** When the multiple is a function of time the variance of the CPPI portfolio value under jump-diffusion model is

$$C_0^2 \exp\left\{\int_0^t 2\left(r + m_s(\mu_s - r) + m_s^2 \sigma_s^2\right) ds\right\}$$
  
 
$$\times \sum_{k=1}^\infty \mathbb{E}\left[\prod_{n=1}^k (1 + m_n Y_n)\right]^2 \frac{e^{-\int_0^t \lambda_s ds} (\int_0^t \lambda_s ds)^k}{k!}$$
  
 
$$-\left[\exp\left\{\int_0^t (r + m_s(\mu_s - r)) ds\right\} \sum_{k=1}^\infty \frac{e^{-\int_0^t \lambda_s ds} (\int_0^t \lambda_s ds)^k}{k!} \mathbb{E}\left[\prod_{n=1}^k (1 + m_n Y_n)\right]\right]^2.$$

Here we consider a special form of  $m_t$ . Let  $0 = t_0 < t_1 < t_2 < ... < t_{n-1} < t_n = T$ where  $t_{k+1} - t_k = \frac{T}{n}$  for k = 0, ..., n - 1. We reconsider the multiple only at time  $t_i$ which i = 0, 1, ..., n. Let

$$m_0 = \eta + 1$$
  

$$m_{t_k} = \eta + e^{aln(S_{t_k}/S_{t_{k-1}})} \text{ When } k \ge 1$$
  

$$m_t = m_{t_k} \text{ When } t \in [t_k, t_{k+1})$$

*Remarks.* The above is called an EPPI strategy, a special case of which would be when the multiple is a function of time. However, since CPPI is a common term in financial mathematics, we still refer the above EPPI as a special case of CPPI.

## 3.3 The CPPI portfolio as a hedging tool

We have proved that the portfolio value is

$$V_t = C_0 \exp\left\{\int_0^t \left(r + m_s(\mu_s - r) - \frac{m_s^2 \sigma_s^2}{2}\right) ds + \int_0^t m_s \sigma_s dW_s\right\} \left[\prod_{n=1}^{N_t} (1 + mY_n)\right] + F_t$$

The following lemma is deduced from the Ito formula and will be used to prove some later theorems.

**Lemma 3.12.** Let  $v(x,t) \in C^{1,2}([0,T] \times \mathbb{R})$  and bounded at infinity. Then the conditional expectation of the composition process v(t,x(t)) satisfies

$$\mathbb{E}[v(t,S_t)|S(0) = S_0] = v(0,S_0) + \mathbb{E}\left[\int_0^t \left(\frac{\partial v}{\partial t} + \mu_t S_{u-}\frac{\partial v}{\partial x} + \frac{1}{2}(\sigma_u S_{u-})^2 \frac{\partial^2 v}{\partial x^2}\right)(u,S_u) + \lambda_u(v(u,S_{u-} + S_{u-}Y_u) - v(u,S_u))dN_u|S(0) = S_0\right].$$

*Proof.* Our risky asset  $S_t$  is given by

$$dS_t = S_{t-}(\mu_t dt + \sigma_t dW_t + Y_t dN_t).$$

By the Ito chain rule,

$$dv(t, S_t) = \left(\frac{\partial v}{\partial t} + \mu_t S_{t-} \frac{\partial v}{\partial x} + \frac{1}{2} (\sigma_t S_{t-})^2 \frac{\partial^2 v}{\partial x^2}\right) (t, S_t) dt + S_{t-} \sigma_t \frac{\partial v}{\partial x} (t, S_t) dW_t + (v(t, S_{t-} + S_{t-} Y_t) - v(t, S_t)) dN_t.$$

When expressed in integral form, we have,

$$v(t, S_t) = v(0, S_0) + \int_0^t \left(\frac{\partial v}{\partial t} + \mu_t S_{u-} \frac{\partial v}{\partial x} + \frac{1}{2} (\sigma_u S_{u-})^2 \frac{\partial^2 v}{\partial x^2} \right) (u, S_u) du$$
$$+ \int_0^t S_{u-} \sigma_u \frac{\partial v}{\partial x} dW_u + (v(u, S_{u-} + S_{u-}Y_u) - v(u, S_{u-})) dN_u.$$

By taking conditional expectation on both sides, we have

$$\mathbb{E}[v(t,S_t)|S(0) = S_0] = v(0,S_0) + \mathbb{E}\left[\int_0^t \left(\frac{\partial v}{\partial t} + \mu_t S_{u-}\frac{\partial v}{\partial x} + \frac{1}{2}(\sigma_u S_{u-})^2 \frac{\partial^2 v}{\partial x^2}\right)(u,S_u) + \lambda_u(v(u,S_{u-} + S_{u-}Y_u) - v(u,S_u))dN_u|S(0) = S_0\right].$$

*Remarks.* The term  $(v(t, S_{t-} + S_{t-}Y_t) - v(t, S_t))dN_t$  describes the difference of the portfolio value as a functional of  $S_t$  when a jump occurs.

In section 4 of [16], the CPPI portfolio is utilized as a hedging tool under the Black-scholes model. See also [26]. In this section, we generalize the above result to our jump-diffusion case.

#### 3.3.1 PIDE Approach

Suppose that  $\eta = g(S_T)$  is a contingent claim that the portfolio's manager is aiming to have at maturity. Can the CPPI portfolio be converted into a synthetic derivative with pay-off specified by  $\eta = g(S_T)$ ?

**Theorem 3.13.** If  $g : \mathbb{R} \to \mathbb{R}$  is sufficiently smooth, there exists a unique self-financed  $g(S_T)$  hedging CPPI portfolio V, defined by

$$V_t = v(t, S_t) \quad t \in [0, T]$$
 (3.15)

where  $v \in C^{1,2}([0,T] \times \mathbb{R})$  is the unique solution of the following partial integrodifferential equations (PIDE).

$$\frac{\partial u}{\partial t}(t,s) + (\mu_t s)\frac{\partial u}{\partial x}(t,s) + \frac{1}{2}(s\sigma_t)^2\frac{\partial^2 u}{\partial x^2}(t,s) - ru(t,s) = 0$$
(3.16)

$$sz\frac{\partial u}{\partial x}(t,s) = u(t,s+sz) - u(t,s)$$

$$54$$

$$(3.17)$$

$$u(T,s) = g(s), \quad (t,s) \in [0,T] \times \mathbb{R}), \quad u \in C^{1,2}([0,T] \times \mathbb{R})$$
(3.18)

In particular the CPPI portfolio's gearing factor is given by:

$$m_t = \frac{\frac{\partial u}{\partial x}(t, S_t)S_{t-}}{V_{t-} - F_t}, \quad t \in [0, T].$$
(3.19)

*Proof.* For V to be a self-financed  $g(S_T)$ -hedging portfolio, it is enough to ensure that at maturity time we have

$$V_T = g(S_T), \quad a.s..$$

Choose a map  $v \in C^{1,2}([0,T] \times \mathbb{R})$  and set  $V_t = v(t, S_t)$   $(t \in [0,T])$ . Then  $v(T, S_T) = g(S_T)$  P-a.s., therefore

$$v(T,s) = g(s), \quad \forall s \in \mathbb{R}.$$

Second by Ito's chain rule,

$$dv(t, S_t) = \left(\frac{\partial v}{\partial t} + \mu_t S_{t-} \frac{\partial v}{\partial x} + \frac{1}{2} (\sigma_t S_{t-})^2 \frac{\partial^2 v}{\partial x^2}\right) (t, S_t) dt + S_{t-} \sigma_t \frac{\partial v}{\partial x} (t, S_t) dW_t + (v(t, S_{t-} + S_{t-} Y_t) - v(t, S_{t-})) dN_t.$$

Now  $V_t$  satisfies

$$dV_t = dC_t + dF_t$$
  
=  $(V_{t-} - F_t)(r + m_t(\mu_t - r))dt + rF_t dt + (V_{t-} - F_t)m_t\sigma_t dW_t + (V_{t-} - F_t)m_tY_t dN_t$   
=  $(rV_{t-} + (V_{t-} - F_t)m_t(\mu_t - r))dt + (V_{t-} - F_t)m_t\sigma_t dW_t + (V_{t-} - F_t)m_tY_t dN_t.$ 

A comparison of the above two equations implies that

$$m_t = \frac{\frac{\partial u}{\partial x}(t, S_t)S_{t-}}{V_{t-} - F_{t_{55}}}, \quad t \in [0, T]$$

and

$$\begin{aligned} &\frac{\partial u}{\partial t}(t,s) + (\mu_t s)\frac{\partial u}{\partial x}(t,s) + \frac{1}{2}(s\sigma_t)^2\frac{\partial^2 u}{\partial x^2}(t,s) - ru(t,s) = 0;\\ &sz\frac{\partial u}{\partial x}(t,s) = u(t,s+sz) - u(t,s). \end{aligned}$$

In a financial turmoil, the portfolio's manager acting on the leverage regime may convert the CPPI portfolio in a suitable synthetic derivative whose price is specified by (3.15)-(3.18). Moreover the required dynamic gearing factor (multiple) can be easily determined, using (3.19). This is the PIDE/PDE approach hedging. Another observation that reveals to be central in the analysis of possible portfolio's hedges is that at any time of the financial horizon the CPPI portfolio value may be regarded as a standard risky asset and therefore as an underlying for any convenient

contingent claim:

**Theorem 3.14.** Under the risk neutral measure  $\mathbb{Q}$ , the discounted CPPI portfolio's value  $\{V_t\}_{t \in [0,T]}$ 

$$\tilde{V}_t = e^{-rt} V_t, \quad t \in [0, T]$$
(3.20)

is a martingale.

Proof. In the proof of Theorem 3.13, we have deduced

$$dV_t = (rV_{t-} + (V_{t-} - F_t)m_t(\mu_t - r))dt + (V_{t-} - F_t)m_t\sigma_t dW_t + (V_{t-} - F_t)m_tY_t dN_t,$$

thus we have

 $dV_t$ 

$$= (rV_{t-} + (V_{t-} - F_t)m_t(\mu_t - r))dt + (V_{t-} - F_t)m_t\sigma_t(dW_t^{\mathbb{Q}} + \theta_t dt) + (V_{t-} - F_t)m_tY_tdN_t = (rV_{t-} + (V_{t-} - F_t)m_t(\mu_t - r + \theta_t))dt + (V_{t-} - F_t)m_t\sigma_tdW_t^{\mathbb{Q}} + (V_{t-} - F_t)m_tY_tdN_t = (rV_{t-} + (V_{t-} - F_t)m_t\sigma_t)dW_t^{\mathbb{Q}} + (V_{t-} - F_t)m_t(-Y_t\psi_t\lambda_t)dt + (V_{t-} - F_t)m_tY_tdN_t = (rV_{t-} + (V_{t-} - F_t)m_t\sigma_t)dW_t^{\mathbb{Q}} + (V_{t-} - F_t)m_tY_tdM_t^{\mathbb{Q}}.$$

Integration by parts implies that

$$d\tilde{V}_t = de^{-rt}V_t = -re^{-rt}V_t dt + e^{-rt}dV_t$$
  
=  $e^{-rt}((V_{t-} - F_t)m_t\sigma_t dW_t^{\mathbb{Q}} + (V_{t-} - F_t)m_tY_t dM_t^{\mathbb{Q}}).$ 

Thus,  $\tilde{V}_t$  is a Q-martingale.

If we substitute  $\mu_s$  by  $r + \lambda_s - e^{\alpha + \frac{\delta^2}{2}} \lambda_s$ , under the probability measure  $\mathbb{P}^{rn}$ , we get the following corollary.

**Corollary 3.15.** In Merton's model, under probability measure  $\mathbb{P}^{rn}$ , the discounted CPPI portfolio's value  $\{V_t\}_{t \in [0,T]}$ 

$$\tilde{V}_t = e^{-rt} V_t, \quad t \in [0, T]$$

is a martingale.

*Proof.* We have

$$dV_t = (rV_{t-} + (V_{t-} - F_t)m_t(\mu_t - r))dt + (V_{t-} - F_t)m_t\sigma_t dW_t + (V_{t-} - F_t)m_tY_t dN_t$$
  
=  $\left(rV_{t-} + (V_{t-} - F_t)m_t\left(\lambda_t - e^{\alpha + \frac{\delta^2}{2}}\lambda_t\right)\right)dt + (V_{t-} - F_t)m_t\sigma_t dW_t$   
57

$$+(V_{t-}-F_t)m_tY_tdN_t.$$

Thus

$$\begin{split} d\tilde{V}_t &= de^{-rt}V_t = -re^{-rt}V_t dt + e^{-rt} dV_t \\ &= -re^{-rt}V_t dt + e^{-rt} \left( rV_{t-} + (V_{t-} - F_t)m_t \left(\lambda_t - e^{\alpha + \frac{\delta^2}{2}}\lambda_t\right) \right) dt \\ &+ (V_{t-} - F_t)m_t \sigma_t dW_t + (V_{t-} - F_t)m_t Y_t dN_t) \\ &= e^{-rt}(V_{t-} - F_t)m_t \left(\lambda_t - e^{\alpha + \frac{\delta^2}{2}}\lambda_t\right) dt + (V_{t-} - F_t)m_t \sigma_t dW_t \\ &+ (V_{t-} - F_t)m_t Y_t dN_t) \\ &= e^{-rt}(V_{t-} - F_t)m_t \left(\lambda_t - e^{\alpha + \frac{\delta^2}{2}}\lambda_t + \lambda_t Y_t\right) dt \\ &+ (V_{t-} - F_t)m_t \sigma_t dW_t + (V_{t-} - F_t)m_t Y_t (dN_t - \lambda_t dt)). \end{split}$$

Since  $dN_t - \lambda_t dt$  is a martingale and

$$\mathbb{E}[Y_t] = \mathbb{E}\left(e^{\ln(1+Y_n)} - 1\right) = e^{\alpha + \frac{\delta^2}{2}} - 1,$$

we get  $\mathbb{E}[(\lambda_t - e^{\alpha + \frac{\delta^2}{2}} \lambda_t)] = 0$ , so we prove  $\tilde{V}_t$  is a  $\mathbb{P}^{nr}$ -martingale.  $\Box$ 

Given any claim  $\eta = g(V_T)$ , which is a function of the terminal portfolio's price, there exists a unique self-financed  $\eta = g(V_T)$ -hedging strategy:

**Theorem 3.16.** Let  $g : \mathbb{R} \to \mathbb{R}$  sufficiently smooth. Then there exists a unique  $\eta = g(V_T)$ -hedging self-financed trading strategy  $(U, \beta)$  defined as

$$U_t = u(t, V_t), \quad \beta_{t-} = \frac{\partial u}{\partial x}(t, V_t), \quad t \in [0, T],$$

where  $u \in C^{1,2}([0,T] \times \mathbb{R})$  is the unique solution of the PIDE.

$$\frac{\partial u}{\partial t}(t,v) + rv\frac{\partial u}{\partial x}(t,v) + \frac{1}{2}m^2\sigma_t^2(v-f)^2\frac{\partial^2 u}{\partial x^2}(t,v) - ru(t,v) = 0 \qquad (3.21)$$

$$mz(v-f)\frac{\partial u}{\partial x}(t,v) = u(t,v+m(v-f)z) - u(t,v)$$
(3.22)

with the final condition u(T, v) = g(v).

*Proof.* Consider an asset  $\{V_t\}_{t \in [0,T]}$ , and pick a self-financed  $g(V_T)$  hedging strategy space  $(U_t, \beta_t)_{t \in [0,T]}$  by setting:

$$dU_t = \beta_{t-}dV_t + (U_{t-} - \beta_{t-}V_{t-})rdt$$

and

$$U_T = g(V_T) \quad a.s.$$

Since

$$dV_t = (rV_{t-} + (V_{t-} - F_t)m_t(\mu_t - r))dt + (V_{t-} - F_t)m_t\sigma_t dW_t + (V_{t-} - F_t)m_tY_t dN_t,$$

the hedging portfolio's equation may be rewritten as:

$$dU_t = \beta_{t-}(rV_{t-} + (V_{t-} - F_t)m_t(\mu_t - r)dt + (V_{t-} - F_t)m_t\sigma_t dW_t + (V_{t-} - F_t)m_tY_t dN_t) + (U_{t-} - \beta_{t-}V_{t-})rdt = (rU_{t-} + \beta_{t-}(V_{t-} - F_t)m(\mu_t - r))dt + \beta_{t-}(V_{t-} - F_t)m_t\sigma_t dW_t + \beta_{t-}(V_{t-} - F_t)m_tY_t dN_t).$$

Pick  $u \in C^{1,2}([0,T] \times \mathbb{R})$  and set  $U_t = u(t, V_t)$ , for  $t \in [0,T]$ . For any  $t \in [0,T]$ , the Ito's formula implies that:

$$du(t, V_t) = \frac{\partial u}{\partial t}(t, V_t) + (rV_{t-} + m(\mu_t - r)(V_{t-} - F_t))\frac{\partial u}{\partial x}(t, V_t)$$

$$+\frac{1}{2}(m\sigma_t)^2(V_{t-}-F_t)^2\frac{\partial^2 u}{\partial x^2}(t,V_t)dt + m\sigma_t(V_{t-}-F_t)\frac{\partial u}{\partial x}(t,V_t)dW_t + (u(t,V_{t-}+m(V_{t-}-F_t)Y_t)-u(t,V_{t-}))dN_t.$$

A comparison between the above two equations implies in particular

$$\beta_{t-} = \frac{\partial u}{\partial x}(t, V_t)$$

and

$$\begin{aligned} \frac{\partial u}{\partial t}(t,v) + (rv + m(\mu_t - r)(v - f))\frac{\partial u}{\partial x}(t,v) + \frac{1}{2}m^2\sigma_t^2(v - f)^2\frac{\partial^2 u}{\partial x^2}(t,v) \\ = ru(t,v) + m(\mu_t - r)(v - f)\frac{\partial u}{\partial x}(t,v). \end{aligned}$$

Thus

$$\frac{\partial u}{\partial t}(t,v) + rv\frac{\partial u}{\partial x}(t,v) + \frac{1}{2}m^2\sigma_t^2(v-f)^2\frac{\partial^2 u}{\partial x^2}(t,v) - ru(t,v) = 0$$

and

$$mz(v-f)\frac{\partial u}{\partial x}(t,v) = u(t,v+m(v-f)z) - u(t,v)$$

with the final condition u(T, v) = g(v).

The rationale in constructing self-financed trading strategies that hedge the CPPI portfolio's terminal price, is that there are contingent claims particularly useful to control both the closing-out-effect and the gap risk. As an example consider the case of a Vanilla option based on the CPPI portfolio's value. For instance being long in an at-the-money put option on the portfolio with a strike at least equal to the protection required is a natural way to hedge gap risk. Similarly being long in an at-the-money call option on the portfolio is a natural way to invest in a CPPI's portfolio preserving 60

the capability to not purse forward the investment in the case of closed out.

#### 3.3.2 Fourier Transformation Approach

[14] and [54] do research on how to use Fourier transform to value option when we know the characteristic function. We refer to their results to value our CPPI option. Under the martingale measure  $\mathbb{Q}$ , the discounted stock price  $\tilde{S}_t = e^{-rt}S_t$  is a martingale. Consider the European option with the pay-off as the function of  $\tilde{S}_T$ , i.e.  $G(\tilde{S}_T)$ , and denote by h its log-payoff function  $G(e^x) \equiv g(x)$  and by  $\Phi$  the characteristic function of  $\ln(\tilde{S}_t)$ . Proposition 10 in [74] states the following result.

**Proposition 3.17.** Suppose that there exists  $R \neq 0$  such that  $h(x)e^{Rx}$  has finite variance on  $\mathbb{R}$ ,  $h(x)e^{-Rx} \in L^1(\mathbb{R})$ ,  $\mathbb{E}^{\mathbb{Q}}\left[e^{RX_{T-t}}\right] < \infty$  and  $\int_{\mathbb{R}} \frac{|\Phi_{T-t}(u-iR)|}{1+|u|} du < \infty.$ 

Then the price at time t of the European option with pay-off function G satisfies

$$P(t, \tilde{S}_t) := e^{-r(T-t)} \mathbb{E}^{\mathbb{Q}} \left[ G(\tilde{S}_T) | \mathfrak{F}_t \right]$$

$$= \frac{e^{-r(T-t)}}{2\pi} \int_{\mathbb{R}} \hat{h}(u+iR) \Phi_{T-t}(-u-iR) \hat{\tilde{S}}_t^{R-iu} du,$$
(3.23)

where  $\hat{h}(u) := \int_{\mathbb{R}} e^{iux} h(x) ds$ .

We are interested in considering the European option whose pay-off a function depends on the discounted CPPI portfolio  $\tilde{V}_T$ , i.e.  $G(\tilde{V}_T)$ . Since  $V_t = C_t + F_t$  and  $C_t$ are in exponential forms, it is more convenient to treat it as a function of the cushion  $\tilde{C}_T$ . Let  $G_2(e^x) = h_2(x)$  and

$$\varepsilon_t = C_0 \exp\left\{\int_0^t \left(m(\mu_s - r) - \frac{m^2 \sigma_s^2}{2}\right) ds + \int_0^t m \sigma_s dW_s\right\} \left[\prod_{n=1}^{N_t} (1 + mY_n)\right]. (3.24)$$

In subsection 4.2.2, we will show that the characteristic function  $\phi_t(u)$  of  $\ln(\frac{\varepsilon_t}{C_0})$  is

given by

$$\phi_t(u) = \exp\left\{i\left(\int_0^t \left(m(\mu_s - r) - \frac{m^2\sigma_s^2}{2}\right)ds\right)u - \frac{1}{2}\left(\int_0^t m\sigma_s^2ds\right)u^2\right\}$$
  
 
$$\times \exp\left\{t\lambda \int_{\mathbb{R}} \left(e^{iux} - 1\right)f_Q\left(\ln\left(1 + \frac{e^x - 1}{m}\right)\right)\frac{e^x}{m + e^x - 1}dx\right\}.$$

Thus the characteristic function of  $\ln(C_T)$  is  $C_0\phi_t(u)$ . We then have

**Proposition 3.18.** Suppose that there exists  $R \neq 0$  such that  $h_2(x)e^{Rx}$  has finite variance on  $\mathbb{R}$ ,  $h_2(x)e^{-Rx} \in L^1(\mathbb{R})$ ,  $\mathbb{E}^{\mathbb{Q}}\left[e^{RX_{T-t}}\right] < \infty$  and  $\int_{\mathbb{R}} \frac{|C_0\phi_{T-t}(u-iR)|}{1+|u|} du < \infty$ .

Then the price at time t of the European option with pay-off function  $G_2$  satisfies

$$P(t, \tilde{V}_t) := e^{-r(T-t)} \mathbb{E}^{\mathbb{Q}} \left[ G_2(\tilde{C}_T) | \mathfrak{F}_t \right]$$

$$= \frac{e^{-r(T-t)}}{2\pi} \int_{\mathbb{R}} \hat{h}_2(u+iR) C_0 \phi_{T-t}(-u-iR) \hat{C}_t^{R-iu} du, \qquad (3.25)$$

where  $\hat{h}(u) := \int_{\mathbb{R}} e^{iux} h(x) ds$ .

*Remarks.* The European call option has pay-off  $G_2(C_T) = (C_T + F_T - K)^+$ , therefore, we have for all R > 1

$$\hat{h}_2(u+iR) = \frac{(K-F_T)^{iu+1-R}}{(R-iu)(R-1-iu)}$$

#### 3.3.3 Martingale Approach

It is possible to obtain a Black-Sholes type formula for pricing Vanilla options based on the CPPI portfolio:

We first consider the general case. We assume that the  $\ln(1 + Y_i)$  are i.i.d. with common density function  $f_Q$ .
**Proposition 3.19.** Let the density of  $\ln(1+Y_i)$  be  $f_Q(x)$  and the density function of

$$\ln(L_t) = \int_0^t \left[ (1 - \psi_s)\lambda_s - \frac{1}{2}\theta_s^2 \right] ds + \int_0^t \theta_s dW_s + \int_0^t \ln\psi_s dN_s$$

be  $f^{L_t}$ , where  $L_t$  is defined by (3.9). Then the vanilla call/put option on the whole CPPI portfolio's value at maturity is completely determined by:

$$= \sum_{k=0}^{\infty} \left( \frac{e^{-\int_0^t \psi_s \lambda_s ds} (\int_0^t \psi_s \lambda_s ds)^k}{k!} \right) \times \int_{\varsigma}^{\infty} \left( C_0 e^x + F_0 - e^{-rT} K \right) p^{(k)} dx$$

and

$$Put(0, v, T, K) = \sum_{k=0}^{\infty} \left( \frac{e^{-\int_0^t \psi_s \lambda_s ds} (\int_0^t \psi_s \lambda_s ds)^k}{k!} \right) \times \int_{\varsigma}^{\infty} (-C_0 e^x - F_0 + e^{-rT} K) p^{(k)} dx,$$

where  $K > F_T$  and

$$p^{(k)} = f_1 * \underbrace{f_{Q'}^{\mathbb{Q}} * \dots * f_{Q'}^{\mathbb{Q}}}_{k \ terms},$$

where  $f_{Q'}$  and  $f_{Q'}^{\mathbb{Q}}$  have the following relation:

$$\int_{\mathbb{R}} \exp\left\{iuf_{Q'}^{\mathbb{Q}}(z)\right\} dz = \int_{\mathbb{R}} \exp\left\{\left[f_{Q'}(\frac{z}{iu})\frac{z}{iu}\right] * f^{L_{T}}(z)\right\} dz$$

and  $f_1$  is the density function of the normal distribution

$$\mathcal{N}\left(\cdot, \int_0^T \left( \left(m(-Y\psi_s\lambda_s) - \frac{m^2\sigma_s^2}{2}\right) ds, \int_0^T m\sigma_s dW_s^{\mathbb{Q}} \right)$$
  
and  $\varsigma = \ln\left(\frac{e^{-rT}K - F_0}{C_0}\right).$ 

a

*Proof.* Consider the process:

$$\begin{aligned} V_t &= C_0 \exp\left\{\int_0^t \left(r + m(\mu_s - r) - \frac{m_s^2 \sigma_s^2}{2}\right) ds + \int_0^t m \sigma_s dW_s\right\} \left[\prod_{n=1}^{N_t} (1 + mY_n)\right] + F_t \\ &= C_0 \exp\left\{\int_0^t \left(r + m(\mu_s - r) + m \sigma_s \theta_s - \frac{m^2 \sigma_s^2}{2}\right) ds + \int_0^t m \sigma_s dW_s^{\mathbb{Q}}\right\} \\ &\times \left[\prod_{n=1}^{N_t} (1 + mY_n)\right] + F_t \\ &= C_0 \exp\left\{\int_0^t \left(r - mY_s \psi_s \lambda_s - \frac{m^2 \sigma_s^2}{2}\right) ds + \int_0^t m \sigma_s dW_s^{\mathbb{Q}}\right\} \left[\prod_{n=1}^{N_t} (1 + mY_n)\right] + F_t. \end{aligned}$$

In the case of  $N_T = k$ , we denote

$$L^{(k)} = e^{-rT} \frac{V_T^k - F_T}{C_0} = \exp\left\{\int_0^T \left(m(-Y\psi_s\lambda_s) - \frac{m^2\sigma_s^2}{2}\right) ds + \int_0^T m\sigma_s dW_s^Q + \left[\sum_{n=1}^k \ln(1+mY_n)\right]\right\}.$$

(see the remark (3) below the proof.) Because

$$\mathbb{P}(\ln(1+mY_i) \le z) = \mathbb{P}\left(\ln(1+Y_i) \le \ln\left(1+\frac{e^z-1}{m}\right)\right) = \int_{-\infty}^{\ln\left(1+\frac{e^z-1}{m}\right)} f_Q(y) dy,$$

the density function  $f_{Q'}$  of the random variable  $\ln(1 + mY_i)$  under the probability measure  $\mathbb{P}$  is

$$f_{Q'}(z) = \frac{d(\mathbb{P}(\ln(1+mY_i) \le z))}{dz} = f_Q\left(\ln\left(1+\frac{e^z-1}{m}\right)\right)\frac{e^z}{m+e^z-1}.$$

Suppose the density function of  $\ln(1 + mY_i)$  under the measure  $\mathbb{Q}$  is  $f_{Q'}^{\mathbb{Q}}$ . By the properties of the Radon-Nikodym derivative and the characteristic function, we have

$$\mathbb{E}^{\mathbb{Q}}[\exp\{iu\ln(1+mY_i)\}] = \mathbb{E}[\exp\{iu\ln(1+mY_i)\}L_T]$$
$$= \mathbb{E}[\exp\{iu\ln(1+mY_i)+\ln L_T\}]$$
64

Under  $\mathbb{P}$ , the density function of  $iu \ln(1+mY_i)$  is  $f_{Q'}\left(\frac{z}{iu}\right)\frac{z}{iu}$ , thus the density function of  $iu \ln(1+mY_i) + \ln L_T$  under  $\mathbb{P}$  is

$$\left[f_{Q'}\left(\frac{z}{iu}\right)\frac{z}{iu}\right]*f^{L_T}(z)$$

and thus  $f_{Q'}$  and  $f_{Q'}^{\mathbb{Q}}$  have the following relation:

$$\int_{\mathbb{R}} \exp\left\{iuf_{Q'}^{\mathbb{Q}}(z)\right\} dz = \int_{\mathbb{R}} \exp\left\{\left[f_{Q'}\left(\frac{z}{iu}\right)\frac{z}{iu}\right] * f^{L_{T}}(z)\right\} dz.$$

Since

$$\int_0^T \left( m(-Y\psi_s\lambda_s) - \frac{m^2\sigma_s^2}{2} \right) ds + \int_0^T m\sigma_s dW_s^{\mathbb{Q}}$$
  
~  $\mathcal{N}\left(\cdot, \int_0^T \left( m(-Y\psi_s\lambda_s) - \frac{m^2\sigma_s^2}{2} \right) ds, \int_0^T m\sigma_s dW_s^{\mathbb{Q}} \right),$ 

we denote its density function by

$$f_1(x) = \phi\left(x, \int_0^T \left(m(-Y\psi_s\lambda_s) - \frac{m^2\sigma_s^2}{2}\right)ds, \int_0^T m\sigma_s dW_s^{\mathbb{Q}}\right)$$

under the probability measure  $\mathbb{Q}$  where  $\phi(x, m, v) = \frac{1}{\sqrt{2\pi v^2}} e^{-\frac{(x-m)^2}{2v^2}}$ . Then the density function  $p^{(k)}(x)$  of  $L^{(k)}$  is

$$p^{(k)} = f_1 * \underbrace{f_{Q'}^{\mathbb{Q}} * \dots * f_{Q'}^{\mathbb{Q}}}_{k \text{ terms}}.$$

We have

$$\mathbb{E}^{\mathbb{Q}}\left(e^{-rT}(V_T^{(k)}-K)^+\right) = \int_{\varsigma}^{\infty} \left(C_0 e^x + F_0 - e^{-rT}K\right) p^{(k)} dx,$$

where

$$\varsigma = \ln\left(\frac{e^{-rT}K - F_0}{C_0}\right),\,$$

 $\operatorname{thus}$ 

$$\operatorname{Call}(0, v, T, K) = \mathbb{E}^{\mathbb{Q}} \left( e^{-rT} (V_T - K)^+ \right) = \sum_{k=0}^{\infty} \mathbb{E}^{\mathbb{Q}} \left( e^{-rT} (V_T^{(k)} - K)^+ \right) \left( \frac{e^{-\int_0^t \psi_s \lambda_s ds} (\int_0^t \psi_s \lambda_s ds)^k}{k!} \right)$$
$$= \sum_{k=0}^{\infty} \left( \frac{e^{-\int_0^t \psi_s \lambda_s ds} (\int_0^t \psi_s \lambda_s ds)^k}{k!} \right) \times \int_{\varsigma}^{\infty} (C_0 e^x + F_0 - e^{-rT} K) p^{(k)} dx.$$

Similarly,

$$\mathbb{E}^{\mathbb{Q}}\left(e^{-rT}(K-V_T^{(k)})^+\right) = \int_{\varsigma}^{\infty} (-C_0 e^x - F_0 + e^{-rT}K)p^{(k)}dx$$

and

$$\operatorname{Put}(0, v, T, K) = \mathbb{E}^{\mathbb{Q}} \left( e^{-rT} (K - V_T)^+ \right) = \sum_{k=0}^{\infty} \mathbb{E}^{\mathbb{Q}} \left( e^{-rT} (K - V_T^{(k)})^+ \right) \left( \frac{e^{-\int_0^t \psi_s \lambda_s ds} (\int_0^t \psi_s \lambda_s ds)^k}{k!} \right)$$
$$= \sum_{k=0}^{\infty} \left( \frac{e^{-\int_0^t \psi_s \lambda_s ds} (\int_0^t \psi_s \lambda_s ds)^k}{k!} \right) \times \int_{\varsigma}^{\infty} (-C_0 e^x - F_0 + e^{-rT} K) p^{(k)} dx.$$

*Remarks.* (1) The expression is not very explicit since they contain measure transformations and convolutions.

(2)When  $\mathbb{Q}$  is the risk neutral measure, the price of a Vanilla call option is given by

Call
$$(t, v, T, K) = E^{\mathbb{Q}} \left[ e^{-r(T-t)} (V_T^{t,v} - K)^+ \right] = E^{\mathbb{Q}} \left[ e^{-r(T-t)} (V_T - K)^+ | V_t = v \right],$$

for any  $t \in [0, T]$ . The CPPI portfolio's value  $\{V_t\}$  is a Markov process so that

$$\operatorname{Call}(t, v, T, K) = \operatorname{Call}(0, v, T - t, K), \text{ for } t \in [0, T]$$

and it is sufficient to cover the case of the Vanilla call option's price at zero.

(3) The value of  $1 + mY_n$  might be negative, in this case  $\ln(1 + mY_n)$  is an imaginary number.

**Corollary 3.20.** In Merton's Model and under the probability measure  $\mathbb{P}^{rn}$ , let the density of  $\ln(1 + Y_i)$  be  $\phi(x, \alpha, \delta^2)$ . Then the Vanilla call/put option on the whole CPPI portfolio's value at maturity is completely determined by

$$Call(0, v, T, K) = \sum_{k=0}^{\infty} \left( \frac{e^{-\int_0^t \lambda_s ds} (\int_0^t \lambda_s ds)^k}{k!} \right) \times \int_{\varsigma}^{\infty} \left( C_0 e^x + F_0 - e^{-rT} K \right) p^{(k)} dx$$

and

$$Put(0, v, T, K) = \sum_{k=0}^{\infty} \left( \frac{e^{-\int_0^t \lambda_s ds} (\int_0^t \lambda_s ds)^k}{k!} \right) \times \int_{\varsigma}^{\infty} \left( -C_0 e^x - F_0 + e^{-rT} K \right) p^{(k)} dx,$$

where  $K > F_T$  and

$$p^{(k)} = f_1 * \underbrace{f_{Q'} * \dots * f_{Q'}}_{k \ terms},$$

$$f_{Q'}(z) = \phi\left(\ln\left(1 + \frac{e^z - 1}{m}\right), \alpha, \,\delta^2\right) \frac{e^z}{m + e^z - 1},$$

and  $f_1$  is the density function of the normal distribution

$$\mathcal{N}\left(\cdot, \int_0^T \left(m\left(\lambda_s - e^{\alpha + \frac{\delta^2}{2}\lambda_s}\right) - \frac{m\sigma_s^2}{2}\right) ds, \int_0^T m\sigma_s dW_s\right)$$

and  $\varsigma = \ln \left( \frac{e^{-rT}K - F_0}{C_0} \right)$ .

In the following proposition we consider the special case that  $Y_n = Y$  is a constant. In this case, the expression is more explicit.

**Proposition 3.21.** In the case that  $Y_n = Y$  is a constant, the vanilla call/put option on the whole CPPI portfolio's value at maturity has the explicit expression:

$$Call(0, v, T, K) = \sum_{k=0}^{\infty} \left( \frac{e^{-\int_0^t \psi_s \lambda_s ds} \left( \int_0^t \psi_s \lambda_s ds \right)^k}{k!} \right)$$
$$\times \left( C_0 e^{M^{(k)} + \frac{1}{2}\sigma_{(k)}^2} \Psi\left( \frac{M^{(k)} + \sigma_{(k)}^2 - \varsigma}{\sigma_{(k)}} \right) - (F_0 - e^{-rT}K) \Psi\left( \frac{M^{(k)} - \varsigma}{\sigma_{(k)}} \right) \right)$$

and

$$Put(0, v, T, K) = \sum_{k=0}^{\infty} \left( \frac{e^{-\int_0^t \psi_s \lambda_s ds} \left( \int_0^t \psi_s \lambda_s ds \right)^k}{k!} \right)$$
$$\times \left( -C_0 e^{M^{(k)} + \frac{1}{2}\sigma_{(k)}^2} \Psi \left( -\frac{M^{(k)} - \sigma_{(k)}^2 + \varsigma}{\sigma_{(k)}} \right) + \left( -F_0 + e^{-rT}K \right) \Psi \left( \frac{-M^{(k)} + \varsigma}{\sigma_{(k)}} \right) \right),$$

where  $K > F_T$  and

$$M^{(k)} = \int_0^T \left( (m - Y\psi_s \lambda_s) - \frac{m\sigma_s^2}{2} \right) ds + k \ln(1 + mY),$$
  
$$\sigma_{(k)}^2 = \int_0^T m\sigma_s dW_s^{\mathbb{Q}},$$

$$\varsigma = \ln\left(\frac{e^{-rT}K - F_0}{C_0}\right)$$

and

$$\Psi(x) = \int_{-\infty}^{x} \frac{1}{\sqrt[6]{2\pi}} e^{-\frac{1}{2}t^{2}} dt.$$

*Proof.* We have

.

$$V_t = C_0 \exp\left\{\int_0^t \left(r + m(\mu_s - r) - \frac{m\sigma_s^2}{2}\right) ds + \int_0^t m\sigma_s dW_s + \left[\sum_{n=1}^{N_t} \ln(1 + mY_n)\right]\right\} + F_t$$
$$= C_0 \exp\left\{\int_0^t \left(r + m(-Y\psi_s\lambda_s) - \frac{m\sigma_s^2}{2}\right) ds + \int_0^t m\sigma_s dW_s^Q + \left[\sum_{n=1}^{N_t} \ln(1 + mY_n)\right]\right\} + F_t.$$

In case that  $N_T = k$ , we have

$$e^{-rT}\frac{V_T^k - F_T}{C_0} = \exp\left\{\int_0^T \left(m(-Y\psi_s\lambda_s) - \frac{m\sigma_s^2}{2}\right)ds + \int_0^T m\sigma_s dW_s^Q + \left[\sum_{n=1}^{N_T}\ln(1+mY_n)\right]\right\}.$$

Then we have

$$\ln\left(e^{-rT}\frac{V_T^k - F_T}{C_0}\right) \sim \mathcal{N}\left(.; M^{(k)}, \sigma^2_{(k)}\right),$$

where

$$M^{(k)} = \int_0^T \left( m(-Y\psi_s\lambda_s) - \frac{m\sigma_s^2}{2} \right) ds + k\ln(1+mY)$$
  
$$\sigma_{(k)}^2 = \int_0^T m\sigma_s dW_s^{\mathbb{Q}}.$$

Thus

$$\mathbb{E}^{\mathbb{Q}}\left(e^{-rT}(V_T^{(k)}-K)^+\right) = \int_{\varsigma}^{\infty} \left(C_0 e^x + F_0 - e^{-rT}K\right) d\left(\mathcal{N}\left(x; M^{(k)}, \sigma_{(k)}^2\right)\right)$$
69

$$= C_0 e^{M^{(k)} + \frac{1}{2}\sigma^2_{(k)}} \Psi\left(\frac{M^{(k)} + \sigma^2_{(k)} - \varsigma}{\sigma_{(k)}}\right) - \left(F_0 - e^{-rT}K\right) \Psi\left(\frac{M^{(k)} - \varsigma}{\sigma_{(k)}}\right),$$

where

$$\varsigma = \ln\left(\frac{e^{-rT}K - F_0}{C_0}\right).$$

and

$$\Psi(x) = \int_{-\infty}^{x} \frac{1}{\sqrt{2\pi}} e^{-\frac{1}{2}t^2} dt.$$

Then

$$Call(0, v, T, K) = \mathbb{E}^{\mathbb{Q}}(e^{-rT}(V_T - K)^+) = \sum_{k=0}^{\infty} \mathbb{E}^{\mathbb{Q}}({}^{-rT}(V_T^{(k)} - K)^+)(\frac{e^{-\int_0^t \psi_s \lambda_s ds}(\int_0^t \psi_s \lambda_s ds)^k}{k!})$$
$$= \sum_{k=0}^{\infty} \left(\frac{e^{-\int_0^t \psi_s \lambda_s ds}\left(\int_0^t \psi_s \lambda_s ds\right)^k}{k!}\right)$$
$$\times \left(C_0 e^{M^{(k)} + \frac{1}{2}\sigma_{(k)}^2} \Psi\left(\frac{M^{(k)} + \sigma_{(k)}^2 - \varsigma}{\sigma_{(k)}}\right) - (F_0 - e^{-rT}K)\Psi\left(\frac{M^{(k)} - \varsigma}{\sigma_{(k)}}\right)\right).$$

Similarly,

$$\mathbb{E}^{\mathbb{Q}}\left(e^{-rT}(K-V_{T}^{(k)})^{+}\right) = \int_{-\infty}^{\varsigma} \left(-C_{0}e^{x} - F_{0} + e^{-rT}K\right) d\left(\mathcal{N}\left(x; M^{(k)}, \sigma_{(k)}^{2}\right)\right)$$
$$= -C_{0}e^{M^{(k)} + \frac{1}{2}\sigma_{(k)}^{2}}\Psi\left(-\frac{M^{(k)} - \sigma_{(k)}^{2} + \varsigma}{\sigma_{(k)}}\right) + \left(-F_{0} + e^{-rT}K\right)\Psi\left(\frac{-M^{(k)} + \varsigma}{\sigma_{(k)}}\right)$$

and

$$= \mathbb{E}^{\mathbb{Q}}(e^{-rT}(K-V_{T})^{+}) = \sum_{k=0}^{\infty} \mathbb{E}^{\mathbb{Q}}({}^{-rT}(K-V_{T}^{(k)})^{+})(\frac{e^{-\int_{0}^{t}\psi_{s}\lambda_{s}ds}(\int_{0}^{t}\psi_{s}\lambda_{s}ds)^{k}}{k!})$$

$$= \sum_{k=0}^{\infty} \left(\frac{e^{-\int_{0}^{t}\psi_{s}\lambda_{s}ds}\left(\int_{0}^{t}\psi_{s}\lambda_{s}ds\right)^{k}}{k!}\right)$$

$$\times \left(-C_{0}e^{M^{(k)}+\frac{1}{2}\sigma_{(k)}^{2}}\Psi\left(-\frac{M^{(k)}-\sigma_{(k)}^{2}+\varsigma}{\sigma_{(k)}}\right) + \left(-F_{0}+e^{-rT}K\right)\Psi\left(\frac{-M^{(k)}+\varsigma}{\sigma_{(k)}}\right)\right).$$

*Remarks.* The assumption of the jump  $Y_n$  be constant is not reasonable, however, it is looks like the option formula in Black-Scholes model with constant coefficient.

### **3.4** Mean-variance Hedging

#### 3.4.1 Introduction

Given a contingent claim H and suppose there is no arbitrage opportunities, then in a complete market H is attainable, i.e. there exists a self-financing strategy with final portfolio value  $Z_T = H$ ,  $\mathbb{P}$ -a.s. However, when in our jump-diffusion model, the market is not complete and so H is not attainable. In this case we consider quadratic hedging. There are two approaches. One approach is risk-minimization; the other approach is mean-variance hedging. See [67]. We employ the notations from that paper.

We consider the mean-variance hedging. For any contingent claim, let the payoff at T be H. Our jump-diffusion model of the risky asset price S is a semimartingale under  $\mathbb{P}$ . The following definition is taken from section 4 in [67].

**Definition 3.22.** We denote by  $\Theta_2$  the set of all  $\vartheta \in L(S)$  such that the stochastic integral process  $G(\vartheta) := \int \vartheta dS$  satisfies  $G_T \in L^2(\mathbb{P})$ . For a fixed linear subspace  $\Theta$ of  $\Theta_2$ , a  $\Theta$ -strategy is a pair  $(Z_0, \vartheta) \in \mathbb{R} \times \Theta$  and its value process is  $Z_0 + G(\vartheta)$ . A  $\Theta$ -strategy  $(\tilde{Z}_0, \tilde{\vartheta})$  is called  $\Theta$ -mean-variance optimal for a given contingent claim  $H \in L^2$  if it minimizes  $||H - Z_0 - G_T(\vartheta)||_{L^2}$  over all  $\Theta$ -strategies  $(Z_0, \vartheta)$  and  $\tilde{Z}_0$  is then called the  $\Theta$ -approximation price for H.

The linear subspace

$$\mathcal{G} := G_T(\Theta) = \left\{ \int_0^T \vartheta_u dS_u | \vartheta \in \Theta \right\}$$

of  $L^2$  describes all outcomes of self-financing  $\Theta$ -strategies with initial wealth  $Z_0 = 0$ and

$$\mathcal{A} = \mathbb{R} + \mathcal{G} = \left\{ Z_0 + \int_0^T \vartheta_u dS_u | (Z_0, \, \vartheta) \in (\mathbb{R} \times \Theta) \right\}$$

is the space of contingent claims replicable by self-financing  $\Theta$ -strategies. Our goal in mean-variance hedging is to find the projection in  $L^2$  of H on  $\mathcal{A}$  and this can be studied for a general linear subspace  $\mathcal{G}$  of  $L^2$  space. In analogy to the above definition, we introduce a  $\mathcal{G}$ -mean-variance optimal pair  $(\tilde{Z}_0, \tilde{g}) \in \mathbb{R} \times \mathcal{G}$  for  $H \in L^2$  and call  $\tilde{Z}_0$ the  $\mathcal{G}$ -approximation price for H. Our goal is to find

$$\min_{(Z_0,\vartheta)\in\mathbb{R}\times\Theta}||H-Z_0-G_T(\vartheta)||_{L^2}.$$

Since

$$dS_t = S_{t-}[\mu_t dt + \sigma_t dW_t + Y_t dN_t],$$

we have

$$\min_{\substack{(Z_0,\vartheta)\in\mathbb{R}\times\Theta\\(Z_0,\vartheta)\in\mathbb{R}\times\Theta}} \left\| H - Z_0 - \int_0^T \vartheta_u dS_u \right\|_{L^2} \\
= \min_{\substack{(Z_0,\vartheta)\in\mathbb{R}\times\Theta\\(Z_0,\vartheta)\in\mathbb{R}\times\Theta}} \left( \mathbb{E}\left\{ H - Z_0 - \int_0^T \vartheta_u S_{u-} \left[\mu_u du + \sigma_u dW_u + Y_t dN_u\right] \right\}^2 \right)^{\frac{1}{2}}.$$

[67] has pointed out that finding the optimal  $\tilde{\vartheta}$  is in general an open problem. On the other hand, in the case of real contingent claim pricing, we should always using the risk-neutral measure. [20] gives the  $\mathcal{G}$ -mean-variance optimal pair  $(\tilde{Z}_0, \tilde{g})$  when the stocks' model is an exponential levy form martingale. For similar consideration also see Chapter 10 in [18].

#### 3.4.2 Our Problem

Now we consider H as a function of  $V_T$  and denote  $H = g(V_T)$ . For any martingale measure  $\mathbb{Q}$  defined in (3.10), we have proved that  $\tilde{V}_t = e^{-rt}V_t$  is a  $\mathbb{Q}$ -martingale. Denote  $\tilde{H} = e^{-rT}H$ . We want to consider the following optimization problem.

$$\min_{(Z_0,\vartheta)\in\mathbb{R}\times\Theta}\mathbb{E}^{\mathbb{Q}}\left(\tilde{H}-Z_0-\int_0^T\vartheta_u d\tilde{V}_u\right)^2.$$
(3.26)

**Proposition 3.23.** The solution of the optimization problem (3.26) is

$$Z_0 = \mathbb{E}^{\mathbb{Q}}\left[\tilde{H}\right]$$

$$\vartheta_t = \frac{\sigma_t(\mathcal{C}_x(t, V_t)) + (\mathcal{C}(t, V_t + (V_{t-} - F_t)m_tY_t) - \mathcal{C}(t, V_t))Y_t\lambda_t\psi_t}{\sigma_t + (V_{t-} - F_t)m_tY_t^2\lambda_t\psi_t}$$

*Proof.* We have

$$\mathbb{E}^{\mathbb{Q}}\left(\tilde{H} - Z_{0} - \int_{0}^{T} \vartheta_{u} d\tilde{V}_{u}\right)^{2} = \mathbb{E}^{\mathbb{Q}}\left(\mathbb{E}^{\mathbb{Q}}\left[\tilde{H}\right] - Z_{0} + \tilde{H} - \mathbb{E}^{\mathbb{Q}}\left[\tilde{H}\right] - \int_{0}^{T} \vartheta_{u} d\tilde{V}_{u}\right)^{2}$$
$$= \mathbb{E}^{\mathbb{Q}}\left[\left(\mathbb{E}^{\mathbb{Q}}\left[\tilde{H}\right] - Z_{0}\right)^{2}\right] + \mathbb{E}^{\mathbb{Q}}\left(\tilde{H} - \mathbb{E}^{\mathbb{Q}}\left[\tilde{H}\right] - \int_{0}^{T} \vartheta_{u} d\tilde{V}_{u}\right)^{2}.$$

We see that the optimal value for the initial capital is  $Z_0 = \mathbb{E}^{\mathbb{Q}} \left[ \tilde{H} \right]$ . Define  $\mathcal{C}(t,x) = e^{rt} \mathbb{E}^{\mathbb{Q}} \left[ \tilde{H} | V_t = x \right]$  and  $\tilde{\mathcal{C}}(t,x) = e^{-rt} \mathcal{C}(t,x)$ . By construction,  $\tilde{\mathcal{C}}(t,x)$  is a  $\mathbb{Q}$ -martingale. We have deduced that

$$dV_t = (rV_{t-} + (V_{t-} - F_t)m_t(\mu_t - r))dt + (V_{t-} - F_t)m_t\sigma_t dW_t$$
$$+ (V_{t-} - F_t)m_tY_t dN_t,$$

and

$$d\tilde{V}_t = e^{-rt} \left( (V_{t-} - F_t) m_t \sigma_t dW_t^{\mathbb{Q}} + (V_{t-} - F_t) m_t Y_t dM_t^{\mathbb{Q}} \right).$$

Then by Ito's formula we have

$$\begin{split} d\tilde{\mathcal{C}}(t,V_t) \\ &= \left( -re^{-rt}\mathcal{C}(t,V_t) + e^{-rt}\mathcal{C}_t(t,V_t) + (rV_{t-} + (V_{t-} - F_t)m_t(\mu_t - r))e^{-rt}\mathcal{C}_x(t,V_t) \right. \\ &+ \frac{1}{2}(V_{t-} - F_t)^2 m_t^2 \sigma_t^2 e^{-rt}\mathcal{C}_{xx}(t,V_t) \Big) dt + (V_{t-} - F_t)m_t \sigma_t e^{-rt}\mathcal{C}_x(t,V_t) dW_t \\ &+ \left( e^{-rt}\mathcal{C}(t,V_t + (V_{t-} - F_t)m_tY_t) - e^{-rt}\mathcal{C}(t,V_t) \right) dN_t \\ &= \left( V_{t-} - F_t \right) m_t \sigma_t e^{-rt}\mathcal{C}_x(t,V_t) dW_t^{\mathbb{Q}} \\ &+ \left( e^{-rt}\mathcal{C}(t,V_t + (V_{t-} - F_t)m_tY_t) - e^{-rt}\mathcal{C}(t,V_t) \right) dM_t^{\mathbb{Q}}. \end{split}$$

Thus we have

$$\begin{split} \tilde{H} &- \mathbb{E}^{\mathbb{Q}} \left[ \tilde{H} \right] - \int_{0}^{T} \vartheta_{u} d\tilde{V}_{u} \\ &= \tilde{\mathcal{C}}(T, V_{T}) - \tilde{\mathcal{C}}(0, V_{0}) - \int_{0}^{T} \vartheta_{t} e^{-rt} \left( (V_{t-} - F_{t}) m_{t} \sigma_{t} dW_{t}^{\mathbb{Q}} + (V_{t-} - F_{t}) m_{t} Y_{t} dM_{t}^{\mathbb{Q}} \right) \\ &= e^{-rt} \bigg( \int_{0}^{T} (V_{t-} - F_{t}) m_{t} \sigma_{t} (\mathcal{C}_{x}(t, V_{t}) - \vartheta_{t}) dW_{t}^{\mathbb{Q}} \\ &+ \int_{0}^{T} ((\mathcal{C}(t, V_{t} + (V_{t-} - F_{t}) m_{t} Y_{t}) - \mathcal{C}(t, V_{t})) - \vartheta_{t} (V_{t-} - F_{t}) m_{t} Y_{t}) dM_{t}^{\mathbb{Q}} \bigg). \end{split}$$

By the Isometry formula, we have

$$\mathbb{E}^{\mathbb{Q}}\left(\tilde{H} - \mathbb{E}^{\mathbb{Q}}\left[\tilde{H}\right] - \int_{0}^{T} \vartheta_{u} d\tilde{V}_{u}\right)^{2}$$

$$= e^{-2rt} \left(\mathbb{E}^{\mathbb{Q}}\left[\int_{0}^{T} ((V_{t-} - F_{t})m_{t}\sigma_{t}(\mathcal{C}_{x}(t, V_{t}) - \vartheta_{t}))^{2} dt\right]$$

$$+\mathbb{E}^{\mathbb{Q}}\left[\int_{0}^{T} (((\mathcal{C}(t, V_{t} + (V_{t-} - F_{t})m_{t}Y_{t}) - \mathcal{C}(t, V_{t})) - \vartheta_{t}(V_{t-} - F_{t})m_{t}Y_{t}))^{2}\lambda_{t}\psi_{t} dt\right]\right).$$

This is the minimizing problem with respect to  $\vartheta_t$ . Differentiating the above expression with respect to  $\vartheta_t$  and letting the first order derivative equal to 0, we have

$$(V_{t-} - F_t)m_t\sigma_t(\mathcal{C}_x(t, V_t) - \vartheta_t) + (((\mathcal{C}(t, V_t + (V_{t-} - F_t)m_tY_t) - \mathcal{C}(t, V_t)) - \vartheta_t(V_{t-} - F_t)m_tY_t))(V_{t-} - F_t)m_tY_t\lambda_t\psi_t = 0,$$

thus

$$\vartheta_t = \frac{\sigma_t(\mathcal{C}_x(t, V_t)) + (\mathcal{C}(t, V_t + (V_{t-} - F_t)m_tY_t) - \mathcal{C}(t, V_t))Y_t\lambda_t\psi_t}{\sigma_t + (V_{t-} - F_t)m_tY_t^2\lambda_t\psi_t}$$

*Remarks.* When the contingent claim is the call option with the strike price K, i.e.

 $H = (V_T - K)^+$ , then

$$Z_0 = \mathbb{E}^{\mathbb{Q}}\left[\tilde{H}\right] = \operatorname{Call}(0, V_0, T, K)$$

and

$$\mathcal{C}(t,x) = e^{rt} \mathbb{E}^{\mathbb{Q}} \left[ \tilde{H} | V_t = x \right] = \operatorname{Call}(t,x,T,K);$$

when the contingent claim is the put option with the strike price K, i.e.  $H = (K - V_T)^+$ , then

$$Z_0 = \mathbb{E}^{\mathbb{Q}}\left[\tilde{H}\right] = \operatorname{Put}(0, V_0, T, K)$$

and

$$\mathcal{C}(t,x) = e^{rt} \mathbb{E}^{\mathbb{Q}} \left[ \tilde{H} | V_t = x \right] = \operatorname{Put}(t,x,T,K).$$

This is consitent with the calculation of call and put options.

# Chapter 4

# Gap risks

## 4.1 Introduction

Let

$$\frac{dS_t}{S_{t-}} = dZ_t. \tag{4.1}$$

where  $Z_t$  is a Levy process, a special case of which would be our jump diffusion. We will show the probability of loss we obtain is consistant with [17] and our result on the expected loss is more explicit and the method is similar to [17]; the result we obtain for the loss distribution is explicit and our method is different from [17]. Two kinds of conditional floors will be introduced in section 4.3. Its idea is similar to

the Value-at-Risk considered in [27]. Meanwhile, four kinds of conditional floor are discussed associated with expected loss and loss distribution.

## 4.2 Gap risk Measure for CPPI strategies in Jumpdiffusion model

#### 4.2.1 Probability of Loss

In practice, a CPPI-insured portfolio incurs a loss (breaks through the floor) if, for some  $t \in [0,T]$ ,  $V_t \leq F_t$ . The event  $V_t \leq F_t$  is equivalent to  $C_t \leq 0$ . It happens at time  $T_i$ , associated with the *i*-th jump of the risky asset,  $1 + mY_i \leq 0$ . We have

**Proposition 4.1.** Let the density of  $\ln(1+Y_n)$  be  $f_Q(y)$ . The probability of the CPPI portfolio value going below the floor taking happen during time [0,T] is given by

$$\mathbb{P}\left[\exists t \in [0,T] : V_t \le F_t\right] = 1 - \exp\left\{\int_0^T \lambda_s ds\left(\int_{\ln\left(1-\frac{1}{m}\right)}^\infty f_Q(y)dy - 1\right)\right\}$$
(4.2)

Proof. Since

$$\mathbb{P}(1+mY_i>0) = \mathbb{P}\left(\ln(1+Y_i) > \ln\left(1-\frac{1}{m}\right)\right) = \int_{\ln\left(1-\frac{1}{m}\right)}^{\infty} f_Q(y)dy,$$

then

$$\begin{split} \mathbb{P}[\exists t \in [0, T] : V_t \leq F_t] &= \mathbb{P}(\exists t \in [0, T] : C_t \leq 0) \\ &= \mathbb{P}(\exists T_i, 1 + mY_i \leq 0) = 1 - \mathbb{P}(\forall T_i, 1 + mY_i > 0) \\ &= 1 - \mathbb{P}\left(\bigcup_{j=0}^{\infty} [\forall T_i 1 + mY_i > 0, N_T = j]\right) \\ &= 1 - \sum_{j=0}^{\infty} \mathbb{P}\left([\forall T_i 1 + mY_i > 0, N_T = j]\right) \\ &= 1 - \sum_{j=0}^{\infty} \mathbb{P}(\forall T_i 1 + mY_i > 0 | N_T = j) \mathbb{P}(N_T = j) \\ &= 1 - \sum_{j=0}^{\infty} \mathbb{P}(\forall T_1, T_2 \dots T_j, 1 + mY_i > 0) \mathbb{P}(N_T = j) \end{split}$$

$$= 1 - \sum_{j=0}^{\infty} \mathbb{P}\left(\bigcap_{i=1}^{j} (1+mY_i > 0)\right) \mathbb{P}(N_T = j)$$

$$= 1 - \sum_{j=0}^{\infty} \prod_{i=1}^{j} \mathbb{P}(1+mY_i > 0) \mathbb{P}(N_T = j)$$

$$= 1 - \sum_{j=0}^{\infty} \frac{e^{-\int_0^T \lambda_s ds} (\int_0^T \lambda_s ds)^j}{j!} \left(\int_{\ln\left(1-\frac{1}{m}\right)}^\infty f_Q(y) dy\right)^j$$

$$= 1 - \exp\left\{\int_0^T \lambda_s ds \left(\int_{\ln\left(1-\frac{1}{m}\right)}^\infty f_Q(y) dy - 1\right)\right\}.$$

*Remarks.* (1) When  $\lambda_s = \lambda$ , the probability of loss is

$$1 - \exp\left\{T\lambda\left(\int_{\ln\left(1-\frac{1}{m}\right)}^{\infty} f_Q(y)dy - 1\right)\right\}$$

$$(4.3)$$

Our conclusion is a special case of Corollary 3.1 in [17], where the probability of loss is given by

$$1 - \exp\left(-T\int_{-\infty}^{\ln(1-1/m)}\nu(dx)\right).$$

In our case the levy measure  $\nu$  is  $\nu(dx) = \lambda f_Q(x) dx$  (See Page 75, [18] or Page 14, [57]), then

$$1 - \exp\left(-T \int_{-\infty}^{\ln(1-1/m)} \nu(dx)\right)$$
  
=  $1 - \exp\left(-T \int_{-\infty}^{\ln(1-1/m)} \lambda f_Q(x) dx\right)$   
=  $1 - \exp\left\{T\lambda\left(\int_{\ln(1-\frac{1}{m})}^{\infty} f_Q(y) dy - 1\right)\right\}.$ 

(2) This proposition describes the loss takes place before the mature time T. Thus, it is naturally to generalize it to the time  $t \in [0, T]$ . We have the following corollary, 79 **Corollary 4.2.** Assume  $\lambda = \lambda_s$  and let  $\tau \leq T$  if the loss take happen i.e.  $C_{\tau} \leq 0$ and  $\tau = \infty$  otherwise. The distribution of  $\tau$  is, for  $t \in [0, T]$ 

$$\mathbb{P}(\tau \le t) = 1 - \exp\left\{t\lambda\left(\int_{\ln(1-\frac{1}{m})}^{\infty} f_Q(y)dy - 1\right)\right\}$$

and the density function  $f_{\tau}$  of  $\tau$  is

$$f_{\tau} = -\exp\left\{t\lambda\left(\int_{\ln(1-\frac{1}{m})}^{\infty} f_Q(y)dy - 1\right)\right\}\left(\lambda\left(\int_{\ln(1-\frac{1}{m})}^{\infty} f_Q(y)dy - 1\right)\right).$$
 (4.4)

*Proof.* The first one is obvious and the second one is the derivative with respect to t.

#### 4.2.2 Expected Loss

Let  $\tau$  be the first time when  $C_{\tau} \leq 0$  and we let  $\tau = \infty$  if the loss never happens. Let

$$\varepsilon_t = C_0 \exp\left\{\int_0^t \left(m(\mu_s - r) - \frac{m^2 \sigma_s^2}{2}\right) ds + \int_0^t m \sigma_s dW_s\right\} \left[\prod_{n=1}^{N_t} (1 + mY_n)\right] \quad (4.5)$$

If a loss takes place, then at time  $\tau$ , the cushion  $C_{\tau} \leq 0$ . If we do not allow short-sell, then, at time  $\tau$ +, we let the exposure be 0. Then, we have the discounted cushion:

$$C_T^* = \varepsilon_T \chi_{\tau > T} + \varepsilon_\tau (1 + m Y_\tau) \chi_{\tau \le T}$$

$$(4.6)$$

*Remarks.* In subsection 3.2 we allowed negative exposure to happen and we have the expression for the cushion  $C_t$  and the portfolio value  $V_t$ . When the CPPI portfolio is considered as an hedging tool in subsection 3.3, short-selling is allowed.

In this subsection, we take the exposure to be 0 at the time when there is a loss and we measure the gap.

When  $t < \tau$ ,  $1 + mY_i > 0$  for  $T_i \le t$ . We first calculate the characteristic function 80

of  $\ln\left(\frac{\varepsilon_t}{C_0}\right)$  for  $t < \tau$ . Since  $1 + mY_i > 0$  when  $t < \tau$ , we have

$$\ln\left(\frac{\varepsilon_t}{C_0}\right) = \int_0^t \left(m(\mu_s - r) - \frac{m^2 \sigma_s^2}{2}\right) ds + \int_0^t m \sigma_s dW_s + \sum_{n=1}^{N_t} \ln(1 + mY_n) ds$$

**Proposition 4.3.** Let the density function of  $\ln(1+Y_i)$  be  $f_Q(x)$ . When  $t < \tau$ , the characteristic function  $\phi_t(u)$  of  $\ln\left(\frac{\varepsilon_t}{C_0}\right)$  is

$$\phi_t(u) = \exp\left\{i\left(\int_0^t \left(m(\mu_s - r) - \frac{m^2\sigma_s^2}{2}\right)ds\right)u - \frac{1}{2}\left(\int_0^t m\sigma_s^2ds\right)u^2\right\} \times \exp\left\{t\lambda\int_{\mathbb{R}}(e^{iux} - 1)f_Q\left(\ln\left(1 + \frac{e^x - 1}{m}\right)\right)\frac{e^x}{m + e^x - 1}dx\right\}.$$

*Proof.* Since  $\int_0^t \left( m(\mu_s - r) - \frac{m^2 \sigma_s^2}{2} \right) ds + \int_0^t m \sigma_s dW_s$  and  $\sum_{n=1}^{N_t} \ln(1 + mY_n)$  are independent, thus the characteristic function of the sum of two random variables is the production of characteristic function of each random variables.

 $\int_0^t \left( (m(\mu_s - r)) - \frac{m^2 \sigma_s^2}{2} \right) ds + \int_0^t m \sigma_s dW_s \text{ is normal distribution with mean} \\ \int_0^t \left( m(\mu_s - r) - \frac{m^2 \sigma_s^2}{2} \right) ds \text{ and variance } \int_0^t m^2 \sigma_s^2 ds \text{ and hence its characteristic function is}$ 

$$\phi_{1,t}(u) = \exp\left\{i\left(\int_0^t \left(m(\mu_s - r) - \frac{m^2\sigma_s^2}{2}\right)ds\right)u - \frac{1}{2}\left(\int_0^t m^2\sigma_s^2ds\right)u^2\right\}.$$
 (4.7)

In section 3.3, we have deduced the density function  $f'_Q$  of the random variable  $\ln(1 + mY_i)$  is

$$f'_Q(z) = f_Q\left(\ln\left(1 + \frac{e^z - 1}{m}\right)\right) \frac{e^z}{m + e^z - 1}.$$

We denote the characteristic function of  $f'_Q$  by  $\hat{f'_Q}$ . Then, the characteristic function

 $\phi_{2,t}(u)$  of  $\sum_{n=1}^{N_t} \ln(1+mY_n)$  is

$$\begin{split} \phi_{2,t}(u) &= \mathbb{E}\left[\exp iu\left(\sum_{n=1}^{N_t}\ln(1+mY_n)\right)\right] \\ &= \mathbb{E}\left[\mathbb{E}\left[\exp iu.\left(\sum_{n=1}^{N_t}\ln(1+mY_n)\right)\right]|N_t\right] = \mathbb{E}\left[\left(\hat{f}'_Q(u)\right)^{N_t}\right] \\ &= \sum_{j=0}^{\infty} \frac{e^{-\lambda t}(-\lambda t)^j \left(\hat{f}'_Q(u)\right)^j}{j!} = \exp\left\{\lambda t \left(\hat{f}'_Q(u)-1\right)\right\} \\ &= \exp\left\{t\lambda \int_{\mathbb{R}} \left(e^{iux}-1\right) f'_Q(dx)\right\} \\ &= \exp\left\{t\lambda \int_{\mathbb{R}} \left(e^{iux}-1\right) f_Q\left(\ln\left(1+\frac{e^x-1}{m}\right)\right) \frac{e^x}{m+e^x-1}dx\right\} \end{split}$$

Then, the characteristic function  $\phi_t(u)$  of  $\ln\left(\frac{\varepsilon_t}{C_0}\right)$  is

$$\begin{split} \phi_t(u) &= \phi_{1,t}(u)\phi_{2,t}(u) \\ &= \exp\left\{i\left(\int_0^t \left(m(\mu_s - r) - \frac{m^2\sigma_s^2}{2}\right)ds\right)u - \frac{1}{2}\left(\int_0^t m^2\sigma_s^2ds\right)u^2\right\} \\ &\times \exp\left\{t\lambda\int_{\mathbb{R}}(e^{iux} - 1)f_Q\left(\ln\left(1 + \frac{e^x - 1}{m}\right)\right)\frac{e^x}{m + e^x - 1}dx\right\}. \end{split}$$

**Definition 4.4.** The conditional expectation of the discounted cushion is called the **conditional expected loss** and we assume that  $\mathbb{E}[C_T^*|\tau \leq T]$ ; while the **unconditional expected loss** is represented by  $\mathbb{E}[C_T^*\chi_{\tau \leq T}]$ .

We have:

**Proposition 4.5.** The expectation of loss conditioned on the fact that a loss has occured is

$$\mathbb{E}[C_T^*|\tau \le T] = \frac{\int_{-\infty}^{\ln(1-1/m)} f_Q(y) dy \int_0^T C_0 \phi_t(-i) f_\tau dt}{1 - \exp\left\{T\lambda\left(\int_{\ln\left(1-\frac{1}{m}\right)}^\infty f_Q(y) dy - 1\right)\right\}}$$
(4.8)

and the unconditional expected loss satisfies

$$\mathbb{E}[C_T^*\chi_{\tau \le T}] = \int_{-\infty}^{\ln(1-1/m)} f_Q(y) dy \int_0^T C_0 \phi_t(-i) f_\tau dt$$
(4.9)

where  $f_{\tau}$  is the density function of  $\tau$  and defined by (4.4) and  $\phi_t$  is the characteristic function of  $\ln\left(\frac{\varepsilon_t}{C_0}\right)$ .

*Proof.* First the discounted cushion is

$$C_T^* = \varepsilon_T \chi_{\tau > T} + \varepsilon_\tau (1 + m Y_\tau) \chi_{\tau \le T}.$$

Then

$$\mathbb{E}[C_T^*\chi_{\tau\leq T}] = \mathbb{E}[(1+mY_{\tau})]\mathbb{E}[\varepsilon_{\tau}].$$

 $(1 + mY_{\tau})$  is the size of the first jump which size is  $Y_i < -1/m$ . Thus,

$$\mathbb{E}\left[\left(1+mY_{\tau}\right)\left(\chi_{\tau\leq T}\right)\right] = \int_{-\infty}^{\ln(1-1/m)} f_Q(y) dy.$$

By the property of the characteristic function, we have

$$\mathbb{E}[\varepsilon_t] = C_0 \phi_t(-i).$$

Therefore

$$\mathbb{E}[C_T^*\chi_{\tau\leq T}] = \mathbb{E}\left[\left(1+mY_{\tau}\right)\left(\chi_{\tau\leq T}\right)\right]\mathbb{E}[\varepsilon_{\tau}\chi_{\tau\leq T}]$$
$$= \int_{-\infty}^{\ln(1-1/m)} f_Q(y)dy \int_0^T C_0\phi_t(-i)f_{\tau}dt,$$

where  $f_{\tau}$  is the density function of  $\tau$  and defined by (4.4). Furthermore, by the

property of conditional expectation, we have

$$\mathbb{E}[C_T^*|\tau \le T] = \frac{\mathbb{E}[C_T^*\chi_{\tau \le T}]}{\mathbb{P}[\tau \le T]} = \frac{\int_{-\infty}^{\ln(1-1/m)} f_Q(y) dy \int_0^T C_0 \phi_t(-i) f_\tau dt}{1 - \exp\left\{T\lambda\left(\int_{\ln\left(1-\frac{1}{m}\right)}^\infty f_Q(y) dy - 1\right)\right\}}.$$

### 4.2.3 Loss Distribution

To compute risk measures, we consider, for x < 0, the quantity

$$\mathbb{P}[C_T^* < x | \tau \le T]. \tag{4.10}$$

This is called the Loss Distribution. We next have:

**Proposition 4.6.** Let the density of  $\ln(1 + Y_n)$  be  $f_Q(y)$  and  $C_T^*$  be the discounted cushion. For x < 0, the unconditional loss distribution is

$$\mathbb{P}\left[C_{T}^{*}\chi_{\tau \leq T} < x\right] \\
= \int_{0}^{\ln\left(-\frac{x}{C_{0}}\right)} \int_{0}^{T} \phi\left(z, \int_{0}^{t} \left(m(\mu_{s} - r) - \frac{m^{2}\sigma_{s}^{2}}{2}\right) ds, \int_{0}^{t} m\sigma_{s}^{2} ds\right) f_{\tau} dt \Big|_{z=l} \qquad (4.11) \\
\times \left(-f_{Q}\left(\ln\left(1 + \frac{-e^{z} - 1}{m}\right)\right) \frac{e^{z}}{-m + e^{z} + 1}\right) \Big|_{z=\ln(-\frac{x}{C_{0}}) - l} dl.$$

and the loss distribution is

$$\mathbb{P}[C_T^* < x | \tau \le T] = \frac{\mathbb{P}[C_T^* \chi_{\tau \le T} < x]}{1 - \exp\left\{T\lambda\left(\int_{\ln\left(1 - \frac{1}{m}\right)}^{\infty} f_Q(y) dy - 1\right)\right\}},\tag{4.12}$$

where  $f_{\tau}$  is the density function of  $\tau$  and defined by (31) and  $\phi(x, m, v^2)$  is the density function of normal distribution which is same as defined in subsection 3.1.1.

*Proof.* For x < 0, the unconditional loss distribution is

$$\mathbb{P}[C_T^* \chi_{\tau \leq T} < x] = \mathbb{P}[\varepsilon_\tau (1 + mY_\tau)\chi_{\tau \leq T} < x]$$

$$= \mathbb{P}\left[\frac{\varepsilon_\tau}{C_0}(-(1 + mY_\tau))\chi_{\tau \leq T} > -\frac{x}{C_0}\right]$$

$$= \mathbb{P}\left[\ln\left(\frac{\varepsilon_\tau}{C_0}\chi_{\tau \leq T}\right) + \ln\left((-(1 + mY_\tau))\chi_{\tau \leq T}\right) > \ln\left(-\frac{x}{C_0}\right)\right]$$

$$= \int_0^{\ln\left(-\frac{x}{C_0}\right)} \frac{d}{dz} \left(\mathbb{P}\left(\ln\left(\frac{\varepsilon_\tau}{C_0}\chi_{\tau \leq T}\right) < z\right)\right)\Big|_{z=l}$$

$$\times \frac{d}{dz} \left(\mathbb{P}\left(\ln\left((-(1 + mY_\tau))\chi_{\tau \leq T}\right) < z\right)\right)\Big|_{z=\ln\left(-\frac{x}{C_0}\right) - l} dl.$$

The last step is by the property of the distribution of the sum of two random variables. Since

$$\frac{d}{dz} \left( \mathbb{P}\left( \ln\left(\frac{\varepsilon_{\tau}}{C_0} \chi_{\tau \le T}\right) < z \right) \right) = \frac{d}{dz} \left( \int_0^T \mathbb{P}\left( \ln\left(\frac{\varepsilon_{\tau}}{C_0}\right) < z \right) f_{\tau} dt \right)$$
$$= \int_0^T \phi\left( z, \int_0^t \left( m(\mu_s - r) - \frac{m^2 \sigma_s^2}{2} \right) ds, \int_0^t m \sigma_s^2 ds \right) f_{\tau} dt$$

where  $f_{\tau}$  is the density function of  $\tau$  and defined by (31) and  $\phi(x, m, v^2)$  is the density function of normal distribution which is same as defined in subsection 3.1.1 and

$$\begin{aligned} &\frac{d}{dz} \left( \mathbb{P}(\ln((-(1+mY_{\tau}))\chi_{\tau \leq T}) < z)) \right) \\ &= \frac{d}{dz} \left( \mathbb{P}\left( (1+mY_{\tau})\chi_{\tau \leq T} > -e^z) \right) \\ &= \frac{d}{dz} \left( \mathbb{P}\left( \ln\left((1+Y_{\tau})\chi_{\tau \leq T}\right) > \ln\left(1+\frac{-e^z-1}{m}\right) \right) \right) \right) \\ &= \frac{d}{dz} \left( \int_{\ln\left(1+\frac{-e^z-1}{m}\right)}^{\ln\left(1-\frac{1}{m}\right)} f_Q(y) dy \right) \\ &= -f_Q \left( \ln\left(1+\frac{-e^z-1}{m}\right) \right) \frac{e^z}{-m+e^z+1}, \end{aligned}$$

substitute the above two expressions, we get

$$\begin{aligned} & \mathbb{P}\left[C_T^*\chi_{\tau \leq T} < x\right] \\ &= \int_0^{\ln\left(-\frac{x}{C_0}\right)} \frac{d}{dz} \left( \mathbb{P}\left(\ln\left(\frac{\varepsilon_{\tau}}{C_0}\chi_{\tau \leq T}\right) < z\right)\right) \Big|_{z=l} \\ & \times \frac{d}{dz} \left( \mathbb{P}(\ln((-(1+mY_{\tau}))\chi_{\tau \leq T}) < z)) \right) \Big|_{z=\ln(-\frac{x}{C_0})-l} dl. \end{aligned}$$

$$&= \int_0^{\ln\left(-\frac{x}{C_0}\right)} \int_0^T \phi\left(z, \int_0^t \left(m(\mu_s - r) - \frac{m^2\sigma_s^2}{2}\right) ds, \int_0^t m\sigma_s^2 ds\right) f_{\tau} dt \Big|_{z=l} \\ & \times \left(-f_Q\left(\ln\left(1 + \frac{-e^z - 1}{m}\right)\right) \frac{e^z}{-m + e^z + 1}\right) \Big|_{z=\ln(-\frac{x}{C_0})-l} dl. \end{aligned}$$

Moreover, the loss distribution is

$$\mathbb{P}[C_T^* < x | \tau \le T] = \frac{\mathbb{P}[C_T^* \chi_{\tau \le T} < x]}{\mathbb{P}[\tau \le T]} = \frac{\mathbb{P}[C_T^* \chi_{\tau \le T} < x]}{1 - \exp\left\{T\lambda\left(\int_{\ln(1-\frac{1}{m})}^{\infty} f_Q(y)dy - 1\right)\right\}}.$$

## 4.3 Conditional Floor and Conditional Multiple of CPPI in the Jump-diffusion Model

#### 4.3.1 Introduction

We want to control the level of the gap by suitably adjusting the floor or/and multiple. For example, if we take m = 1, then the portfolio value is always greater than the floor, and thus there is no gap risk in the case. Another case is if we make the floor equal to initial portfolio measure, then there is also no gap risk. Risk occurs when we choose large enough multiples or low floors which result in more exposures. [1] and [2] describe how the conditional multiple and conditional floor control gap risks in the continuous case. Risks occur because the trading time is discrete.

#### 4.3.2 Probability of Loss

Let the density of  $\ln(1 + Y_n)$  be  $f_Q(y)$ . Recall that the probability that the CPPI portfolio value falls below the floor during the time interval [0, T] is given by (4.2). We see that the probability of loss is irrelevant to the floor. It is also irrelevant to the continuous part of the risky asset model. It is only related to the jump part of risky asset model and the multiple m. Moreover, we have the following proposition.

**Proposition 4.7.** The probability of loss given in (4.2) is monotone increased function as the multiple m.

*Proof.* For m > 1 in general,

$$m \text{ increased} \Longrightarrow$$
$$\ln\left(1 - \frac{1}{m}\right) \text{ is increasing} \Longrightarrow$$
$$\int_{\ln\left(1 - \frac{1}{m}\right)}^{\infty} f_Q(y) dy - 1 \text{ is decreasing} \Longrightarrow$$
$$1 - \exp\left\{\int_0^T \lambda_s ds\left(\int_{\ln\left(1 - \frac{1}{m}\right)}^{\infty} f_Q(y) dy - 1\right)\right\} \text{ is increasing.}$$

Like for the Value-at-Risk (VaR) (See [27]), we define:

**Definition 4.8.** For  $\epsilon > 0$ , the multiple  $m = m_0$  makes

$$\mathbb{P}[\exists t \in [0, T] : V_t \le F_t] = \epsilon$$

is called the  $\epsilon$ -conditional multiple.

 $m_0$  can be treated as a quantile. Since the probability of loss is monotone increased as the function of the multiple m. Then for  $m < m_0$ ,

$$\mathbb{P}\left[\exists t \in [0,T] : V_t \le F_t\right] < \epsilon.$$

Let the distribution function of  $\ln(1+Y_n)$  be  $F_Q$ , then the quantile point  $m_0$  is given by:

$$\mathbb{P}\left[\exists t \in [0,T] : V_t \leq F_t\right] = \epsilon$$

$$\iff 1 - \exp\left\{\int_0^T \lambda_s ds\left(\int_{\ln(1-\frac{1}{m_0})}^\infty f_Q(y) dy - 1\right)\right\} = \epsilon$$

$$\iff -\frac{\ln(1-\epsilon)}{\int_0^T \lambda_s ds} = \int_{-\infty}^{\ln\left(1-\frac{1}{m_0}\right)} f_Q(y) dy$$

$$\iff \ln\left(1-\frac{1}{m_0}\right) = F_Q^{-1}\left(-\frac{\ln(1-\epsilon)}{\int_0^T \lambda_s ds}\right)$$

$$\iff m_0 = \frac{1}{1-\exp\left\{F_Q^{-1}\left(-\frac{\ln(1-\epsilon)}{\int_0^T \lambda_s ds}\right)\right\}}$$

When we know the distribution function of the jump-part, it is easy to determine the  $\epsilon$ -conditional multiple and hence the strategies accordingly.

#### 4.3.3 Expected Loss

Through the notion of Probability of Loss, we determine the conditional multiple and hence control the risk of the gap occurrence. From (4.8) and (4.9), we have the following proposition:

**Proposition 4.9.** Given a fixed multiple, both the conditional expected loss given by (4.8) and the unconditional expected loss given by (4.9) are monotone decreased functions of the initial floor  $F_0$ .

*Proof.* From (4.8) and (4.9), we see that they are increasing functions of the initial cushion  $C_0$ , and  $C_0 = V_0 - F_0$ .

Similar to the concept of  $\epsilon$ -conditional multiple, we define the following:

**Definition 4.10.** For  $\rho < 0$  and  $m = m_0$ , the floor  $F_0 = F^{c1}$  which causes

$$\mathbb{E}[C_T^* | \tau \le T] = \varrho$$

is called the first type  $\rho$ -m<sub>0</sub>-conditional floor.

and

**Definition 4.11.** For  $\rho < 0$  and  $m = m_0$ , the floor  $F_0 = F^{c_2}$  which causes

$$\mathbb{E}[C_T^*\chi_{\tau\leq T}] = \varrho$$

is called the Second type  $\rho$ -m<sub>0</sub>-conditional floor.

From (4.8) and (4.9), we can solve the two conditional floors easily.

### 4.3.4 Loss Distribution

Similar to the case of expected loss, we define the conditional floor in terms of loss distribution. Equation (4.12) gives the unconditional loss distribution and equation (4.12) gives the conditional loss distribution. The following propositions are immediate:

**Proposition 4.12.** Given a fixed multiple and x < 0, the expressions (4.11) and (4.12) are monotone increasing functions of the initial floor  $F_0$ 

*Proof.* From equations (4.11) and (4.12), we see that they are increasing functions of  $C_0$ . Since  $C_0 = V_0 - F_0$ ,

$$F_0 \text{ increased} \Longrightarrow$$
  
 $C_0 \text{ decreased} \Longrightarrow$   
 $\ln\left(-\frac{x}{C_0}\right) \text{ increased} \Longrightarrow$ 

Both the expression (4.11) and (4.12) are increased.

As in the case of  $\epsilon$ -conditional multiple, we define the following two concepts:

**Definition 4.13.** For  $\epsilon > 0$ ,  $\rho < 0$  and  $m = m_0$ , the floor  $F_0 = F^{c3}$  associated with the condition

$$\mathbb{E}[C_T^*\chi_{\tau\leq T} < \varrho] = \epsilon$$

is called the **third type**  $\epsilon$ - $\rho$ - $m_0$ -conditional floor.

and

**Definition 4.14.** For  $\epsilon > 0$ ,  $\rho < 0$  and  $m = m_0$ , the floor  $F_0 = F^{c4}$  which gives

$$\mathbb{E}[C_T^* < \varrho | \tau \le T] = \epsilon$$

is called the **fourth type**  $\epsilon$ - $\rho$ - $m_0$ -conditional floor.

### 4.3.5 Conclusion

The conditional multiple and four conditional floors defined in our section can be used to the investment. The investor can determine them according to their risk-aversion level.

# Chapter 5

# CPPI in the jump-diffusion model when the trading time is discrete

## 5.1 Introduction

In this chapter we discuss the case of discrete trading time. The risky asset model is the same as in chapter 3 and 4.

In section 5.2, as in section 3.2, we calculate the CPPI portfolio value, its expectation and variance.

The gap risks are occurred because the risky model has jumps and also the trading time is discrete. As in section 4.2, we measure the gap risk from three aspects in section 5.3: the probability of loss, the expected loss and the loss distribution.

In section 5.4, similar to the ideas given in section 4.3, we define the conditional multiples associated with the probability of loss as well as the conditional floors from the views of expected loss and loss distribution. It could be treated as an application of 5.2.

In section 5.5, we prove that as the interval of the trading times tends to zero, the CPPI strategies in discrete trading time is agrees with the CPPI strategies in continuous time.

## 5.2 The strategy

Let  $\tau^N = \{t_0 = 0 < t_1 < t_2 < ... < t_N = T\}$  be a sequence of equidistant refinements of the interval [0, T], where  $t_{k+1} - t_k = \frac{T}{N}$  for k = 0, ..., N - 1. Suppose that the trading times are restricted to the discrete set  $\tau^N$ . Furthermore we suppose

 $\mathbb{P}[T_i = t_j] = 0 \quad \forall i = 0, 1, 2, 3, \dots \text{ and } j = 0, 1, 2, \dots, N.$ 

Hence we may assume  $T_i \neq t_j$  for  $\forall i = 0, 1, 2, 3, \dots$  and  $j = 0, 1, 2, \dots, N$ . We have

$$C_{t_{k+1}} = C_{t_k} \left( m \frac{S_{t_{k+1}}}{S_{t_k}} - (m-1)e^{rT/N} \right),$$
(5.1)

then

$$C_T = C_{t_N} = C_0 \prod_{k=0}^{N-1} \left( m \frac{S_{t_{k+1}}}{S_{t_k}} - (m-1)e^{rT/N} \right),$$

thus

$$V_T = C_T + G.$$

Since  $\frac{S_{t_{k+1}}}{S_{t_k}}$ , k = 0, 1, 2, ..., n - 1 are manually independent and also they have the identity distribution, i.e.

$$\frac{S_{t_{k+1}}}{S_{t_k}} = \exp\left[\int_{t_k}^{t_{k+1}} \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds + \int_{t_k}^{t_{k+1}} \sigma_s dW_s + \sum_{n_k = N_{t_k}}^{N_{t_{k+1}}} \ln(1 + Y_{n_k})\right],$$

then we have

$$\mathbb{E}\left[\frac{S_{t_{k+1}}}{S_{t_k}}\right] = \mathbb{E}\left[\exp(\mu \frac{T}{N} + \sigma W_{T/N} - \frac{1}{2}\sigma^2 \frac{T}{N})\right] \mathbb{E}\prod_{n_k=N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k})$$

$$= \exp\left(\mu \frac{T}{N}\right) \mathbb{E} \prod_{n_k=N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k});$$

and

$$\begin{split} \mathbb{E}\left[\frac{S_{t_{k+1}}}{S_{t_k}}\right]^2 &= \mathbb{E}\left[\exp\left(2\mu\frac{T}{N} + 2\sigma W_{T/N} - \sigma^2\frac{T}{N}\right)\right] \mathbb{E}\prod_{n_k=N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k})^2 \\ &= \mathbb{E}\left[\exp\left(2\mu\frac{T}{N} + \sigma^2\frac{T}{N} + 2\sigma W_{T/N} - \frac{1}{2}(2\sigma)^2\frac{T}{N}\right)\right] \mathbb{E}\prod_{n=N_{t_k}}^{N_{t_{k+1}}} (1+Y_n)^2 \\ &= \exp\left(2\mu\frac{T}{N} + \sigma^2\frac{T}{N}\right) \mathbb{E}\prod_{n_k=N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k})^2. \end{split}$$

**Lemma 5.1.** Let the density function of  $\ln(1 + Y_n)$  be  $f_Q$ , then we have

$$\mathbb{E}\prod_{n_k=N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k}) = \sum_{j=1}^{\infty} \frac{e^{-\int_{t_k}^{t_{k+1}} \lambda_s ds} (\int_{t_k}^{t_{k+1}} \lambda_s ds)^j}{j!} \times \int_{\mathbb{R}} \exp\left\{\frac{\int_Q * f_Q * \dots * f_Q(x)}{j \ items}\right\} dx$$
(5.2)

and

$$\mathbb{E}\prod_{n_k=N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k})^2 = \sum_{j=1}^{\infty} \frac{e^{-\int_{t_k}^{t_{k+1}} \lambda_s ds} (\int_{t_k}^{t_{k+1}} \lambda_s ds)^j}{j!}$$

$$\times \int_{\mathbb{R}} \exp\left\{2\underbrace{f_Q * f_Q * \dots * f_Q(x)}_{j \ terms}\right\} dx.$$
(5.3)

 $\mathit{Proof.}$  As the proof of proposition 3.6, we have

$$\mathbb{E}\prod_{n_k=N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k})$$

$$= \sum_{j=1}^{\infty} \frac{e^{-\int_{t_{k}}^{t_{k+1}} \lambda_{s} ds} (\int_{t_{k}}^{t_{k+1}} \lambda_{s} ds)^{j}}{j!} \mathbb{E} \left[ \prod_{n_{k}=1}^{j} (1+mY_{n_{k}}) \right] \\ = \sum_{j=1}^{\infty} \frac{e^{-\int_{t_{k}}^{t_{k+1}} \lambda_{s} ds} (\int_{t_{k}}^{t_{k+1}} \lambda_{s} ds)^{j}}{j!} \mathbb{E} \left[ \exp \left\{ \sum_{n_{k}=1}^{j} (1+mY_{n_{k}}) \right\} \right] \\ = \sum_{j=1}^{\infty} \frac{e^{-\int_{t_{k}}^{t_{k+1}} \lambda_{s} ds} (\int_{t_{k}}^{t_{k+1}} \lambda_{s} ds)^{j}}{j!} \int_{\mathbb{R}} \exp \left\{ \frac{f_{Q} * f_{Q} * \dots * f_{Q}(x)}{j \text{ items}} \right\} dx,$$

and

$$\mathbb{E} \prod_{n_k=N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k})$$

$$= \sum_{j=1}^{\infty} \frac{e^{-\int_{t_k}^{t_{k+1}} \lambda_s ds} (\int_{t_k}^{t_{k+1}} \lambda_s ds)^j}{j!} \mathbb{E} \left[ \prod_{n_k=1}^{j} (1+mY_{n_k})^2 \right]$$

$$= \sum_{j=1}^{\infty} \frac{e^{-\int_{t_k}^{t_{k+1}} \lambda_s ds} (\int_{t_k}^{t_{k+1}} \lambda_s ds)^j}{j!} \mathbb{E} \left[ \exp \left\{ 2 \sum_{n_k=1}^{j} (1+mY_{n_k}) \right\} \right]$$

$$= \sum_{j=1}^{\infty} \frac{e^{-\int_{t_k}^{t_{k+1}} \lambda_s ds} (\int_{t_k}^{t_{k+1}} \lambda_s ds)^j}{j!} \int_{\mathbb{R}} \exp \left\{ 2 \underbrace{f_Q * f_Q * \dots * f_Q(x)}_{j \text{ items}} \right\} dx.$$

Next we calculate the expectation and variance of the terminal CPPI portfolio value:

**Proposition 5.2.** The expected terminal CPPI portfolio value in discrete trading time case under the jump-diffusion model is

$$\mathbb{E}[V_T] = C_0 \prod_{k=0}^{N-1} \left( m \left[ \exp\left(\mu \frac{T}{N}\right) \mathbb{E} \prod_{n_k = N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k}) \right] - (m-1)e^{rT/N} \right) + G,$$

where  $\mathbb{E} \prod_{n_k=N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k})$  is given by (5.2).

Proof.

$$\mathbb{E}[V_T] = \mathbb{E}[C_T] + G$$

$$= C_0 \mathbb{E}\left[\prod_{k=0}^{N-1} \left(m \frac{S_{t_{k+1}}}{S_{t_k}} - (m-1)e^{rT/N}\right)\right] + G$$

$$= C_0 \prod_{k=0}^{N-1} \left(m \mathbb{E}\frac{S_{t_{k+1}}}{S_{t_k}} - (m-1)e^{rT/N}\right) + G$$

$$= C_0 \prod_{k=0}^{N-1} \left(m \left[\exp\left(\mu \frac{T}{N}\right) \mathbb{E}\prod_{n_k=N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k})\right] - (m-1)e^{rT/N}\right) + G,$$

where  $\mathbb{E} \prod_{n_k=N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k})$  is given by (5.2).

Proposition 5.3. The variance of terminal CPPI portfolio value in discrete time

 $case \ under \ the \ jump-diffusion \ model \ is$ 

$$\begin{aligned} &Var[V_T] = C_0^2 \Bigg[ \prod_{k=0}^{N-1} \left( \Bigg[ m^2 \left( \exp\left(2\mu \frac{T}{N} + \sigma^2 \frac{T}{N}\right) \mathbb{E} \prod_{n_k = N_{t_k}}^{N_{t_{k+1}}} (1 + Y_{n_k})^2 \right) \\ &+ (m-1)^2 e^{2rT/N} - 2m(m-1)e^{rT/N} \left( \exp\left(\mu \frac{T}{N}\right) \mathbb{E} \prod_{n_k = N_{t_k}}^{N_{t_{k+1}}} (1 + Y_{n_k}) \right) \Bigg] \Bigg) \\ &- \prod_{k=0}^{N-1} \left( \Bigg[ m \left( \exp\left(\mu \frac{T}{N}\right) \mathbb{E} \prod_{n_k = N_{t_k}}^{N_{t_{k+1}}} (1 + Y_{n_k}) \right) - (m-1)e^{rT/N} \Bigg] \right)^2 \Bigg], \end{aligned}$$

where  $\mathbb{E}\prod_{n_k=N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k})$  is given by (5.2) and  $\mathbb{E}\prod_{n_k=N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k})^2$  is given by (5.3).

*Proof.* By Lemma 2.7, we have

$$\operatorname{Var}[V_{T}] = \operatorname{Var}[C_{T}] = \operatorname{Var}\left[C_{0}\prod_{k=0}^{N-1} \left(m\frac{S_{t_{k+1}}}{S_{t_{k}}} - (m-1)e^{rT/N}\right)\right]$$
$$= C_{0}^{2}\left[\prod_{k=0}^{N-1} \left(\mathbb{E}\left[m\frac{S_{t_{k+1}}}{S_{t_{k}}} - (m-1)e^{rT/N}\right]^{2}\right) - \prod_{k=0}^{N-1} \left(\mathbb{E}\left[m\frac{S_{t_{k+1}}}{S_{t_{k}}} - (m-1)e^{rT/N}\right]\right)^{2}\right]$$
$$= C_{0}^{2}\left[\prod_{k=0}^{N-1} \left(\left[m^{2}\mathbb{E}\left(\frac{S_{t_{k+1}}}{S_{t_{k}}}\right)^{2} + (m-1)^{2}e^{2rT/N} - 2m(m-1)e^{rT/N}\mathbb{E}\frac{S_{t_{k+1}}}{S_{t_{k}}}\right]\right)$$
95

$$-\prod_{k=0}^{N-1} \left( \left[ m \mathbb{E} \frac{S_{t_{k+1}}}{S_{t_k}} - (m-1)e^{rT/N} \right] \right)^2 \right]$$

$$= C_0^2 \left[ \prod_{k=0}^{N-1} \left( \left[ m^2 \left( \exp\left(2\mu \frac{T}{N} + \sigma^2 \frac{T}{N}\right) \mathbb{E} \prod_{n_k=N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k})^2 \right) + (m-1)^2 e^{2rT/N} - 2m(m-1)e^{rT/N} \left( \exp\left(\mu \frac{T}{N}\right) \mathbb{E} \prod_{n_k=N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k}) \right) \right] \right) - \prod_{k=0}^{N-1} \left( \left[ m \left( \exp\left(\mu \frac{T}{N}\right) \mathbb{E} \prod_{n_k=N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k}) - (m-1)e^{rT/N} \right] \right)^2 \right],$$

where  $\mathbb{E} \prod_{n_k=N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k})$  is given by (5.2) and  $\mathbb{E} \prod_{n_k=N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k})^2$  is given by (5.3).

# 5.3 Measure the Gap risk for CPPI strategies in the jump-diffusion model-the discrete time case

### 5.3.1 Probability of Loss

In practice, suppose that a CPPI-insured portfolio incurs a loss. That is, for some  $t_i \in \tau^N$ ,  $V_{t_i} \leq F_{t_i}$ , which is equivalent to  $C_{t_i} \leq 0$ . We consider the following probabilities:

Definition 5.4. The probability

$$\mathbb{P}_{t_i, t_{i+1}}^{PLL} := \mathbb{P}(V_{t_{i+1}} \le F_{t_{i+1}} | V_{t_i} > F_{t_i})$$
(5.4)

is called the **probability of local loss**.

and

Definition 5.5. The probability

$$\mathbb{P}^{PL} := \mathbb{P}(\text{if for some } t_i \in \tau^N : V_{t_i} \le F_{t_i})$$
(5.5)

is called the **probability of loss**.

*Remarks.* We refer the definition of *probability of local loss* to page 209 in [5].

The following proposition gives a relation between the **probability of local loss** and the **probability of loss**.

**Proposition 5.6.** The probability of loss defined by (5.5) and probability of local loss defined by (5.4) have the following relation:

$$\mathbb{P}^{PL} = 1 - \prod_{i=1}^{N} \left( 1 - \mathbb{P}^{PLL}_{t_{i-1}, t_i} \right).$$
(5.6)

*Proof.* We have

$$\mathbb{P}^{PL} = \mathbb{P}(\text{if for some }, t_i \in \tau^N : V_{t_i} \leq F_{t_i})$$

$$= 1 - \mathbb{P}(\forall t_i \in \tau^N : V_{t_i} > F_{t_i}) = 1 - \mathbb{P}\left(\bigcap_{i=1}^N \{V_{t_i} > F_{t_i} | V_{t_{i-1}} > F_{t_{i-1}}\}\right)$$

$$= 1 - \prod_{i=1}^N \mathbb{P}(\{V_{t_i} > F_{t_i} | V_{t_{i-1}} > F_{t_{i-1}}\})$$

$$= 1 - \prod_{i=1}^N \left(1 - \mathbb{P}(\{V_{t_i} \leq F_{t_i} | V_{t_{i-1}} > F_{t_{i-1}}\})\right) = 1 - \prod_{i=1}^N \left(1 - \mathbb{P}_{t_{i-1}, t_i}^{PLL}\right).$$

#### **Proposition 5.7.** The probability of local loss defined by (5.4) is given by

$$\mathbb{P}_{t_i, t_{i+1}}^{PLL} = \int_{-\infty}^{\ln(\frac{m-1}{m}) + \frac{rT}{N}} p^{(i)}(x) dx, \qquad (5.7)$$

where

$$p^{(i)}(x) = \sum_{j=0}^{\infty} \frac{e^{-\int_0^t \lambda_s ds} (\int_{t_i}^{t_{i+1}} \lambda_s ds)^j}{j!}$$

$$\times \int_{-\infty}^{\infty} \phi\left(x - y; \int_0^t \left(\mu_s - \frac{\sigma_s^2}{97^2}\right) ds, \int_0^t \sigma_s^2 ds\right) f_Q^{(j)}(y) dy,$$
(5.8)

where 
$$f_Q^{(j)}(y) = \underbrace{f_Q(y) * f_Q(y) * \dots f_Q(y)}_{Convolved \ j \ times}$$
.

*Proof.* We have

$$\begin{split} \mathbb{P}_{t_i, t_{i+1}}^{PLL} &= \mathbb{P}(V_{t_{i+1}} \le F_{t_{i+1}} | V_{t_i} > F_{t_i}) = \mathbb{P}(C_{i+1} \le 0 | C_i > 0) \\ &= \mathbb{P}\left(m \frac{S_{t_{i+1}}}{S_{t_i}} - (m-1)e^{rT/N} \right) \le 0\right) \\ &= \mathbb{P}\left(m \exp\left[\int_{t_i}^{t_{i+1}} \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds + \int_{t_i}^{t_{i+1}} \sigma_s dW_s + \sum_{n=N_{t_i}}^{N_{t_{i+1}}} \ln(1+Y_n)\right] \\ &- (m-1)e^{rT/N} \le 0\right) \\ &= \mathbb{P}\left(\int_{t_i}^{t_{i+1}} \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds + \int_{t_i}^{t_{i+1}} \sigma_s dW_s + \sum_{n=N_{t_i}}^{N_{t_{i+1}}} \ln(1+Y_n) \le \ln\left(\frac{m-1}{m}\right) + \frac{rT}{N}\right). \end{split}$$

The proof of Proposition 3.2 shows the density function  $p^{(i)}(x)$  of

$$\int_{t_i}^{t_{i+1}} \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds + \int_{t_i}^{t_{i+1}} \sigma_s dW_s + \sum_{n=N_{t_i}}^{N_{t_{i+1}}} \ln(1+Y_n)$$

is

$$p^{(i)}(x) = \sum_{j=0}^{\infty} \frac{e^{-\int_0^t \lambda_s ds} \left(\int_{t_i}^{t_{i+1}} \lambda_s ds\right)^j}{j!}$$
$$\times \int_{-\infty}^{\infty} \phi\left(x-y; \int_0^t \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds, \int_0^t \sigma_s^2 ds\right) f_Q^{(j)}(y) dy,$$

where  $f_Q^{(j)}(y) = \underbrace{f_Q(y) * f_Q(y) * \dots f_Q(y)}_{j \text{ terms}}$ .

Thus,

$$\mathbb{P}_{t_i, t_{i+1}}^{PLL} = \mathbb{P}(V_{t_{i+1}} \le F_{t_{i+1}} | V_{t_i} > F_{t_i})$$

$$= \mathbb{P}\left(\int_{t_i}^{t_{i+1}} \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds + \int_{t_i}^{t_{i+1}} \sigma_s dW_s + \sum_{n=N_{t_i}}^{N_{t_{i+1}}} \ln(1+Y_n) \le \ln\left(\frac{m-1}{m}\right) + \frac{rT}{N}\right)$$
$$= \int_{-\infty}^{\ln\left(\frac{m-1}{m}\right) + \frac{rT}{N}} p^{(i)}(x) dx$$

By (5.6), the probability of loss  $\mathbb{P}^{PL}$  can be obtained.

#### 5.3.2 Expected Loss

Suppose that the first loss takes place at  $\tau$ . I.e.  $C_{\tau} \leq 0$ . We let  $\tau = \infty$  if a loss never happens. i.e.

$$\begin{aligned} \tau &= t_i \text{ if } V_{t_i} \leq F_{t_i} \text{ and } V_{t_j} > F_{t_j} \text{ for } j = 0, 1, 2, ..., i - 1; \\ \tau &= +\infty \text{ if } V_{t_j} > F_{t_j} \text{ for } j = 0, 1, 2, ..., N. \end{aligned}$$

Since  $V_0 > F_0$ , then

$$\tau = +\infty$$
 if  $V_{t_i} > F_{t_i}$  for  $j = 0, 1, 2, ..., N$ 

which is equivalent to

$$\tau = +\infty$$
 if  $V_{t_j} > F_{t_j}$  for  $j = 1, 2, ..., N$ .

By the definition,  $\tau$  is a stopping time.

We consider the following situation. If a loss happens at time  $\tau$ , the cushion  $C_{\tau} \leq 0$ . If we do not allow the short-sell, then at this trading time  $\tau$ , we take the exposure to be 0. Let

$$\varepsilon_{t_i} = C_0 \prod_{k=0}^{i-1} \left( m \frac{S_{t_{k+1}}}{S_{t_k}} - (m-1)e^{rT/N} \right), \tag{5.9}$$

where

$$\frac{S_{t_{k+1}}}{S_{t_k}} = \exp\left[\int_{t_k}^{t_{k+1}} \left(\mu_s - \frac{\sigma_s^2}{2}\right) ds + \int_{t_k}^{t_{k+1}} \sigma_s dW_s + \sum_{n_k=N_{t_k}}^{N_{t_{k+1}}} \ln(1+Y_{n_k})\right].$$

Then the discounted cushion is

$$C_T^* = \exp(-rT)\varepsilon_{t_N}\chi_{\tau>T} + \exp(-rT)\sum_{j=0}^{N-1}\varepsilon_{t_j}\chi_{\tau=t_j}.$$
(5.10)

**Definition 5.8.** If a loss happens, the conditional expectation of the discounted cushion is called the **conditional expected loss** and we denote this by  $\mathbb{E}[C_T^*|\tau \leq T]$ . The expectation of the discounted cushion is called the **unconditional expected loss** and we use  $\mathbb{E}[C_T^*\chi_{\tau \leq T}]$  to represent it.

The following proposition gives the distribution of the break time  $\tau$ .

**Proposition 5.9.** The distribution of  $\tau$  defined above is

$$\mathbb{P}(\tau = t_i) = \mathbb{P}_{t_i-1, t_i}^{PLL} \times \prod_{j=1}^{i-1} \left( 1 - \mathbb{P}_{t_{j-1}, t_j}^{PLL} \right)$$
(5.11)

*Remarks.* In the above, if j - 1 < 0, let  $\mathbb{P}_{t_{j-1}, t_j}^{PLL} = 0$ . In this case,  $\mathbb{P}(\tau = t_0) = 0$  as expected.

Proof.

$$\begin{split} & \mathbb{P}(\tau = t_i) = \mathbb{P}(V_{t_i} \le F_{t_i}, \text{ and } V_{t_j} > F_{t_j} \text{ for } j = 1, 2, ..., i - 1) \\ &= \mathbb{P}\left(\{V_{t_i} \le F_{t_i} | V_{t_{i-1}} > F_{t_{i-1}}\} \bigcap \bigcap_{j=1}^{i-1} \{V_{t_j} > F_{t_j} | V_{t_{j-1}} > F_{t_{j-1}}\}\right) \\ &= \mathbb{P}\left(\{V_{t_i} \le F_{t_i} | V_{t_{i-1}} > F_{t_{i-1}}\}\right) \times \prod_{j=1}^{i-1} \mathbb{P}(\{V_{t_j} > F_{t_j} | V_{t_{j-1}} > F_{t_{j-1}}\}\right) \\ &= \mathbb{P}\left(\{V_{t_i} \le F_{t_i} | V_{t_{i-1}} > F_{t_{i-1}}\}\right) \times \prod_{j=1}^{i-1} (1 - \mathbb{P}(\{V_{t_j} \le F_{t_j} | V_{t_{j-1}} > F_{t_{j-1}}\})\right) \\ &= \mathbb{P}_{t_i-1, t_i} \times \prod_{j=1}^{i-1} \left(1 - \mathbb{P}_{t_{j-1}, t_j}^{PLL}\right). \end{split}$$

н				
L				
L.	_	_	_	

#### Lemma 5.10.

$$\mathbb{E}[\varepsilon_{t_i}] = C_0 \prod_{k=0}^{i-1} \left( m \left[ \exp\left(\mu \frac{T}{N}\right) \mathbb{E} \prod_{n_k = N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k}) \right] - (m-1)e^{rT/N} \right) + G.$$
(5.12)

where  $\mathbb{E} \prod_{n_k=N_{t_k}}^{N_{t_{k+1}}} (1+Y_{n_k})$  is given by (5.2).

*Proof.* This is an corollary of Proposition 4.2 when substitute i to N.

**Proposition 5.11.** The expectation of loss conditional on the fact that a loss occur is

$$\mathbb{E}[C_T^*|\tau \le T] = \frac{\exp(-rT)\sum_{j=0}^{N-1} \mathbb{E}[\varepsilon_{t_j}]\mathbb{P}[\tau = t_j]}{\sum_{i=1}^N \mathbb{P}[\tau = t_i]}$$
(5.13)

and the unconditional expected loss satisfies

$$\mathbb{E}[C_T^*\chi_{\tau \le T}] = \exp(-rT) \sum_{j=0}^{N-1} \mathbb{E}[\varepsilon_{t_j}]\mathbb{P}[\tau = t_j], \qquad (5.14)$$

where  $\mathbb{E}[\varepsilon_{t_j}]$  is given by (5.12) and  $\mathbb{P}[\tau = t_j]$  is given by (5.11).

Proof.

$$\mathbb{E}[C_T^*\chi_{\tau \le T}] = \mathbb{E}\left[\exp(-rT)\sum_{j=0}^{N-1}\varepsilon_{t_j}\chi_{\tau=t_j}\right]$$
$$= \exp(-rT)\sum_{j=0}^{N-1}\mathbb{E}[\varepsilon_{t_j}\chi_{\tau=t_j}] = \exp(-rT)\sum_{j=0}^{N-1}\mathbb{E}[\varepsilon_{t_j}]\mathbb{P}[\tau=t_j].$$

 $\mathbb{E}[\varepsilon_{t_j}]$  is given by (5.12) and  $\mathbb{P}[\tau = t_j]$  is given by (5.11). Thus we prove (5.14). Moreover, by the property of conditional expectation, we have

$$\mathbb{E}[C_T^*|\tau \le T] = \frac{\mathbb{E}[C_T^*\chi_{\tau \le T}]}{\mathbb{P}[\tau \le T]} = \frac{\exp(-rT)\sum_{j=0}^{N-1}\mathbb{E}[\varepsilon_{t_j}]\mathbb{P}[\tau = t_j]}{\sum_{i=1}^N\mathbb{P}[\tau = t_i]}.$$

This is (5.13).

н		
L		
н		

#### 5.3.3 Loss Distribution

In order to compute risk measures, we utilize the distribution function of the loss. We compute, for x < 0, the quantity

$$\mathbb{P}[C_T^* < x | \tau \le T]. \tag{5.15}$$

We call it the **Loss Distribution**. For x < 0, the quantity

$$\mathbb{P}[C_T^* \chi_{\tau \le T} < x] \tag{5.16}$$

is called **unconditional loss distribution**.

**Proposition 5.12.** Let the density of  $\ln(1 + Y_n)$  be  $f_Q(y)$  and  $C_T^*$  be the discounted cushion. For x < 0, the unconditional loss distribution is

$$\mathbb{P}[C_T^*\chi_{\tau \le T} < x] = \sum_{j=0}^{N-1} \left[ \int_{y_{i-1}}^{+\infty} \int_{y_{i-2}}^{+\infty} \dots \int_{y_0}^{+\infty} \sum_{k=0}^{i-2} \ln\left(mx_k - (m-1)e^{rT/N}\right) + \ln\left(-\left(mx_{i-1} - (m-1)e^{rT/N}\right)\right) p^{(0)}(x) dx_0 p^{(1)}(x_1) dx_1 \qquad (5.17)$$
$$\dots p^{(i-2)}(x_{i-2}) dx_{i-2} p^{(i-1)}(x_{i-1}) dx_{i-1} \mathbb{P}[\tau = t_j] \right]$$

and the loss distribution is

$$\mathbb{P}[C_T^* < x | \tau \le T] = \frac{\mathbb{P}[C_T^* \chi_{\tau \le T} < x]}{\sum_{i=1}^N \mathbb{P}[\tau = t_i]},$$
(5.18)

where  $p^{(i)}(x)$  is given by (5.8) and  $\mathbb{P}[\tau = t_j]$  is given by (5.11) and

$$(y_0, y_1, y_2, \dots y_{i-1}) \in \begin{cases} (y_0, y_1, y_2, \dots y_{i-1}) \in \mathbb{R}^i \\ y_k > \frac{m-1}{m} e^{rT/N} \text{ for } k = 0, 1, 2, \dots i-2 \end{cases}$$

$$\sum_{k=0}^{i-2} \ln(my_k - (m-1)e^{rT/N}) + \ln\left(-\left(my_{i-1} - (m-1)e^{rT/N}\right)\right) > \ln\frac{-x}{C_0} + rT \bigg\}.$$

*Proof.* We have

$$\mathbb{P}[C_T^* \chi_{\tau \le T} < x] = \mathbb{P}\left[\exp(-rT)\sum_{j=0}^{N-1} \varepsilon_{t_j} < x\right]$$
$$= \sum_{j=0}^{N-1} \mathbb{P}[\exp(-rT)\varepsilon_{t_j} < x|\tau = t_j]\mathbb{P}[\tau = t_j]].$$

We now calculate  $\mathbb{P}[\exp(-rT)\varepsilon_{t_j} < x | \tau = t_j].$ 

$$\begin{split} \mathbb{P}[\exp(-rT)\varepsilon_{t_{j}} < x|\tau = t_{j}] \\ &= \mathbb{P}\left[\exp(-rT)\prod_{k=0}^{i-1}C_{0}\left(m\frac{S_{t_{k+1}}}{S_{t_{k}}} - (m-1)e^{rT/N}\right) < x|\tau = t_{j}\right] \\ &= \mathbb{P}\left[\exp(-rT)\left(\prod_{k=0}^{i-2}C_{0}\left(m\frac{S_{t_{k+1}}}{S_{t_{k}}} - (m-1)e^{rT/N}\right)\right)\right) \\ &\left(m\frac{S_{t_{i+1}}}{S_{t_{i}}} - (m-1)e^{rT/N}\right) < x|\tau = t_{j}\right] \\ &= \mathbb{P}\left[\exp(-rT)\left(\prod_{k=0}^{i-2}C_{0}\left(m\frac{S_{t_{k+1}}}{S_{t_{k}}} - (m-1)e^{rT/N}\right)\right)\right) \\ &\left(-\left(m\frac{S_{t_{i+1}}}{S_{t_{i}}} - (m-1)e^{rT/N}\right)\right) > -x|\tau = t_{j}\right] \\ &= \mathbb{P}\left[\sum_{k=0}^{i-2}\ln\left(m\frac{S_{t_{k+1}}}{S_{t_{k}}} - (m-1)e^{rT/N}\right)\right) \\ &+ \ln\left(-\left(m\frac{S_{t_{i+1}}}{S_{t_{k}}} - (m-1)e^{rT/N}\right)\right) > \ln\frac{-x}{C_{0}} + rT|\tau = \tau_{j}\right] \\ &= \int_{y_{i-1}}^{+\infty}\int_{y_{i-2}}^{+\infty}\dots\int_{y_{0}}^{+\infty}\sum_{k=0}^{i-2}\ln(mx_{k} - (m-1)e^{rT/N}) \\ &+ \ln\left(-(mx_{i-1} - (m-1)e^{rT/N})\right) \\ &p^{(0)}(x_{0})dx_{0}p^{(1)}(x_{1})dx_{1}\dots p^{(i-2)}(x_{i-2})dx_{i-2}p^{(i-1)}(x_{i-1})dx_{i-1}, \end{split}$$

where

$$(y_0, y_1, y_2, \dots y_{i-1}) \in \left\{ (y_0, y_1, y_2, \dots y_{i-1}) \in \mathbb{R}^i \\ y_k > \frac{m-1}{m} e^{rT/N} \text{ for } k = 0, 1, 2, \dots i-2 \\ \sum_{k=0}^{i-2} \ln \left( my_k - (m-1)e^{rT/N} \right) \\ + \ln \left( - \left( my_{i-1} - (m-1)e^{rT/N} \right) \right) > \ln \frac{-x}{C_0} + rT \right\}.$$

Thus,

$$\mathbb{P}[C_T^*\chi_{\tau \le T} < x] = \mathbb{P}\left[\exp(-rT)\sum_{j=0}^{N-1}\varepsilon_{t_j} < x\right]$$
$$= \sum_{j=0}^{N-1}\left[\int_{y_{i-1}}^{+\infty}\int_{y_{i-2}}^{+\infty}\dots\int_{y_0}^{+\infty}\sum_{k=0}^{i-2}\ln(mx_k - (m-1)e^{rT/N}) + \ln(-(mx_{i-1} - (m-1)e^{rT/N}))p^{(0)}(x_0)dx_0p^{(1)}(x_1)dx_1 \\\dots p^{(i-2)}(x_{i-2})dx_{i-2}p^{(i-1)}(x_{i-1})dx_{i-1}\mathbb{P}[\tau = t_j]\right].$$

Thus, we obtain (5.17). Through the property of conditional probability, we obtain (5.18).  $\hfill \square$ 

#### 5.3.4 Conclusion

The definition of probability of loss, expected loss and loss distribution in the jumpdiffusion model with discrete trading time is corresponding to the continuous trading time case.

### 5.4 Conditional Floor and Conditional Multiple of CPPI under Jump-diffusion Model in Discrete Trading Time

#### 5.4.1 Introduction

In this section we study the conditional floor and conditional multiple from three aspects: the probability of loss, expected loss and loss distribution.

#### 5.4.2 Probability of Loss

Similar to proposition 1 in [2], we have the following proposition.

**Proposition 5.13.** The condition  $C_{t_k} > 0$  is satisfied at any time  $t_k$  of the management period with probability 1 if and only if:

$$1 - e^{-rT/N} \min_{k=0,1,\dots,N-1} \frac{S_{t_{k+1}}}{S_{t_k}} < \frac{1}{m}.$$
(5.19)

*Proof.*  $C_{t_i}$  has the relation in (5.1).

$$C_{t_{k+1}} = C_{t_k} \left( m \frac{S_{t_{k+1}}}{S_{t_k}} - (m-1)e^{rT/N} \right).$$

The condition  $C_{t_k} > 0$  is true for any time  $t_k$ , if and only if

$$m\frac{S_{t_{k+1}}}{S_{t_k}} - (m-1)e^{rT/N} > 0$$

for all  $k = 0, 1, \dots N - 1$ . This is equivalent to

$$1 - e^{-rT/N} \frac{S_{t_{k+1}}}{S_{t_k}} < \frac{1}{m}$$

for all k = 0, 1, ..., N - 1 or equivalent to

$$1 - e^{-rT/N} \min_{k=0, 1, \dots, N-1} \frac{S_{t_{k+1}}}{S_{t_k}} < \frac{1}{m}.$$

**Proposition 5.14.** The probability of loss defined by (5.5) and probability of local loss defined by (5.4) are monotone increasing functions of the multiple m. Moreover both of them are irrelevant with the floor  $F_t$ .

*Proof.* We have proved the probability of local loss defined by (5.4) is given by

$$\mathbb{P}_{t_i, t_{i+1}}^{PLL} = \int_{-\infty}^{\ln(\frac{m-1}{m}) + \frac{rT}{N}} p^{(i)}(x) dx.$$

$$\begin{split} m \text{ increased} & \Longrightarrow \ln(\frac{m-1}{m}) \text{ increased} \\ & \Longrightarrow \int_{-\infty}^{\ln(\frac{m-1}{m}) + \frac{rT}{N}} p^{(i)}(x) dx \text{ increased for each } i = 0, 1, \dots N-1 \\ & \Longrightarrow \mathbb{P}_{t_i, t_{i+1}}^{PLL} \text{ increased for each } i = 0, 1, \dots N-1 \\ & \text{ and by (5.6) implies } \mathbb{P}^{PL} \text{ increased.} \end{split}$$

From the expressions in (5.4), (5.5) and (5.6), we see both of them are irrelevant with the floor  $F_t$ .

Similar to the Value-at-Risk (VaR) concept (See [27]), and as in the continuous trading time case, we define:

**Definition 5.15.** For  $\epsilon > 0$ , the multiple  $m = m_0$  which satisfies

$$\mathbb{P}[\exists t_i \in \tau^N : V_{t_i} \le F_{t_i}] = \epsilon$$

is called the  $\epsilon$ -conditional multiple.

 $m_0$  can be treated as a quantile. Since the probability of loss is monotone increasing as a function of the multiple m, then for  $m < m_0$ ,

$$\mathbb{P}\left[\exists t_i \in \tau^N : V_{t_i} \le F_{t_i}\right] < \epsilon.$$

For

$$\mathbb{P}^{PL} = 1 - \prod_{i=1}^{N} \left( 1 - \mathbb{P}^{PLL}_{t_{i-1}, t_i} \right) = \epsilon,$$

if we assume all the probability of local losses are the same, we obtain

$$\mathbb{P}_{t_{i-1},t_i}^{PLL} = 1 - (1-\epsilon)^{\frac{1}{N}}.$$

From (5.7), we obtain the expression for  $m_0$ .

#### 5.4.3 Expected Loss

First we have, from (5.13) and (5.14), the following proposition:

**Proposition 5.16.** Given a fixed multiple, both the conditional expected loss given by (5.13) and the unconditional expected loss given by (5.14) are monotone decreasing functions of the initial floor  $F_0$ .

*Proof.* They are direct consequences of (5.13) and (5.14).

Similar to the  $\epsilon$ -conditional multiple, we define following two concepts.

**Definition 5.17.** For  $\rho < 0$  and  $m = m_0$ , the floor  $F_0 = F^{c1}$  which satisfies

$$\mathbb{E}[C_T^*|\tau \le T] = \varrho$$

is called the first type  $\rho$ -m<sub>0</sub>-conditional floor.

and

**Definition 5.18.** For  $\rho < 0$  and  $m = m_0$ , the floor  $F_0 = F^{c^2}$  which satisfies

$$\mathbb{E}[C_T^*\chi_{\tau\leq T}] = \varrho$$

is called the **Second type**  $\rho$ - $m_0$ -conditional floor.

Similar to (5.13) and (5.14), we can solve for the two conditional floors immediately.

#### 5.4.4 Loss Distribution

**Proposition 5.19.** Given a fixed multiple and x < 0, the expressions (5.17) and (5.18) are monotone increasing function of the initial floor  $F_0$ 

*Proof.* Since  $C_0 = V_0 - F_0$ .

$$F_0 \text{ increased} \Longrightarrow$$
  
 $C_0 \text{ decreased} \Longrightarrow$   
 $\ln(-\frac{x}{C_0}) \text{ increased} \Longrightarrow$ 

That is, both the expressions (5.17) and (5.18) are increasing.

Next we define the following two concepts.

**Definition 5.20.** For  $\epsilon > 0$ ,  $\rho < 0$  and  $m = m_0$ , the floor  $F_0 = F^{c3}$  which satisfies

$$\mathbb{E}[C_T^*\chi_{\tau\leq T} < \varrho] = \epsilon$$

is called the **third type**  $\epsilon$ - $\rho$ - $m_0$ -conditional floor.

and

**Definition 5.21.** For  $\epsilon > 0$ ,  $\rho < 0$  and  $m = m_0$ , the floor  $F_0 = F^{c4}$  which satisfies

$$\mathbb{E}[C_T^* < \varrho | \tau \le T] = \epsilon$$

is called the fourth type  $\epsilon - \rho - m_0$ -conditional floor.

The above two conditional floors are useful in numerical computations.

### 5.5 Convergence

In this section, we consider the relation between the case when the trading time is continuous and the case when the trading time is disccrete.

Recall (5.1).

$$C_{t_{k+1}} = C_{t_k} \left( m \frac{S_{t_{k+1}}}{S_{t_k}} - (m-1)e^{rT/N} \right),$$

When  $N \to \infty$ ,  $\Delta t = T/N \to 0$ 

$$\exp\left(r\frac{T}{N}\right) \sim 1 + r\frac{T}{N}.$$

Thus, we got

$$\frac{C_{t_{k+1}} - C_{t_k}}{C_{t_k}} + 1 = \left(m\left[\frac{S_{t_{k+1}} - S_{t_k}}{S_{t_k}} + 1\right] - (m-1)e^{rT/N}\right).$$

Let  $N \to \infty$ , we have

$$\frac{dC_t}{C_{t-}} + 1 = \left(m\left[\frac{dS_t}{S_{t-}} + 1\right] - (m-1)(1+rdt)\right),\,$$

and this is equivalent to

$$\frac{dC_t}{C_{t-}} = \left( m \frac{dS_t}{S_{t-}} - (m-1)(rdt) \right).$$

This is consistant with the continuous case (3.14). We have the following proposition:

**Proposition 5.22.** For  $N \to \infty$ , the portfolio value in discrete trading time converges a.s. to the portfolio value in continuous trading time.

# Chapter 6

# Stochastic and dynamic floors

#### 6.1 Introduction

In section 6.2, we will consider the case of stochastic floor which is equal to the maximum of its past value and a given percentage of the portfolio value. The idea is that when the portfolio value is large enough, we will increase the level of the floor. Both the continuous and discrete trading time cases will be analyzed. We will also calculate the distribution of the time.

In section 6.3, we will consider the case of stochastic floor which is indexed by the given portfolio performance. The idea is similar to that as in section 6.2. We will also calculate the distribution of the first-time-change of the floor.

In section 6.4, we will deal with Ratchet and Margin CPPI strategies with the timechange of strategy defined on the exposition variance. We will show that in the discrete trading time case, the Ratchet CPPI is equivalent to the stochastic floor which is indexed by the given portfolio performance. In the cases of CPPI with margin when the floor is close to the portfolio value, the exposure will be very small and we will reduce the floor. We will discuss the distribution of the first-change-time of the floor when the trading time is continuous.

# 6.2 When the floor equals to the maximum of its past value and a given percentage of the port-folio value

In this section, the current floor value is the maximum of the past floor value and a given percentage of the current portfolio value.

#### 6.2.1 Discrete-time case with fixed multiple

Let

$$\tau^n = \{ t_0 = 0 < t_1 < t_2 < \dots < t_n = T \}$$

denote a sequence of equidistant refinements of the interval [0, T], where  $t_{k+1} - t_k = \frac{T}{n} =: \Delta$  for k = 0, ..., n - 1. Let

$$F_{t_k} = \max\{F_{t_{k-1}}\exp(r\Delta), \quad xV_{t_k}\}$$

and the initial floor  $F_0 = Ge^{-rT}$  be the same as before and suppose x is an arbitrary but fixed percentage of the portfolio value. This definition means that the floor is equal to the maximum of its past value and a given percentage of the portfolio value. As the portfolio value increases and if we keep the floor unchanged, the cushion will be very big. Our idea is that as the portfolio value increase to a specific level, we will also increase the level of the floor. In general, we assume  $xV_0 \leq F_0$ .

Let  $T_1 = \min\{t > 0 : F_t = xV_t\}$ . Denote respectively by  $\theta_0^B$  and  $\theta_0^S$  the shares invested

on the riskless and risky assets. We have:

$$\theta_0^B = (V_0 - \theta_0^S S_0)/B_0,$$
  
 $\theta_0^S = m(V_0 - F_0)/S_0.$ 

The following proposition calculates the probability of the first-time-change of the floor taking place at  $t_1$ .

**Proposition 6.1.** For the jump-diffusion model, if we assume  $x_{i+1} = \ln(\frac{S_{i+1}}{S_i})$ , i = 0, 1, 2, ... be i.i.d. and their density function be p(x). Then the probability of the first-time-change of the floor which takes happen at  $t_1$  is

$$\mathbb{P}[T_1 = t_1] = \int_{\ln(e^{r\Delta} \frac{F_0/x - \theta_0^B B_0}{m(V_0 - F_0)})}^{\infty} p(x) dx.$$

Proof.

$$\mathbb{P}[T_{1} = t_{1}] = \mathbb{P}[F_{t_{1}} \le xV_{t_{1}}] = \mathbb{P}[F_{0}e^{r\Delta} \le x(F_{t_{1}} + e_{t_{1}})]$$

$$= \mathbb{P}\left[F_{0}e^{r\Delta} \le x\left(\theta_{0}^{B}B_{0}e^{r\Delta} + m(V_{0} - F_{0})\right) \\ \times \exp\left[\int_{0}^{t_{1}}\left(\mu_{s} - \frac{\sigma_{s}^{2}}{2}\right)ds + \int_{0}^{t_{1}}\sigma_{s}dW_{s} + \sum_{n=1}^{N_{t_{1}}}\ln(1 + Y_{n})\right]\right) \right]$$

$$= \mathbb{P}\left[\exp\left[\int_{0}^{t_{1}}\left(\mu_{s} - \frac{\sigma_{s}^{2}}{2}\right)ds + \int_{0}^{t_{1}}\sigma_{s}dW_{s} + \sum_{n=1}^{N_{t_{1}}}\ln(1 + Y_{n})\right] \ge e^{r\Delta}\frac{F_{0}/x - \theta_{0}^{B}B_{0}}{m(V_{0} - F_{0})}\right]$$

$$= \int_{\ln\left(e^{r\Delta}\frac{F_{0}/x - \theta_{0}^{B}B_{0}}{m(V_{0} - F_{0})}\right)}p(x)dx.$$

ю		
н		

*Remarks.* In the simple CPPI case,  $Y_n = 0$ ,  $\mu_s$  and  $\sigma_s = \sigma$ , then

$$\mathbb{P}[F_{t_1} = xV_{t_1}] = \mathbb{P}\left[\exp\left(\mu - \frac{1}{2}\sigma^2\right)t_1 + \sigma W_{t_1} \ge e^{r\Delta}\frac{F_0/x - \theta_0^B B_0}{m(V_0 - F_0)}\right]$$

$$= 1 - N\left(\frac{1}{\sigma\sqrt{\Delta}}\left(\ln\left[e^{r\Delta}\frac{F_0/x - \theta_0^B B_0}{m(V_0 - F_0)}\right] - \left(\mu - \frac{1}{2}\sigma^2\right)\Delta\right)\right)$$

This is the first part of the proposition 1 on [59].

In the following, we consider the probability that  $T_1 = t_N$ .

**Proposition 6.2.** For the jump-diffusion model, if assume the density function of  $x_i$  is p(x), the probability of the first-time-change of floor at  $t_N$  is

$$\mathbb{P}[T_1 = t_N] = \int \dots \int_{D_N} p(u_i) \dots p(u_N) du_1 du_2 \dots du_N,$$

where

$$(u_1, ..., u_N) \in D_n \text{ iff}$$
  
$$\forall i \leq N - 1, F_0 e^{ri\Delta} > x \left[ F_0 e^{ri\Delta} + C_0 \prod_{j=1}^i g(u_j) \right];$$
  
for  $i = N, F_0 e^{rN\Delta} \leq x \left[ F_0 e^{rN\Delta} + C_0 \prod_{j=1}^i g(u_N) \right].$ 

Proof. We have

$$\mathbb{P}[T_1 = t_N] = P[F_{t_1} > xV_{t_1}, ..., F_{t_{N-1}} > xV_{t_{N-1}}, F_{t_N} \le xV_{t_N}]$$

and

$$V_{t_i} = \theta_{t_{i-1}}^B B_{t_i} + \theta_{t_{i-1}}^S S_{t_i} = F_{t_i} + C_{t_i}$$
  
=  $F_0 e^{rt_i} + C_0 \prod_{t=1}^{t_i} \left[ 1 + (1-m) \frac{B_t - B_{t-1}}{B_{t-1}} + m \frac{S_t - S_{t-1}}{S_{t-1}} \right].$ 

Let  $g(x) = 1 + (1 - m)(e^{r\Delta} - 1) + m(e^x - 1)$ , then

$$V_{t_i} = F_0 e^{rt_i} + C_0 \prod_{t=1}^{t_i} g(x_t).$$

Thus,

$$\mathbb{P}[T_1 = t_N] = \mathbb{P}[F_{t_1} > xV_{t_1}, ..., F_{t_{N-1}} > xV_{t_{N-1}}, F_{t_N} \le xV_{t_N}]$$
  
=  $\mathbb{P}\left[F_0 e^{r\Delta} > x[F_0 e^{r\Delta} + C_0 g(x_1)], ..., F_0 e^{r(N-1)\Delta} > x\left[F_0 e^{r(N-1)\Delta} + C_0 \prod_{t=1}^{t_{N-1}} g(x_t)\right], F_0 e^{rN\Delta} \le x\left[F_0 e^{rN\Delta} + C_0 \prod_{t=1}^{t_N} g(x_t)\right]\right].$ 

Let

$$(u_1, ..., u_N) \in D_n \text{ iff}$$
  

$$\forall i \leq N - 1, F_0 e^{ri\Delta} > x \left[ F_0 e^{ri\Delta} + C_0 \prod_{j=1}^i g(u_j) \right];$$
  
for  $i = N, F_0 e^{rN\Delta} \leq x \left[ F_0 e^{rN\Delta} + C_0 \prod_{j=1}^i g(u_N) \right].$ 

For the jump-diffusion model, when the density function of  $x_i$  is p(x), we have

$$\mathbb{P}[T_1 = t_N] = \int \dots \int_{D_N} p(u_i) \dots p(u_N) du_1 du_2 \dots du_N.$$

*Remarks.* The second part of proposition 1 on [59] is a special case. Also, when the density function p(x) of  $x_i$  is given, the associated probability can be calculated explicitly.

Next we have the following proposition (see also [59]):

**Proposition 6.3.** For any  $t_i$ , the stochastic floor F is equal to the stochastic floor Q defined by:

$$Q_{t_i} = \max\left[\tilde{F}_{t_i}, \sup_{j \le i} e^{r(t_i - t_j)} V_{t_j}\right].$$

*Proof.* (1) Firstly, the stochastic floor F is above the deterministic floor  $\tilde{F}$ 

$$F_{t_i} \ge \tilde{F}_{t_i} = P_0 e^{rt_i},$$

and secondly, we have:

$$F_{t_i} \ge x \sup_{j \le i} e^{r(t_i - t_j)} V_{t_j}.$$

Indeed, by recursion we have:

$$F_{t_i} \geq e^{r\delta} F_{t_{i-1}} \text{ and } F_{t_i} \geq x V_{t_i},$$
  
$$F_{t_{i-1}} \geq e^{r\delta} F_{t_{i-2}} \text{ and } F_{t_{i-1}} \geq x V_{t_{i-1}}.$$

Thus,

$$F_{t_i} \ge \max(e^{r\delta} V_{t_{i-1}}; V_{t_i}),$$

which, by iteration, leads to the inequality  $F_{t_i} \ge Q_{t_i}$ .

(2) Conversely, if  $F_{t_i} = xV_{t_i}$ , then

$$V_{t_i} = \sup_{j \le i} e^{r(t_i - t_j)} V_{t_j}.$$

Therefore, since we have  $Q_{t_i} \ge e^{r(t_i - t_j)} V_{t_j}$  for all  $j \le i$ , we deduce that  $F_{t_i} \le Q_{t_i}$ .  $\Box$ 

The proposition shows that the previous CPPI strategy with floor F is the discretetime version of Time Invariant Portfolio Protection strategy(TIPP).

#### 6.2.2 Continuous-time case with a fixed multiple

As in the previous section, when the current floor value is the maximum of the past floor value and a given percentage of the current portfolio value, the strategy is equivalent to the TIPP strategy. Standard convergence results lead to the following model, in continuous-time:

$$F_t = \max \left[ \tilde{F}_t, x \sup_{s \le t} e^{r(t-s)} V_s \right],$$
  
$$e_t = mC_t = m(V_t - F_t).$$

Define

$$T_1^c = \inf \left[ t \le T : F_t = x \sup_{s \le t} e^{r(t-s)} V_s \right].$$

This is **the first-time-change** of floor. We will consider the probability distribution of  $T_1^c$ .

Before  $T_1^c$ , we have

$$V_t = C_0 \exp\left\{\int_0^t \left(r + m(\mu_s - r) - \frac{m^2 \sigma_s^2}{2}\right) ds + \int_0^t m \sigma_s dW_s + \sum_{n=1}^{N_t} \ln(1 + mY_n)\right\} + F_0 e^{rt}.$$

Denote

$$X_t = \int_0^t \left( m(\mu_s - r) - \frac{m^2 \sigma_s^2}{2} \right) ds + \int_0^t m \sigma_s dW_s + \sum_{n=1}^{N_t} \ln(1 + mY_n).$$

Thus, we have

$$T_1^c = \inf \left[ t \le T : F_t = x \sup_{s \le t} e^{r(t-s)} V_s \right]$$
  
= 
$$\inf \left[ t \le T : x \sup_{s \le t} e^{r(t-s)} V_s = x \left( C_0 e^{rt} \exp \left\{ \sup_{s \le t} X_s \right\} + e^{rt} F_0 \right) = F_0 e^{rt} \right]$$
  
117

$$= \inf \left[ t \le T : \sup_{s \le t} X_s \ge \ln \left[ \frac{F_0}{C_0} \left( \frac{1}{x} - 1 \right) \right] \right].$$

When  $\mu_s = \mu$  and  $\sigma_s = \sigma$  is constant and then

$$X_t = \left(m(\mu - r) - \frac{1}{2}m^2\sigma^2\right)t + m\sigma W_t + \sum_{n=1}^{N_t}\ln(1 + mY_n).$$

Let

$$A = \frac{(\mu - r) - \frac{1}{2}m\sigma^2}{\sigma}$$

and

$$W_s^{(A)} = \left(As + W_s + \frac{1}{m\sigma}\sum_{n=1}^{N_t}\ln(1+mY_n)\right).$$

Then, we can calculate the distribution of  $\sup_{s\leq t} W^{(A)}_s$  is

$$\begin{split} & \mathbb{P}\left(\sup_{s \leq t} W_s^{(A)} \leq y\right) \\ &= \mathbb{P}\left[\bigcup_{k=1}^{\infty} \left(\sup_{s \leq t} (As + W_s) + \frac{1}{m\sigma} \sum_{n=1}^{N_t} \ln(1 + mY_n) \leq y, N_t = k\right)\right] \\ &= \sum_{k=1}^{\infty} \mathbb{P}\left(\sup_{s \leq t} (As + W_s) + \frac{1}{m\sigma} \sum_{n=1}^{N_t} \ln(1 + mY_n) \leq y | N_t = k\right) \mathbb{P}(N_t = k) \\ &= \sum_{k=1}^{\infty} \frac{\mathbb{P}\left(\sup_{s \leq t} (As + W_s) + \frac{1}{m\sigma} \sum_{n=1}^{N_t} \ln(1 + mY_n) \leq y, N_t = k\right)}{\mathbb{P}(N_t = k)} \mathbb{P}(N_t = k) \\ &= \sum_{k=1}^{\infty} \frac{\mathbb{P}\left(\sup_{s \leq t} (As + W_s) + \frac{1}{m\sigma} \sum_{n=1}^{N_t} \ln(1 + mY_n) \leq y\right) \mathbb{P}(N_t = k)}{\mathbb{P}(N_t = k)} \\ &= \sum_{k=1}^{\infty} \mathbb{P}\left(\sup_{s \leq t} (As + W_s) + \frac{1}{m\sigma} \sum_{n=1}^{N_t} \ln(1 + mY_n) \leq y\right) \mathbb{P}(N_t = k) \\ &= \sum_{k=1}^{\infty} \mathbb{P}\left(\sup_{s \leq t} (As + W_s) + \frac{1}{m\sigma} \sum_{n=1}^{k} \ln(1 + mY_n) \leq y\right) \frac{e^{-\int_0^t \lambda_s ds} \left(\int_0^t \lambda_s ds\right)^k}{k!}. \\ &= \sum_{k=1}^{\infty} \mathbb{P}\left(\sup_{s \leq t} (As + W_s) + \frac{1}{m\sigma} \sum_{n=1}^k \ln(1 + mY_n) \leq y\right) \frac{e^{-\int_0^t \lambda_s ds} \left(\int_0^t \lambda_s ds\right)^k}{k!}. \end{split}$$

Recall that a property possessed by the maximum value of the Brownian motion with drift gives:

$$\mathbb{P}\left(\sup_{s\leq t}(As+W_s) + \frac{1}{m\sigma}\sum_{n=1}^k \ln(1+mY_n) \leq y\right)$$
$$= \int_{-\infty}^{\infty} \left(1 - \frac{1}{2}\operatorname{Erfc}\left(\frac{y-y_2}{\sqrt{2t}} - A\frac{\sqrt{t}}{\sqrt{2}}\right)\right)$$
$$-\frac{1}{2}e^{2A(y-y_2)}\operatorname{Erfc}\left(\frac{y-y_2}{\sqrt{2t}} + A\frac{\sqrt{t}}{\sqrt{2}}\right)\right)dF_k(y_2),$$

where the function Erfc is given by:

$$\operatorname{Erfc}(x) = \frac{2}{\sqrt{\pi}} \int_{x}^{\infty} e^{-u^2} du$$

and the  $F_k(y_2)$  is the distribution function of  $\frac{1}{m\sigma}\sum_{n=1}^k \ln(1+mY_n)$ . Then we have

$$\mathbb{P}\left(\sup_{s \le t} W_s^{(A)} \le y\right)$$

$$= \sum_{k=1}^{\infty} \frac{e^{-\int_0^t \lambda_s ds} \left(\int_0^t \lambda_s ds\right)^k}{k!} \int_{-\infty}^{\infty} (1 - \frac{1}{2} \operatorname{Erfc}\left(\frac{y - y_2}{\sqrt{2}t} - A\frac{\sqrt{t}}{\sqrt{2}}\right)$$

$$-\frac{1}{2} e^{2A(y - y_2)} \operatorname{Erfc}\left(\frac{y - y_2}{\sqrt{2}t} + A\frac{\sqrt{t}}{\sqrt{2}}\right)\right) dF_k(y_2).$$

Therefore, we have deduced the following proposition:

**Proposition 6.4.** When assume  $\mu_s = \mu$  and  $\sigma_s = \sigma$  be constant, the cdf of the first time  $T_1^c$  before maturity T at which  $F_t = \sup_{s \le t} e^{r(t-s)} V_s$  is given by:

$$\mathbb{P}[T_1^c \le t] = \sum_{k=1}^{\infty} \frac{e^{-\int_0^t \lambda_s ds} \left(\int_0^t \lambda_s ds\right)^k}{k!} \int_{-\infty}^{\infty} \left(\frac{1}{2} Erfc\left(\frac{y-y_2}{\sqrt{2}t} - A\frac{\sqrt{t}}{\sqrt{2}}\right)\right)$$

$$+\frac{1}{2}e^{2A(y-y_2)}Erfc\left(\frac{y-y_2}{\sqrt{2}t}+A\frac{\sqrt{t}}{\sqrt{2}}\right)\right)dF_k(y_2)$$

Proposition 5 of [59] is a special case for the diffusion model without jump.

#### 6.2.3 Capped CPPI

Assume that the portfolio manager does not want selling short on the money market account (condition  $\theta_t^B \ge 0$ ).

Therefore the exposure e is bounded by a fixed proportion  $\varpi$  of the portfolio value V. We call it the **Capped CPPI**. This leads to the following conditions on the CPPI strategy in continuous time:

(a)

$$F_t = \max\left[\tilde{F}_t, x \sup_{s \le t} e^{r(t-s)} V_s\right]$$

The floor equals to the maximum of its past value and a given percentage of the portfolio value.

(b)

$$e_t = \inf(\varpi V_t, mC_t)$$

We always assume that

$$C_t = V_t - F_t.$$

There are four cases have to be analyzed:

Case 1 (C1):  $F_t = \tilde{F}_t$  and  $e_t = mC_t$  (Standard CPPI); Case 2 (C2):  $F_t = \tilde{F}_t$  and  $e_t = \varpi V_t$  (Standard capped CPPI); Case 3 (C3):  $F_t = x \sup_{s \leq t} e^{r(t-s)V_s}$  and  $e_t \equiv mC_t$ ; Case 4 (C4):  $F_t = x \sup_{s \leq t} e^{r(t-s)V_s}$  and  $e_t = \varpi V_t$ . For (C1): We have  $F_t = \tilde{F}_t$ , Thus,  $\tilde{F}_t \geq xV_t$  or equivalently  $-\tilde{F}_t \leq -xV_t$ . Since  $e_t = mC_t$ , we have  $\varpi V_t \geq mC_t$ . Then  $e_t = mC_t = m(V_t - \tilde{F}_t)$  with  $-\tilde{F}_t \leq -xV_t$ . Therefore, we get:

$$e_t \le m(1-x)V_t.$$

Additionally, since  $C_t \leq (1-x)V_t$  and  $\varpi V_t \geq mC_t$ , we deduce:

$$C_t \le \min\left[(1-x), \frac{\overline{\omega}}{m}\right] V_t.$$

For (C2):

We have  $F_t = \tilde{F}_t$  and  $e_t = \varpi V_t$ . Thus,  $\varpi V_t \leq mC_t$ . Then:

$$\varpi V_t \le m(V_t - \tilde{F}_t) \le m(1 - x)V_t$$

Consequently, we have:

$$\varpi \le m(1-x).$$

Equivalently, if  $F_t = \tilde{F}_t$  and  $\varpi > m(1 - x)$ , then  $e_t = mC_t$ , which means that the TIPP strategy does not need to be capped, in that case.

For (C3) and (C4), whenever there exists a ratchet effect, the portfolio value  $V_t$  satisfies  $V_t = \sup_{s \le t} e^{r(t-s)} V_s$ . we have discuss the (C3) on section 6.2 and for the (C4) we will do it on section 6.4.

# 6.3 CPPI with a floor indexed on a given portfolio performance

In this section, the floor value is indexed accordingly on a given portfolio performance.

#### 6.3.1 Discrete-time with a fixed multiple

For  $\tau \in \{t_0 = 0, ..., t_N = T\}$ , the CPPI strategy is defined as follows. The floor is now assumed to be standard (deterministic) until the portfolio return  $\frac{V_{t_i}}{V_0}$  becomes higher than a deterministic value  $\alpha e^{rt}$  where the coefficient  $\alpha$  is higher than 1. As soon as  $\frac{V_{t_i}}{V_0} > \alpha e^{rt_i}$ , the floor is equal to a fixed proportion  $\beta$  of the portfolio value with  $0 < \beta < 1$ . Therefore, the floor F is determined as follows. Denote by  $T_1^{d,\alpha}$  the first time at which the portfolio return  $\frac{V_{t_i}}{V_0}$  is higher than  $\alpha e^{rt_i}$ .

**Proposition 6.5.** Under above assumption, the time  $T_1^{d,\alpha}$  is characterized by the relation:

$$T_1^{d,\alpha} = \inf\{t_i \le T : V_{t_i} \ge \alpha V_0 e^{rt_i}\}.$$

Thus the floor is given by:

$$F_{t_j} = \tilde{F}_{t_j} = F_0 e^{rt_j} \quad for \quad t_j \le T_1^{d,\alpha};$$
  
$$F_{t_j} = \beta V_{T_1^{d,\alpha}} e^{r(t_j - T_1^{d,\alpha})} \quad for \quad t_j > T_1^{d,\alpha}.$$

In the following, we calculate the the probability of  $T_1^{d,\alpha} = t_1$ .

**Proposition 6.6.** For the jump-diffusion model, if we assume  $x_{i+1} = \ln(\frac{S_{i+1}}{S_i})$ , i = 0, 1, 2, ... is i.i.d. and their density function is p(x), then the probability of the first-time-change of floor which takes happen at  $t_1$  is

$$\mathbb{P}[V_{t_1} \ge \alpha V_0 e^{r\Delta}] = \int_{\ln\left(e^{r\Delta} \frac{\alpha V_0 - \theta_0^B B_0}{m(V_0 - F_0)}\right)}^{\infty} p(x) dx.$$
122

Proof. We have

$$\begin{split} \mathbb{P}[T_{1}^{d,\alpha} = t_{1}] &= \mathbb{P}\left[V_{t_{1}} \geq \alpha V_{0}e^{r\Delta}\right] \\ &= \mathbb{P}\left[\theta_{0}^{B}B_{0}e^{r\Delta} + m(V_{0} - F_{0}) \times \exp\left[\int_{0}^{t_{1}} \left(\mu_{s} - \frac{\sigma_{s}^{2}}{2}\right)ds + \int_{0}^{t_{1}} \sigma_{s}dW_{s} \right. \\ &+ \sum_{n=1}^{N_{t_{1}}} \ln(1 + Y_{n})\right] \geq \alpha V_{0}e^{r\Delta} \\ &= \mathbb{P}\left[\exp\left[\int_{0}^{t_{1}} \left(\mu_{s} - \frac{\sigma_{s}^{2}}{2}\right)ds + \int_{0}^{t_{1}} \sigma_{s}dW_{s} + \sum_{n=1}^{N_{t_{1}}} \ln(1 + Y_{n})\right] \\ &\geq e^{r\Delta}\frac{\alpha V_{0} - \theta_{0}^{B}B_{0}}{m(V_{0} - F_{0})}\right]. \\ &= \int_{\ln\left(e^{r\Delta}\frac{\alpha V_{0} - \theta_{0}^{B}B_{0}}{m(V_{0} - F_{0})}\right)} p(x)dx. \end{split}$$

- 6		
- 1		
- 1		

*Remarks.* For the simple CPPI case,  $Y_n = 0$ ,  $\mu_s$  and  $\sigma_s = \sigma$ , then

$$\mathbb{P}\left[V_{t_1} \ge \alpha V_0 e^{r\Delta}\right] = \mathbb{P}\left[\exp\left(\mu - \frac{1}{2}\sigma^2\right)t_1 + \sigma W_{t_1} \ge e^{r\Delta}\frac{F_0/x - \theta_0^B B_0}{m(V_0 - F_0)}\right]$$
$$= 1 - N\left(\frac{1}{\sigma\sqrt{\Delta}}\left(\ln\left[e^{r\Delta}\frac{\alpha V_0 - \theta_0^B B_0}{m(V_0 - F_0)}\right] - \left(\mu - \frac{1}{2}\sigma^2\right)\Delta\right)\right).$$

In the following, we consider the probability that  $T_1 = t_N$ .

**Proposition 6.7.** For the jump-diffusion model, if we assume  $x_{i+1} = \ln\left(\frac{S_{i+1}}{S_i}\right)$ ,  $i = 0, 1, 2, \dots$  is i.i.d. and their density function is p(x), then the probability of the first-time-change of floor which takes happen at  $t_N$  is

$$\mathbb{P}\left[T_1^{d,\alpha} = t_N\right] = \int \dots \int_{D_N} p(u_i) \dots p(u_N) du_1 du_2 \dots du_N,$$

where

$$(u_1, ..., u_N) \in D_n \text{ iff}$$
  
$$\forall i \leq N - 1, F_0 e^{ri\Delta} + C_0 \prod_{t=1}^{t_i} g(x_t) < \alpha V_0 e^{ri\Delta};$$
  
for  $i = N, F_0 e^{rN\Delta} + C_0 \prod_{t=1}^{t_N} g(x_t) \geq \alpha V_0 e^{rN\Delta}.$ 

*Proof.* We have

$$\mathbb{P}[T_1 = t_N] = \mathbb{P}\left[F_{t_1} > xV_{t_1}, .., F_{t_{N-1}} > xV_{t_{N-1}}, F_{t_N} \le xV_{t_N}\right]$$

and

$$\begin{aligned} V_{t_i} &= \theta^B_{t_{i-1}} B_{t_i} + \theta^S_{t_{i-1}} S_{t_i} = F_{t_i} + C_{t_i} \\ &= F_0 e^{rt_i} + C_0 \prod_{t=1}^{t_i} \left[ 1 + (1-m) \frac{B_t - B_{t-1}}{B_{t-1}} + m \frac{S_t - S_{t-1}}{S_{t-1}} \right]. \end{aligned}$$

Let  $g(x) = 1 + (1 - m)(e^{r\Delta} - 1) + m(e^x - 1)$ , then

$$V_{t_i} = F_0 e^{rt_i} + C_0 \prod_{t=1}^{t_i} g(x_t)$$

and

$$\mathbb{P}[T_{1}^{d,\alpha} = t_{N}]$$

$$= \mathbb{P}\left[V_{t_{1}} < \alpha V_{0}e^{r\Delta}, ..., V_{t_{N-1}} < \alpha V_{0}e^{r(N-1)\Delta}, V_{t_{N}} \ge \alpha V_{0}e^{rN\Delta}\right]$$

$$= \mathbb{P}\left[F_{0}e^{r\Delta} + C_{0}g(x_{1}) < \alpha V_{0}e^{r\Delta}, ..., F_{0}e^{r(N-1)\Delta} + C_{0}\prod_{t=1}^{t_{N-1}}g(x_{t}) < \alpha V_{0}e^{r(N-1)\Delta}, \right.$$

$$F_{0}e^{rN\Delta} + C_{0}\prod_{t=1}^{t_{N}}g(x_{t}) \ge e^{rN\Delta}\right].$$

Assume

$$(u_1, ..., u_N) \in D_n \text{ iff}$$
  
$$\forall i \le N - 1, \ F_0 e^{ri\Delta} + C_0 \prod_{t=1}^{t_i} g(x_t) < \alpha V_0 e^{ri\Delta};$$
  
For  $i = N, \ F_0 e^{rN\Delta} + C_0 \prod_{t=1}^{t_N} g(x_t) \ge \alpha V_0 e^{rN\Delta};$ 

then, we have

$$\mathbb{P}\left[T_1^{d,\alpha} = t_N\right] = \int \dots \int_{D_N} p(u_i) \dots p(u_N) du_1 du_2 \dots du_N.$$

. 1	_	-	_	٦

#### 6.3.2 Continuous-time case

The floor is now assumed to be standard (deterministic) until the portfolio return  $V_t/V_0$  is higher than a deterministic value of the form  $\alpha e^{rt}$  where the coefficient  $\alpha$  is higher than 1. As soon as  $V_t/V_0 > \alpha e^{rt}$ , the floor is equal to a fixed proportion  $\beta$  of the portfolio value with  $0 < \beta < 1$ . Therefore, the floor F is determined as follows. Denote by  $T_1^{c,\alpha}$  the first time at which the portfolio return  $V_t/V_0$  is higher than  $\alpha e^{rt}$ .

**Proposition 6.8.** Under above assumption, the time  $T_1^{c,\alpha}$  is characterized by the relation:

$$T_1^{c,\alpha} = \inf\{t \le T : V_t \ge \alpha V_0 e^{rt}\}.$$

Thus, the floor is given by:

$$F_t = \tilde{F}_t = F_0 e^{rt} \text{ for } t \le T_1^{c,\alpha};$$
  

$$F_t = \beta V_{T_1^{c,\alpha}} e^{r(t-T_1^{c,\alpha})}_{125} \text{ for } t > T_1^{c,\alpha}$$

The stochastic floor is also defined by:

$$F_t = \tilde{F}_t \chi_{t \le T_1^{c,\alpha}} + \beta V_{T_1^{c,\alpha}} e^{r(t - T_1^{c,\alpha})} \chi_{t > T_1^{c,\alpha}}.$$

We assume that the exposure satisfies:  $e_t = mC_t$ . Therefore at time  $T_1^{c,\alpha}$ , the portfolio value is such that  $V_{T_1^{c,\alpha}} \ge \alpha V_0 e^{rT_1^{c,\alpha}}$ . Thus, at time  $T_1^{c,\alpha}$ , the floor is equal to  $\beta V_{T_1^{c,\alpha}}$  and the cushion is equal to  $(1 - \beta)V_{T_1^{c,\alpha}}$ .

As before we have

**Proposition 6.9.** The portfolio value after  $T_1^{c,\alpha}$   $(T_1^{c,\alpha} < t \le T)$  is

$$(1-\beta)V_{T_1^{c,\alpha}}\exp\left\{\int_{T_1^{c,\alpha}}^t \left((r+m(\mu_s-r))-\frac{m\sigma_s^2}{2}\right)ds\right.\\\left.+\int_{T_1^{c,\alpha}}^t m\sigma_s dW_s\right\}\left[\prod_{n=N_{T_1^{c,\alpha}}}^{N_t} (1+mY_n)\right]+\beta V_{T_1^{c,\alpha}}.$$

and

**Proposition 6.10.** When assume  $\mu_s = \mu$  and  $\sigma_s = \sigma$  be constant, the cumulative distribution function of  $T_1^{c,\alpha}$  is given by

$$\mathbb{P}(T_{1}^{c,\alpha} \leq t) = 1 - \sum_{k=1}^{\infty} \frac{e^{-\int_{0}^{t} \lambda_{s} ds} \left(\int_{0}^{t} \lambda_{s} ds\right)^{k}}{k!} \int_{-\infty}^{\infty} \left(1 - \frac{1}{2} Erfc\left(\frac{y - y_{2}}{\sqrt{2}t} - A\frac{\sqrt{t}}{\sqrt{2}}\right)\right) - \frac{1}{2} e^{2A(y - y_{2})} Erfc\left(\frac{y - y_{2}}{\sqrt{2}t} + A\frac{\sqrt{t}}{\sqrt{2}}\right) dF_{k}(y_{2}),$$

where the  $F_k(y_2)$  is the distribution function of  $\frac{1}{m\sigma}\sum_{n=1}^k \ln(1+mY_n)$  and

$$y = \frac{1}{m\sigma} \ln \frac{(\alpha - 1)F_0 + \alpha C_0}{C_0}.$$

*Proof.* We have

$$\mathbb{P}(T_1^{c,\alpha} \le t) = 1 - \mathbb{P}(T_1^{c,\alpha} > t)$$

Before  $T_1^{c,\alpha}$ , we have

$$V_{t} = C_{0} \exp\left\{\int_{0}^{t} \left((r + m(\mu_{s} - r)) - \frac{m^{2}\sigma_{s}^{2}}{2}\right) ds + \int_{0}^{t} m\sigma_{s} dW_{s} + \sum_{n=1}^{N_{t}} \ln(1 + mY_{n})\right\} + F_{0}e^{rt}.$$

Denote

$$X_t = \int_0^t \left( m(\mu_s - r) - \frac{m^2 \sigma_s^2}{2} \right) ds + \int_0^t m \sigma_s dW_s + \sum_{n=1}^{N_t} \ln(1 + mY_n).$$

When  $\mu_s = \mu$  and  $\sigma_s = \sigma$  is constant, let

$$A = \frac{(\mu - r) - \frac{1}{2}m\sigma^2}{\sigma}$$

and

$$W_s^{(A)} = \left(As + W_s + \frac{1}{m\sigma} \sum_{n=1}^{N_t} \ln(1 + mY_n)\right),$$

then we have

$$\mathbb{P}(T_1^{c,\alpha} > t) = \mathbb{P}\left(\sup_{0 \le s \le t} \frac{V_t}{V_s} < \alpha e^{rt}\right) = \mathbb{P}\left(\sup_{0 \le s \le t} \left(C_0 e^{X_s} + F_0\right) < \alpha V_0\right)$$
$$= \mathbb{P}\left(\sup_{0 \le s \le t} X_s < \ln \frac{(\alpha V_0 - F_0)}{C_0}\right) = \mathbb{P}\left(\sup_{0 \le s \le t} W_s^{(A)} < \frac{1}{m\sigma} \ln \frac{(\alpha - 1)F_0 + \alpha C_0}{C_0}\right).$$

By subsection 6.2.2 we deduce that

$$\mathbb{P}\left(\sup_{s\leq t} W_s^{(A)} \leq y\right) = \sum_{k=1}^{\infty} \frac{e^{-\int_0^t \lambda_s ds} \left(\int_0^t \lambda_s ds\right)^k}{k!} \int_{-\infty}^{\infty} \left(1 - \frac{1}{2} \operatorname{Erfc}\left(\frac{y - y_2}{\sqrt{2}t} - A\frac{\sqrt{t}}{\sqrt{2}}\right)\right) \\ -\frac{1}{2} e^{2A(y - y_2)} \operatorname{Erfc}\left(\frac{y - y_2}{\sqrt{2}t} + A\frac{\sqrt{t}}{\sqrt{2}}\right) dF_k(y_2)$$

where  $F_k(y_2)$  is the distribution function of  $\frac{1}{m\sigma} \sum_{n=1}^k \ln(1+mY_n)$ . Let  $y = \frac{1}{m\sigma} \ln \frac{(\alpha-1)F_0 + \alpha C_0}{C_0}$ , then we have the conclusion.

# 6.4 CPPI with a floor indexed on the exposition variance

#### 6.4.1 The "Ratchet" CPPI

#### Discrete-time case with fixed multiple

For  $\tau \in \{t_0 = 0, ..., t_N = T\}$ , the CPPI strategy is defined as follows. The floor is based on the difference between the two potential values of the exposure.

As usual, the exposure is defined as the minimum between the standard cushion multiplied by the multiple and a given percentage of the portfolio value:

$$e_{t_k} = \inf[m(V_{t_k} - \tilde{F}_{t_k}), \varpi V_{t_k}].$$
 (6.1)

We have

$$e_{t_k} = \inf[m(V_{t_k} - \tilde{F}_{t_k}), \varpi V_{t_k}] \iff F_{t_k} = \max\left[\tilde{F}_{t_k}, \frac{m - \varpi}{m} V_{t_k}\right]$$

This is equivalent to the situation in subsection 4.2.1 with the percentage

$$x = \frac{m - \varpi}{m}.$$

Denote by  $T_1^{c,r}$  the first time at which  $m(V_{t_k} - \tilde{F}_{t_k})$  greater than  $\varpi V_{t_k}$ . Its properties is a special case as in subsection 4.2.1 with the percentage

$$x = \frac{m - \varpi}{m}.$$

#### Continuous-time case with fixed multiple

The floor is based on the difference between the two potential values of the exposure. As usual, the exposure is defined as the minimum between the standard cushion multiplied by the multiple and a given percentage of the portfolio value:

$$e_t = \inf[mC_t, \varpi V_t]. \tag{6.2}$$

At time 0, the exposure  $e_0$  is assumed to be equal to  $mC_0$ . Consider the first time  $T_1^{d,r}$  at which  $mC_t$  becomes higher than  $\varpi V_t$ . That is:

**Proposition 6.11.** Under above assumption, the time  $T_1^{d,r}$  is characterized by the relation:

$$T_1^{d,r} = \inf\{t \le T : mC_t \ge \varpi V_t\}.$$

Then, the floor is defined as follows:

$$F_t^r = F_t \text{ if } t < T_1^r;$$

$$F_t^r = \left(\frac{m - \varpi}{m}\right) V_{T_1^r} e^{r(t - T_1^r)} \text{ if } t \ge T_1^r.$$

$$129$$

We have

$$e_t = \inf[mC_t, \varpi V_t] \iff m(V_t - F_t) = \inf[m(V_t - \tilde{F}_t), \varpi V_t]$$
$$\iff F_t = \max\left[\tilde{F}_t, \frac{m - \varpi}{m} V_t\right]$$

On the other hand, the standard convergence result in the discrete-time case and proposition 4.3 lead to the equivalent situation given in subsection 4.2.2 with the percentage

$$x = \frac{m - \varpi}{m}.$$

Thus, we can calculate the probability distribution of  $T_1^{d,r}$  using the result in subsection 4.2.2.

#### 6.4.2 CPPI with margin

This kind of strategy can be applied in the situation when the initial exposition is too high.

The initial floor is chosen to be higher than the reference floor. The difference, called *the margin*, can be used later if the exposure gets too small.

Denote by  $F_0$  the initial reference level of the floor. The initial value of the stochastic floor  $F_0$  is equal to the reference level plus an initial margin equal to  $M_0$ . Thus we have:

$$F_0 = \tilde{F}_0 + M_0.$$

The exposition e is equal to mC with C = V - F. Assume that  $F_t = F_0 e^{rt}$  until the time  $T_1^{marg}$  at which the exposure e becomes less than or equal to 0. The floor F is then:

$$F_{T_1^{marg}} = (\tilde{F}_0 + \gamma M_0) e^{r T_1^{marg}} \text{ with } 0 < \gamma < 1.$$
(6.3)

That is, at time  $T_1^{marg}$ , the reduction of the floor equals to

$$(1-\gamma)M_0e^{rT_1^{marg}}.$$

Usually, the parameter  $\gamma$  is set to 1/2.

The probability distribution of the time  $T_1^{marg}$  is determined as follows. We consider a "small"  $\varepsilon > 0$  and examine the time  $T_1^{marg}(\varepsilon)$  at which  $V_t$  is equal or less than  $(F_0 + \varepsilon)e^{rt}$ . We have: for any  $t \leq T_1^{marg}(\varepsilon)$ 

$$V_t = C_0 \exp\left\{\int_0^t \left(r + m(\mu_s - r) - \frac{m^2 \sigma_s^2}{2}\right) ds + \int_0^t m \sigma_s dW_s + \sum_{n=1}^{N_t} \ln(1 + mY_n)\right\} + F_0 e^{rt}.$$

Denote

$$X_t = \int_0^t \left( (m(\mu_s - r)) - \frac{m^2 \sigma_s^2}{2} \right) ds + \int_0^t m \sigma_s dW_s + \sum_{n=1}^{N_t} \ln(1 + mY_n)$$

and

$$T_1^{\text{marg}}(\varepsilon) = \inf \left\{ t \le T | \inf_{0 \le s \le t} (X_s) \le \ln \left[ \frac{\varepsilon}{C_0} \right] \right\}.$$

When the  $\mu_s = \mu$  and  $\sigma_s = \sigma$  are constants we have

$$X_{t} = \left(m(\mu - r) - \frac{1}{2}m^{2}\sigma^{2}\right)t + m\sigma W_{t} + \sum_{n=1}^{N_{t}}\ln(1 + mY_{n}).$$

Let

$$A = \frac{(\mu - r) - \frac{1}{2}m\sigma^2}{\sigma}$$

and

$$W_s^{(A)} = \left(As + W_s + \frac{1}{m\sigma} \sum_{n=1}^{N_t} \ln(1 + mY_n)\right),$$

so that

$$T_1^{\text{marg}}(\varepsilon) = \inf \left\{ t \le T \middle| \inf_{0 \le s \le t} \left( W_s^A \right) \le \frac{\ln \left[ \frac{\varepsilon}{C_0} \right]}{m\sigma} \right\}.$$

Denote

$$y = \frac{\ln \left[\frac{\varepsilon}{C_0}\right]}{m\sigma},$$

then

$$\mathbb{P}\left(\inf_{0\leq s\leq t}\left(W_{s}^{A}\right)\leq y\right)=\mathbb{P}\left(-\inf_{0\leq s\leq t}\left(W_{s}^{A}\right)\leq -y\right)=\mathbb{P}\left(\sup_{0\leq s\leq t}\left(-W_{s}^{A}\right)\leq -y\right).$$

Similar to our discussions in subsection 4.2.2, we get

$$\mathbb{P}\left(\inf_{0\leq s\leq t} \left(W_{s}^{A}\right)\leq y\right) = \mathbb{P}\left(\sup_{0\leq s\leq t} \left(-W_{s}^{A}\right)\leq -y\right)$$
$$= \sum_{k=1}^{\infty} \frac{e^{-\int_{0}^{t} \lambda_{s} ds} \left(\int_{0}^{t} \lambda_{s} ds\right)^{k}}{k!} \int_{-\infty}^{\infty} \left(\frac{1}{2} \operatorname{Erfc}\left(\frac{-y-y_{2}}{\sqrt{2}t} + A\frac{\sqrt{t}}{\sqrt{2}}\right)\right)$$
$$+ \frac{1}{2} e^{-2A(-y-y_{2})} \operatorname{Erfc}\left(\frac{-y-y_{2}}{\sqrt{2}t} - A\frac{\sqrt{t}}{\sqrt{2}}\right)\right) dF_{k}'(y_{2}),$$

where  $F'_k(y_2)$  is the distribution function of

$$-\frac{1}{m\sigma}\sum_{n=1}^k \ln(1+mY_n).$$

Therefore, we have:

**Proposition 6.12.** Suppose that  $\mu_s = \mu$  and  $\sigma_s = \sigma$  are constants. Then the cdf of the time  $T_1^{marg}(\varepsilon)$  is given by

$$\mathbb{P}\left(T_1^{marg}(\varepsilon) \le t\right)$$

$$= \sum_{k=1}^{\infty} \frac{e^{-\int_0^t \lambda_s ds} (\int_0^t \lambda_s ds)^k}{k!} \int_{-\infty}^{\infty} \left(1 - \frac{1}{2} Erfc\left(\frac{-y - y_2}{\sqrt{2}t} + A\frac{\sqrt{t}}{\sqrt{2}}\right)\right)$$

$$- \frac{1}{2} e^{-2A(-y - y_2)} Erfc\left(\frac{-y - y_2}{\sqrt{2}t} - A\frac{\sqrt{t}}{\sqrt{2}}\right) \right) dF'_k(y_2)$$

Proof.

$$\mathbb{P}\left(T_1^{\mathrm{marg}}(\varepsilon) \le t\right) = 1 - \mathbb{P}\left(T_1^{\mathrm{marg}}(\varepsilon) > t\right) = 1 - \mathbb{P}\left(\inf_{0 \le s \le t} \left(W_s^A\right) \le y\right).$$

Substitute last term and we get the conclusion.

# Chapter 7

# CPPI in the Fractional Brownian Markets

#### 7.1 Fractional Brownian Markets

Define

$$\phi(s,t) = H(2H-1)|s-t|^{2H-2}, \quad s,t \in \mathbb{R}.$$
(7.1)

and suppose that  $B_H(t)$  is a fractional Brownian motion with Hurst parameter Hin (1/2, 1) defined on the probability space  $(\Omega, \mathfrak{F}, \mu_{\phi})$ . Let  $\mathfrak{F}_t^{(H)}$  be the filtration generated by  $B_H(t)$ .

Reference [22] discusses the fractional Ito Integrals in terms of the Wick product associated with the fractional Brownian motion having Hurst parameter in (1/2, 1). i.e.

$$\int_{a}^{b} f(t,\omega) dB_{H}(t) = \lim_{|\Delta| \to 0} \sum_{k=0}^{n-1} f(t_{k},\omega) \diamond (B_{H}(t_{k+1}) - B_{H}(t_{k})).$$
(7.2)

See also [35] for some finance applications .

For the detail of the wick product and construction of fractional Brownian motion 134
with Hurst parameter H, see references [22] and [35].

**Definition 7.1.** The **fractional Black-Scholes market** has two investment components:

(1)A bank account or a bond, where the price A(t) satisfies:

$$dA(t) = rA(t)dt, \quad A(0) = 1; \quad 0 \le t \le T.$$
 (7.3)

(2) A stock, where the price S(t) satisfies:

$$dS(t) = \mu S(t)dt + \sigma S(t)dB_H(t); \quad S(0) = x > 0,$$
(7.4)

and its solution is

$$S(t) = x \exp\left(\sigma B_H(t) + \mu t - \frac{1}{2}\sigma^2 t^{2H}\right) \quad t \ge 0.$$
(7.5)

**Definition 7.2.** A portfolio or trading strategy  $\theta(t) = \theta(t, \omega) = (u(t), v(t))$  is an  $\mathfrak{F}_t^{(H)}$ -adapted two-dimensional process giving the number of units u(t), v(t) held at time t of the bond and the stock, respectively.

We assume the corresponding value process  $Z(t) = Z^{\theta}(t, \omega)$  is given by

$$Z^{\theta}(t,\omega) = u(t)A(t) + v(t) \diamond S(t).$$
(7.6)

Definition 7.3. The portfolio is called self-financing if

$$dZ^{\theta}(t,\omega) = u(t)dA(t) + v(t) \diamond dS(t)$$
  
$$:= u(t)dA(t) + \mu v(t) \diamond S(t)dt + \sigma v(t) \diamond S(t)dB_{H}(t); \quad t \in [0,T].$$
(7.7)

The Girsanov theorem for the fractional Brownian motion (Theorem 3.18 in [35])

shows that

$$\hat{B}_H(t) := \frac{\mu - r}{\sigma} t + B_H(t) \tag{7.8}$$

is a fractional Brownian motion with respect to the measure  $\hat{\mu}_{\phi}$  defined on  $\mathfrak{F}_T^H$  by

$$d\hat{\mu}_{\phi}(\omega) = \exp\left(-\int_0^T K(s)dB_H(s) - \frac{1}{2}|K|_{\phi}^2\right)d\mu_{\phi}(\omega),\tag{7.9}$$

where K(s) = K(T, s) is defined by the following properties: supp  $K \subset [0, T]$  and

$$\int_0^T K(T,s)\phi(t,s)ds = \frac{\mu - r}{\sigma}, \quad \text{for } 0 \le t \le T.$$
(7.10)

For the self-financing portfolio, from (7.6) and (7.7), we have

$$dZ^{\theta}(t) = rZ^{\theta}(t)dt + \sigma v(t) \diamond S(t) \left[\frac{\mu - r}{\sigma}dt + dB_{H}(t)\right]$$
(7.11)

$$= rZ^{\theta}(t)dt + \sigma v(t) \diamond S(t)d\hat{B}_{H}(t)$$
(7.12)

Let  $\hat{L}^{1,2}_{\phi}(\mathbb{R})$  denote the completion of the set of all  $\mathfrak{F}^{(H)}_t$ -adapted processes  $f(t) = f(t,\omega)$  such that

$$||f||_{\hat{L}^{1,2}_{\phi}(\mathbb{R})} := \mathbb{E}_{\hat{\mu}_{\phi}} \left[ \int_{\mathbb{R}} \int_{\mathbb{R}} f(s) f(t) \phi(s,t) ds dt \right] + \mathbb{E}_{\hat{\mu}_{\phi}} \left[ \left( \int_{\mathbb{R}} D_s^{\phi} f(s) ds \right)^2 \right] < \infty.$$

**Definition 7.4.** A portfolio is called **admissible** if it is self-financing and  $v \diamond S \in \hat{L}^{1,2}_{\phi}(\mathbb{R})$ .

**Definition 7.5.** An admissible portfolio  $\theta$  is called an **arbitrage** for the market in  $t \in [0, T]$  if

$$Z^{\theta}(0) \leq 0, Z^{\theta}(T) \geq 0$$
 a.s. and  
 $\mu_{\phi} \left( \omega : Z^{\theta}(T, \omega) > 0 \right) > 0.$ 

**Definition 7.6.** The market  $(A(t), S(t)); t \in [0, T]$  is called **complete** if for every  $\mathfrak{F}_{T}^{(H)}$ -measurable bounded random variable  $F(\omega)$  there exists  $z \in \mathbb{R}$  and portfolio  $\theta = (u, v)$  such that

$$F(\omega) = Z^{\theta, z}(T, \omega). \tag{7.13}$$

This is the same as the condition that

$$e^{-rT}F(\omega) = z + \int_0^T e^{-rt}\sigma v(t) \diamond S(t)d\hat{B}_H(t).$$
(7.14)

Reference [35] showes that the fractional Black-Scholes market (7.3) and (7.4) has no arbitrage opportunities and it is complete.

#### 7.2 CPPI in the Fractional Black-Scholes market

Recall that  $V_t$  is the portfolio value,  $F_t = rF_t dt$ ,  $F_T = G$  is the floor,  $C_t = V_t - F_t$  is the cushion, m is the multiplier and  $e_t = mC_t$  is the exposure.

**Proposition 7.7.** The portfolio value of CPPI under the fractional Black-Scholes model in continuous trading time is

$$V_t = (V_0 - F_0) \exp\left[(r + m(\mu - r))t - \frac{1}{2}m^2\sigma^2 t^{2H} + m\sigma B_H(t)\right] + F_t.$$
 (7.15)

*Proof.* With the trading strategies denoted by  $\theta(t) = (u(t), v(t))$ , we have the portfolio value  $V_t$ 

$$V_t = u_t A_t + v_t \diamond S_t, \tag{7.16}$$

$$dV_t = u_t dA_{t_1} + v_t \diamond dS_t, \tag{7.17}$$

and

$$v_t \diamond S_t = m(V_t - F_t). \tag{7.18}$$

By (7.11), we have

$$dV_t = rV_t dt + \sigma v_t \diamond S_t \left[ \frac{\mu - r}{\sigma} dt + dB_H(t) \right].$$
(7.19)

Substitute (7.18) into (7.19), we obtain,

$$dV_t = rV_t dt + \sigma m(V_t - F_t) \left[\frac{\mu - r}{\sigma} dt + dB_H(t)\right].$$
(7.20)

Since  $C_t = V_t - F_t$  and  $dF_t = rF_t dt$ , we have

$$d(V_t - F_t) = r(V_t - F_t)dt + \sigma m(V_t - F_t) \left[\frac{\mu - r}{\sigma}dt + dB_H(t)\right],$$
(7.21)

thus,

$$dC_t = rC_t dt + \sigma mC_t \left[\frac{\mu - r}{\sigma} dt + dB_H(t)\right]$$
(7.22)

$$= C_t (r + m(\mu - r)dt + m\sigma dB_H(t)).$$
 (7.23)

Then

$$C_t = C_0 \exp\left[ (r + m(\mu - r))t - \frac{1}{2}m^2\sigma^2 t^{2H} + m\sigma B_H(t) \right].$$
 (7.24)

Therefore, we have (7.15).

By (3.50) in [35], we have

$$\mathbb{E}_{\mu_{\phi}}[C_t] = C_0 \exp\left[(r + m(\mu - r))t\right].$$

Thus we have

**Proposition 7.8.** The expectation of CPPI portfolio value under the fractional Black-Scholes model in continuous time trading is

$$(V_0 - F_0) \exp\left[(r + m(\mu - r))t\right] + F_t.$$
(7.25)

**Proposition 7.9.** The variance of the CPPI portfolio value under the fractional Black-Scholes model in continuous time trading is

$$Var[V_t] = (V_0 - F_0)^2 \exp\left[2(r + m(\mu - r))t\right] \left[\exp\left[m^2 \sigma^2 t^{2H}\right] - 1\right].$$
 (7.26)

Proof.

$$\operatorname{Var}[V_t] = \operatorname{Var}[C_t]$$
  
=  $C_0^2 \exp\left[2(r+m(\mu-r))t\right] \operatorname{Var}\left[\exp\left[-\frac{1}{2}m^2\sigma^2 t^{2H} + m\sigma B_H(t)\right]\right].$ 

For Var  $\left[\exp\left[-\frac{1}{2}m^2\sigma^2t^{2H}+m\sigma B_H(t)\right]\right]$ , we have

$$\operatorname{Var}\left[\exp\left[-\frac{1}{2}m^{2}\sigma^{2}t^{2H}+m\sigma B_{H}(t)\right]\right]$$
$$= \mathbb{E}_{\mu_{\phi}}\left[\exp\left[-m^{2}\sigma^{2}t^{2H}+2m\sigma B_{H}(t)\right]\right]$$
$$-\left(\mathbb{E}_{\mu_{\phi}}\left[\exp\left[-\frac{1}{2}m^{2}\sigma^{2}t^{2H}+m\sigma B_{H}(t)\right]\right]\right)^{2}$$
$$= \exp\left[m^{2}\sigma^{2}t^{2H}\right]-1.$$

For the last step, we have used (3.50) in [35]. Therefore, we get

$$\operatorname{Var}[V_t] = C_0^2 \exp\left[2(r + m(\mu - r))t\right] \left[\exp\left[m^2 \sigma^2 t^{2H}\right] - 1\right].$$

139

### 7.3 CPPI Option

We consider the Vanilla options underlying the CPPI portfolio.

**Proposition 7.10.** The pricing of CPPI portfolio call option under the fractional Black-Scholes model is

$$e^{-rT} \mathbb{E}_{\hat{\mu}_{\phi}}[(V_T - K)^+] = (V_0 - F_0) \Phi\left(\eta + \frac{1}{2}m\sigma T^H\right) - (G - K)e^{-rT} \Phi\left(\eta - \frac{1}{2}m\sigma T^H\right),$$
(7.27)

where

$$\eta = (m\sigma)^{-1}T^{-H}\left(\ln\frac{V_0 - F_0}{G - K}\right) + rT$$

and  $\Phi(t)$  is the normal distribution function.

Proof. Since

$$e^{-rT}\mathbb{E}_{\hat{\mu}_{\phi}}[(V_T - K)^+] = e^{-rT}\mathbb{E}_{\hat{\mu}_{\phi}}[(C_T + G - K)^+]$$

and  $C_t$  has the expression (7.23), when compared with (5.2) in [35], we see that the result is the same as the one given in corollary 5.5 in [35] where we use G - K,  $r + m(\mu - r)$ ,  $V_0 - F_0$ , r and  $m\sigma$  to substitute for c,  $\mu$ , x,  $\rho$  and  $\sigma$  in (5.23) of [35] respectively. Therefore,

$$e^{-rT} \mathbb{E}_{\hat{\mu}_{\phi}}[(V_T - K)^+] = (V_0 - F_0) \Phi\left(\eta + \frac{1}{2}m\sigma T^H\right)$$
$$-(G - K)e^{-rT} \Phi\left(\eta - \frac{1}{2}m\sigma T^H\right),$$

where

$$\eta = (m\sigma)^{-1} T^{-H} \left( \ln \frac{V_0 - F_0}{G - K} \right) + rT$$
140

and  $\Phi(t)$  is the normal distribution function.

### 7.4 PDE Approach

**Theorem 7.11.** For any contingent claim of the form  $g(S_t)$ , there exists a unique self-financed  $g(S_T)$ -hedging CPPI portfolio V; defined as

$$V_t = v(t, S_t) \quad t \in [0, T]$$
 (7.28)

for  $v \in C^{1,2}([0,T] \times \mathbb{R})$  being the unique solution of the partial differential equation (PDE).

$$\frac{\partial u}{\partial t}(t,s) + rs\frac{\partial u}{\partial x}(t,s) + \sigma^2 s^2 H \frac{\partial^2 u}{\partial x^2}(t,s) t^{2H-1} - ru(t,s) = 0;$$
(7.29)

$$u(T,s) = g(s), \quad (t,s) \in [0,T] \times \mathbb{R}), \quad u \in C^{1,2}([0,T] \times \mathbb{R});$$
 (7.30)

In particular the CPPI portfolio's gearing factor is given by:

$$m = \frac{\frac{\partial u}{\partial x}(t, S_t)S_t}{V_t - F_t}, \quad t \in [0, T].$$

$$(7.31)$$

*Proof.* In order to have V is a self-financed  $g(S_T)$ -hedging portfolio, it is enough to ensure that at maturity:

$$V_T = g(S_T), \quad a.s..$$

Choose  $v \in C^{1,2}([0,T] \times \mathbb{R})$  and set  $V_t = v(t, S_t)$   $(t \in [0,T])$ . Now,  $v(T, S_T) = g(S_T)$  P-a.s., so that:

$$v(T,s) = g(s), \quad s \in \mathbb{R}.$$
  
141

Then, by the FBM version of Ito's formula (see [25]),

$$dv(t, S_t) = \left[\frac{\partial v}{\partial t} + \mu S_t \frac{\partial v}{\partial x} + \sigma^2 S_t^2 H \frac{\partial^2 v}{\partial x^2} t^{2H-1}\right] (t, S_t) dt + \sigma S_t \frac{\partial v}{\partial x} \diamond dB_H(s).$$

On the other hand, by (7.20),  $V_t$  satisfies

$$dV_t = rV_t dt + \sigma m(V_t - F_t) \left[ \frac{\mu - r}{\sigma} dt + dB_H(t) \right],$$

A comparison between the above two equations gives

$$m = \frac{\frac{\partial v}{\partial x}(t, S_t)S_t}{V_t - F_t}$$

and

$$\frac{\partial v}{\partial t}(t,s) + \mu s \frac{\partial v}{\partial x}(t,s) + \sigma^2 s^2 H \frac{\partial^2 v}{\partial x^2}(t,s) t^{2H-1} = rv(t,s) + (\mu - r)s \frac{\partial v}{\partial x}(t,s).$$

That is

$$\frac{\partial v}{\partial t}(t,s) + rs\frac{\partial v}{\partial x}(t,s) + \sigma^2 s^2 H \frac{\partial^2 v}{\partial x^2}(t,s) t^{2H-1} - rv(t,s) = 0.$$

Hence given any contingent claim  $\eta = g(V_T)$ , there exists a unique self-financed  $\eta = g(V_T)$ -hedging strategy:

**Theorem 7.12.** For any map  $g : \mathbb{R} \to \mathbb{R}$  sufficiently smooth, there exists a unique  $\eta = g(V_T)$ -hedging self-financed trading strategy  $(U, \beta)$  defined as

$$U_t = u(t, V_t), \quad \beta_t = \frac{\partial u}{\partial x}(t, V_t), \quad t \in [0, T],$$

where  $u \in C^{1,2}([0,T] \times \mathbb{R})$  is the unique solution of the PDE:

$$\frac{\partial u}{\partial t}(t,v) + rv\frac{\partial u}{\partial x}(t,v) + Ht^{2H-1}(m\sigma)^2(v-f)^2\frac{\partial^2 u}{\partial x^2}(t,v) - ru(t,v) = 0 \qquad (7.32)$$

with the final condition u(T, v) = g(v).

*Proof.* Consider  $\{V_t\}_{t\in[0,T]}$  as an asset, and pick a self-financed  $g(V_T)$  hedging strategy  $(U_t, \beta_t)_{t\in[0,T]}$  by setting:

$$dU_t = \beta_t dV_t + (U_t - \beta_t V_t) r dt$$

and

$$U_T = g(V_T) \quad a.s.$$

Since

$$dV_t = rV_t dt + \sigma m(V_t - F_t) \left[ \frac{\mu - r}{\sigma} dt + dB_H(t) \right],$$

the hedging portfolio's equation may be rewritten as:

$$dU_t = \beta_t \left( rV_t dt + \sigma m(V_t - F_t) \left[ \frac{\mu - r}{\sigma} dt + dB_H(t) \right] \right) + (U_t - \beta_t V_t) r dt$$
  
=  $[rU_t + \beta_t (V_t - F_t) m(\mu - r)] dt + \sigma m \beta_t (V_t - F_t) dB_H(t).$ 

Pick  $u \in C^{1,2}([0,T] \times \mathbb{R})$  and set  $U_t = u(t, V_t), \quad t \in [0,T].$ 

For any  $t \in [0, T]$ , the FBM Ito's formula implies that:

$$du(t, V_t) = \left[\frac{\partial u}{\partial t}(t, V_t) + (rV_t + m(\mu - r)(V_t - F_t))\frac{\partial u}{\partial x}(t, V_t) + Ht^{2H-1}(m\sigma)^2(V_t - F_t)^2\frac{\partial^2 u}{\partial x^2}(t, V_t)\right]dt + m\sigma(V_t - F_t)\frac{\partial u}{\partial x}(t, V_t)dB_H(t).$$

$$143$$

A comparison between the above two equations implies in particular

$$\beta_t = \frac{\partial u}{\partial x}(t, V_t)$$

and

$$\begin{aligned} \frac{\partial u}{\partial t}(t,v) + (rv + m(\mu - r)(v - f))\frac{\partial u}{\partial x}(t,v) + Ht^{2H-1}(m\sigma)^2(v - f)^2\frac{\partial^2 u}{\partial x^2}(t,v) \\ &= ru(t,v) + m(v - f)(\mu - r)\frac{\partial u}{\partial x}(t,v). \end{aligned}$$

Thus

$$\frac{\partial u}{\partial t}(t,v) + rv\frac{\partial u}{\partial x}(t,v) + Ht^{2H-1}(m\sigma)^2(v-f)^2\frac{\partial^2 u}{\partial x^2}(t,v) - ru(t,v) = 0$$

with the final condition u(T, v) = g(v).

_	_
-	-

## Chapter 8

# CPPI in Fractional Brownian Markets with Jumps

#### 8.1 Fractional Brownian Markets with Jumps

As before consider:

$$\phi(s,t) = H(2H-1)|s-t|^{2H-2}, \quad s,t \in \mathbb{R}.$$
(8.1)

Let  $B_H(t)$  be a fractional Brownian motion with Hurst parameter H in the interval (1/2, 1), living under the probability space  $(\Omega, \mathfrak{F}, \mu_{\phi})$ . Moreover,  $\mathfrak{F}_t^{(H)}$  denotes the filtration generated by  $B_H(t)$ .

[22] introduces the fractional Ito Integrals in terms of the Wick product. That is,

$$\int_{a}^{b} f(t,\omega) dB_{H}(t) = \lim_{|\Delta| \to 0} \sum_{k=0}^{n-1} f(t_{k},\omega) \diamond (B_{H}(t_{k+1}) - B_{H}(t_{k})).$$
(8.2)

Let the price  $S_t$  of a risky asset (usually stocks or their benchmark) be a right continuous with left limits stochastic process on this probability space which jumps at the random times  $T_1$ ,  $T_2$ ,... and suppose that the relative/proportional change in 145 its value at a jump time is given by  $Y_1$ ,  $Y_2$ ,... respectively. We usually assume the  $\ln(1 + Y_n)$ s are i.i.d. and in our paper, we denote the density function of  $\ln(1 + Y_n)$ s by  $f_Q$ . We assume that, between any two consecutive jump times, the price  $S_t$  follows the fractional Black-Scholes model. The  $T_n$ 's are the jump times of a Poisson process  $N_t$  with intensity  $\lambda_t$  and the  $Y_n$ 's are a sequence of random variables with values in  $(-1, +\infty)$ . The description of the model can be formalized by letting, on the intervals  $t \in [T_n, T_{n+1})$ ,

$$dS_t = S_t(\mu dt + \sigma dB_H(t)). \tag{8.3}$$

Where, at  $t = T_n$ , the jump size is given by  $\Delta S_n = S_{T_n} - S_{T_n} = S_{T_n-}Y_n$ , so that

$$S_{T_n} = S_{T_n}(1+Y_n)$$

and by assumption,  $Y_n > -1$ , leads to positive values of the prices. At the generic time t,  $S_t$  satisfies

$$dS(t) = S(t)(\mu dt + \sigma dB_H(t)) + S(t-)Y_t dN_t$$

$$(8.4)$$

where  $Y_t$  is obtained from  $Y_n$  by a piecewise constant and left continuous time interpolation, i.e.

$$Y_t = Y_n \quad \text{if} \quad T_n < t \le T_{n+1},$$

here we let  $T_0 = 0$ .

We have

$$S_{t} = S_{0} \exp\left(\sigma B_{H}(t) + \mu t - \frac{1}{2}\sigma^{2}t^{2H}\right) \left[\prod_{n=1}^{N_{t}} (1+Y_{n})\right]$$
(8.5)

$$= S_0 \exp\left[\sigma B_H(t) + \mu t - \frac{1}{2}\sigma^2 t^{2H} + \sum_{n=1}^{N_t} \ln(1+Y_n)\right]$$
(8.6)

$$= S_0 \exp\left[\sigma B_H(t) + \mu t - \frac{1}{2}\sigma^2 t^{2H} + \int_0^t \ln(1+Y_s)dN_s\right].$$
 (8.7)

The following definition is redefined by [35] and we adopt them.

**Definition 8.1.** The **fractional Black-Scholes market with jumps** has two possible types of investment:

(1)A bank account or a bond, where the price A(t) satisfies:

$$dA(t) = rA(t)dt, \quad A(0) = 1; \quad 0 \le t \le T.$$
 (8.8)

(2) A stock, where the price S(t) satisfies (8.4).

**Definition 8.2.** A portfolio or trading strategy  $\theta(t) = \theta(t, \omega) = (u(t), v(t))$  is an  $\mathfrak{F}_t^{(H)}$ -adapted two-dimensional process giving the number of units u(t), v(t) held at time t of the bond and the stock, respectively.

We assume that the corresponding value process  $Z(t) = Z^{\theta}(t, \omega)$  is given by

$$Z^{\theta}(t,\omega) = u(t)A(t) + v(t) \diamond S(t).$$
(8.9)

Definition 8.3. The portfolio is called self-financing if

$$dZ^{\theta}(t,\omega) = u(t)dA(t) + v(t) \diamond dS(t)$$
  

$$:= u(t)dA(t) + \mu v(t) \diamond S(t)dt + \sigma v(t) \diamond S(t)dB_{H}(t) \qquad (8.10)$$
  

$$+ v(t) \diamond S(t-)Y_{t}dN_{t}; \quad t \in [0,T].$$

Consider a predictable  $\mathfrak{F}_t^{(H)}$ -process  $\psi_t$ , such that  $\int_0^t \psi_t \lambda_s ds < \infty$ . Choose  $\theta$  and

 $\psi_t$  such that

$$\mu + \sigma\theta + Y_t \psi_t \lambda_t = r \tag{8.11}$$

and

$$\psi_t \ge 0.$$

We see that

$$\theta = \sigma^{-1}(r - \mu - Y_t \psi_t \lambda_t) \tag{8.12}$$

where the choice of  $\psi_t$  is arbitrary. Define

$$L_t = \exp\left\{\int_0^t [(1-\psi_s)\lambda_s]ds + \int_0^t \ln\psi_s dN_s - \int_0^t K(s)dB_H(s) - \frac{1}{2}|K|_{\phi}^2\right\}$$
(8.13)

for  $t \in [0,T]$  where K(s) = K(T,s) is defined by the following properties: supp  $K \subset [0,T]$  and

$$\int_0^T K(T,s)\phi(t,s)ds = -\theta, \quad \text{for } 0 \le t \le T.$$
(8.14)

and the Radon-Nikodym derivative is

$$d\hat{\mu}_{\phi}(\omega) = L_T d\mu_{\phi}(\omega). \tag{8.15}$$

Define

$$\hat{B}_H(t) := -\theta t + B_H(t).$$
 (8.16)

Then we have

#### Theorem 8.4. (Girsanov Formula )

(a.) $\hat{B}_{H}(t)$  defined by (8.16) is a fractional Brownian motion that has the hurst parameter  $H \in (1/2, 1)$  with respect to the measure  $\hat{\mu}_{\phi}$ . 148

### (b.) $N_t$ is a Poisson process with intensity $\lambda_t \psi_t$ with respect to the measure $\hat{\mu}_{\phi}$ .

*Proof.* (a.) For any  $f \in \mathcal{S}(\mathbb{R})$ ,  $\operatorname{supp} f \subset [0, T]$  we have

$$\begin{split} \mathbb{E}_{\hat{\mu}_{\phi}} \exp\left(\int_{0}^{T} f(t)d(-\theta t + B_{H}(t))\right) \\ &= \mathbb{E}_{\mu_{\phi}} \left[\exp\left(\int_{0}^{T} -f(t)\theta dt + \int_{0}^{T} f(t)dB_{H}(t)\right) \\ &\times \exp\left(\int_{0}^{T} [(1-\psi_{s})\lambda_{s}]ds + \int_{0}^{T} \ln\psi_{s}dN_{s} - \int_{0}^{T} K(s)dB_{H}(s) - \frac{1}{2}|K|_{\phi}^{2}\right)\right] \\ &= \mathbb{E}_{\mu_{\phi}} \exp\left(\int_{0}^{T} [(1-\psi_{s})\lambda_{s}]ds + \int_{0}^{T} \ln\psi_{s}dN_{s}\right) \exp\left(\int_{0}^{T} -f(t)\theta dt\right) \\ &\times \mathbb{E}_{\hat{\mu}_{\phi}} \exp\left(\int_{0}^{T} f(t)d(-\theta t + B_{H}(t))\right) \exp\left(-\frac{1}{2}|K|_{\phi}^{2}\right). \end{split}$$

For  $\mathbb{E}_{\mu_{\phi}} \exp\left(\int_{0}^{T} [(1-\psi_{s})\lambda_{s}] ds + \int_{0}^{T} \ln \psi_{s} dN_{s}\right)$ , and we have

$$\begin{split} & \mathbb{E}_{\mu_{\phi}} \exp\left(\int_{0}^{T} [(1-\psi_{s})\lambda_{s}]ds + \int_{0}^{T} \ln\psi_{s}dN_{s}\right) \\ &= \mathbb{E}_{\mu_{\phi}} \exp\left(\int_{0}^{T} \ln\psi_{s}dN_{s}\right) \mathbb{E}_{\mu_{\phi}} \exp\left(\int_{0}^{T} [(1-\psi_{s})\lambda_{s}]ds\right) \\ &= \mathbb{E}_{\mu_{\phi}} \prod_{n=1}^{N_{T}} \psi_{n} \mathbb{E}_{\mu_{\phi}} \exp\left(\int_{0}^{T} [(1-\psi_{s})\lambda_{s}]ds\right) \\ &= \sum_{k=0}^{\infty} \mathbb{E}_{\mu_{\phi}} \psi_{n}^{k} \mathbb{P}(N_{T}=k) \mathbb{E}_{\mu_{\phi}} \exp\left(\int_{0}^{T} [(1-\psi_{s})\lambda_{s}]ds\right) \\ &= \exp\left(\int_{0}^{T} -\lambda_{s}ds\right) \sum_{k=0}^{\infty} \mathbb{E}_{\mu_{\phi}} \frac{\left(\int_{0}^{T} -\lambda_{s}\psi_{s}ds\right)^{k}}{k!} \mathbb{E}_{\mu_{\phi}} \exp\left(\int_{0}^{T} [(1-\psi_{s})\lambda_{s}]ds\right) \\ &= \exp\left(\int_{0}^{T} -\lambda_{s}ds\right) \mathbb{E}_{\mu_{\phi}} \exp\left(\int_{0}^{T} \psi_{s}\lambda_{s}ds\right) \mathbb{E}_{\mu_{\phi}} \exp\left(\int_{0}^{T} [(1-\psi_{s})\lambda_{s}]ds\right) \\ &= 1. \end{split}$$

On the other hand, since  $\operatorname{supp} f \subset [0, T]$  and  $\operatorname{supp} K \subset [0, T]$ , we have

$$\int_0^T -f(t)\theta dt = \int_0^T f(t) \int_0^t \frac{K(t,s)\phi(t,s)dsdt}{149} = \langle K, f \rangle_{\phi}^2.$$

Moreover, we have

$$\mathbb{E}_{\hat{\mu}_{\phi}} \exp\left(\int_{0}^{T} f(t) dB_{H}(t) - \int_{0}^{T} K(s) dB_{H}(s)\right) = \exp\left(\frac{1}{2}|f - K|_{\phi}\right).$$

Thus, we have

$$\mathbb{E}_{\hat{\mu}_{\phi}} \exp\left(\int_{0}^{T} f(t)d(-\theta t + B_{H}(t))\right) = \exp\left(\frac{1}{2}|f - K|_{\phi}^{2} - \frac{1}{2}|K|_{\phi}^{2} + \langle K, f \rangle_{\phi}\right)$$
  
=  $\exp\left(\frac{1}{2}(|f|_{\phi}^{2} + |K|_{\phi}^{2} - 2\langle K, f \rangle_{\phi}) - \frac{1}{2}|K|_{\phi}^{2} + \langle K, f \rangle_{\phi}\right)$   
=  $\exp\left(\frac{1}{2}|f|_{\phi}^{2}\right) = \mathbb{E}_{\mu_{\phi}} \exp\left(\int_{0}^{T} f(t)dB_{H}(t)\right).$ 

Thus, we have proved that  $\hat{B}_H(t)$  defined by (8.16) is a fractional Brownian motion with Hurst parameter  $H \in (1/2, 1)$  with respect to the measure  $\hat{\mu}_{\phi}$ .

(b.) Using the result of  $\mathbb{E}_{\mu_{\phi}} \exp\left(\int_{0}^{T} [(1-\psi_{s})\lambda_{s}]ds + \int_{0}^{T} \ln\psi_{s}dN_{s}\right)$  in part (a), for any nonnegative integer k, we have

$$\begin{aligned} \hat{\mu}_{\phi}(N_{T}=k) &= \mathbb{E}_{\hat{\mu}_{\phi}} \mathbf{1}_{N_{T}=k} = \mathbb{E}_{\mu_{\phi}} \mathbf{1}_{N_{T}=k} L_{T} \\ &= \mathbb{E}_{\mu_{\phi}} \mathbf{1}_{N_{T}=k} \exp\left(\int_{0}^{t} [(1-\psi_{s})\lambda_{s}]ds + \int_{0}^{t} \ln\psi_{s}dN_{s}\right) \\ &\times \mathbb{E}_{\mu_{\phi}} \exp\left(-\int_{0}^{t} K(s)dB_{H}(s) - \frac{1}{2}|K|_{\phi}^{2}\right) \\ &= \mathbb{E}_{\mu_{\phi}} \mathbf{1}_{N_{T}=k} \exp\left(\int_{0}^{t} [(1-\psi_{s})\lambda_{s}]ds + \int_{0}^{t} \ln\psi_{s}dN_{s}\right) \\ &= \mathbf{1}_{N_{T}=k} \exp\left(\int_{0}^{T} -\lambda_{s}ds\right) \sum_{i=0}^{\infty} \mathbb{E}_{\mu_{\phi}} \frac{\left(\int_{0}^{T} -\lambda_{s}\psi_{s}ds\right)^{i}}{i!} \exp\left(\int_{0}^{t} [(1-\psi_{s})\lambda_{s}]ds\right) \\ &= \exp\left(\int_{0}^{T} -\lambda_{s}ds\right) \mathbb{E}_{\mu_{\phi}} \frac{\left(\int_{0}^{T} -\lambda_{s}\psi_{s}ds\right)^{k}}{k!} \exp\left(\int_{0}^{t} [(1-\psi_{s})\lambda_{s}]ds\right) \\ &= \exp\left(\int_{0}^{T} -\lambda_{s}\psi_{s}ds\right) \frac{\left(\int_{0}^{T} -\lambda_{s}\psi_{s}ds\right)^{k}}{k!}. \end{aligned}$$

Thus, we have proved that  $N_t$  is a Poisson process with intensity  $\lambda_t \psi_t$  with respect 150

to the measure  $\hat{\mu}_{\phi}$ .

Assume that  $\theta = (u, v)$  is self-financing. Then by (8.9), we have

$$u(t) = \frac{Z^{\theta}(t) - v(t) \diamond S(t)}{A(t)}$$
(8.17)

which, substituted into (8.10) gives

$$dZ^{\theta}(t) = \frac{Z^{\theta}(t) - v(t) \diamond S(t)}{A(t)} dA(t) + \mu v(t) \diamond S(t) dt + \sigma v(t) \diamond S(t) dB_{H}(t)$$
  
+  $v(t)S(t-)Y_{t}dN_{t}$   
=  $rZ^{\theta}(t) - rv(t) \diamond S(t) dt + \mu v(t) \diamond S(t) dt + \sigma v(t) \diamond S(t) dB_{H}(t)$  (8.18)  
+  $v(t) \diamond S(t-)Y_{t}dN_{t}$   
=  $rZ^{\theta}(t) + \sigma v(t) \diamond S(t) [dB_{H}(t) - \theta dt] + v(t) \diamond S(t)Y_{t}(dN_{t} - \psi_{t}\lambda_{t}dt).$ 

Let  $\hat{L}^{1,2}_{\phi}(\mathbb{R})$  denote the completion of the set of all  $\mathfrak{F}^{(H)}_t$ -adapted processes  $f(t) = f(t,\omega)$  such that

$$||f||_{\hat{L}^{1,2}_{\phi}(\mathbb{R})} := \mathbb{E}_{\hat{\mu}_{\phi}} \left[ \int_{\mathbb{R}} \int_{\mathbb{R}} f(s) f(t) \phi(s,t) ds dt \right] + \mathbb{E}_{\hat{\mu}_{\phi}} \left[ \left( \int_{\mathbb{R}} D_s^{\phi} f(s) ds \right)^2 \right] < \infty.$$

**Definition 8.5.** A portfolio is called **admissible** if it is self-financing and  $v \diamond S \in \hat{L}^{1,2}_{\phi}(\mathbb{R})$ .

**Definition 8.6.** An admissible portfolio  $\theta$  is called an **arbitrage** for the market in  $t \in [0, T]$  if

$$Z^{\theta}(0) \le 0, Z^{\theta}(T) \ge 0$$
 a.s. and  
 $\mu_{\phi}\left(\omega : Z^{\theta}(T, \omega) > 0\right) > 0.$ 

From (8.18), we see that

$$\mathbb{E}_{\hat{\mu}_{\phi}}\left[e^{-rT}Z^{\theta}(T)\right] = Z^{\theta}(0), \qquad (8.19)$$

thus, no arbitrage exists.

**Definition 8.7.** The market  $(A(t), S(t)); t \in [0, T]$  is called **complete** if for every  $\mathfrak{F}_T^{(H)}$ -measurable bounded random variable  $F(\omega)$  there exists  $z \in \mathbb{R}$  and portfolio  $\theta = (u, v)$  such that

$$F(\omega) = Z^{\theta, z}(T, \omega). \tag{8.20}$$

**Proposition 8.8.** The fractional Black-Scholes market with jumps is not complete.

*Proof.*  $\hat{\mu}_{\phi}$  is not unique since we could choose different  $\psi_s$  in (8.13). Thus, it is without loss of generality to assume  $\hat{\mu}_{\phi,1}$  and  $\hat{\mu}_{\phi,2}$  as two distinguished measures on probability space  $(\Omega, \mathfrak{F})$ .

If the fractional Black-Scholes market with jumps is complete, then for every  $\mathfrak{F}_T^{(H)}$ measurable bounded random variable  $F(\omega)$  there exist  $z \in \mathbb{R}$  and portfolio  $\theta = (u, v)$ such that

$$F(\omega) = Z^{\theta, z}(T, \omega).$$

By (8.18), we see

$$\mathbb{E}_{\hat{\mu}_{\phi,1}} e^{-rT} F(\omega) = \mathbb{E}_{\hat{\mu}_{\phi,2}} e^{-rT} F(\omega).$$
(8.21)

This contradicts our assumption that the  $\hat{\mu}_{\phi,1}$  and  $\hat{\mu}_{\phi,2}$  are distinct measures on the probability space  $(\Omega, \mathfrak{F})$ . Therefore, The fractional Black-Scholes market with jumps is not complete.

#### 8.1.1 Esscher transform

From our previous section we know that the Radon-Nikodym measure transform (8.15) is not unique. The Esscher Transform technique in [30] provides us a unique risk-neutral transform. Here we apply the Esscher transform on our fractional Brownian Markets with jumps model.

Denote

$$X(t) = \sigma B_H(t) + \mu t - \frac{1}{2}\sigma^2 t^{2H} + \sum_{n=1}^{N_t} \ln(1+Y_n)$$

and with density f(x, t). Then the stock price can be expressed as  $S_t = S_0 \exp[X(t)]$ . By the Esscher transform, the density function of  $X_t$  is: (refer [30])

$$f(x,t;h) = \frac{e^{hx}f(x,t)}{\int_{-\infty}^{\infty} e^{hy}f(y,t)dy} = \frac{e^{hx}f(x,t)}{M(h,t)},$$
(8.22)

where  $M(h,t) := \int_{-\infty}^{\infty} e^{hy} f(y,t) dy$  is the generating function. Denote by M(z,t;h)the moment generating function of X(t). From reference [30] we have

$$M(z,t;h) = \frac{M(z+h,t)}{M(h,t)},$$
(8.23)

and

$$M(z,t;h) = [M(z,1;h)]^t.$$
(8.24)

As in [30], we define the risk-neutral Esscher transform as follows:

**Definition 8.9.** The **risk-neutral Esscher transform** is the Esscher transform with the parameter  $h = h^*$  and denote by  $\mu_{\phi}^*$  the correspondent probability measure, such that

$$S(0) = \mathbb{E}_{\mu_{\phi}^{*}}[e^{-rt}S(t)]$$
(8.25)

[30] deduces that

$$e^r = M(1, 1; h^*).$$
 (8.26)  
153

On the other hand, we have

$$M(z,t) = \mathbb{E}_{\mu_{\phi}} \left[ e^{zX(t)} \right]$$
  
=  $\mathbb{E}_{\mu_{\phi}} \left[ e^{z(\sigma B_{H}(t) + \mu t - \frac{1}{2}\sigma^{2}t^{2H} + \sum_{n=1}^{N_{t}}\ln(1+Y_{n}))} \right]$   
=  $e^{z+\mu t} \mathbb{E}_{\mu_{\phi}} \left[ e^{(\sum_{n=1}^{N_{t}}\ln(1+Y_{n}))} \right]$   
=  $e^{z+\mu t} \mathbb{E}_{\mu_{\phi}} \prod_{n=1}^{N_{t}} (1+Y_{n})$   
=  $e^{z+\mu t} \sum_{k=1}^{\infty} \frac{e^{-\int_{0}^{t}\lambda_{s}ds}(\int_{0}^{t}\lambda_{s}ds)^{k}}{k!} \mathbb{E}_{\mu_{\phi}} \left[ \prod_{n=1}^{k} (1+Y_{n}) \right].$ 

# 8.2 CPPI in fractional Black-Scholes market with jumps

Recall that  $V_t$  represents the portfolio value,  $F_t = rF_t dt$ ,  $F_T = G$  is the floor,  $C_t = V_t - F_t$  is the cushion, m is the multiplier and  $e_t = mC_t$  is the exposure.

**Proposition 8.10.** The portfolio value of CPPI under the fractional Black-Scholes model with jumps in continuous time trading is

$$V_{t} = (V_{0} - F_{0}) \exp\left[(m\mu - r(m-1))t - \frac{1}{2}m^{2}\sigma^{2}t^{2H} + m\sigma B_{H}(t)\right] \times \left[\prod_{n=1}^{N_{t}} (1 + mY_{n})\right] + F_{t}.$$
(8.27)

*Proof.* With  $\theta(t) = (u(t), v(t))$  as the trading strategy, we have:

$$V_t = u_t A_t + v_t \diamond S_t, \tag{8.28}$$

$$dV_t = u_t dA_t + v_t \diamond dS_t, \tag{8.29}$$

and

$$v_t \diamond S_t = m(V_t - F_t). \tag{8.30}$$

By (8.18), we have

$$dV_t = rV_t dt - rv(t) \diamond S(t) dt + \mu v(t) \diamond S(t) dt + \sigma v(t) \diamond S(t) dB_H(t) + v(t) \diamond S(t-)Y_t dN_t.$$
(8.31)

Substitute (8.30) into (8.31), we obtain,

$$dV_t = rV_t dt - rm(V_t - F_t)dt + \mu m(V_t - F_t)dt + \sigma m(V_t - F_t)dB_H(t) + m(V_{t-} - F_t)Y_t dN_t.$$
(8.32)

Since  $C_t = V_t - F_t$  and  $dF_t = rF_t dt$ , we have

$$d(V_t - F_t) = -r(m-1)(V_t - F_t)dt + \mu m(V_t - F_t)dt + \sigma m(V_t - F_t)dB_H(t) + m(V_{t-} - F_t)Y_tdN_t.$$
(8.33)

Thus,

$$dC_t = -r(m-1)C_t dt + \mu m C_t dt + \sigma m C_t dB_H(t) + m C_{t-} Y_t dN_t,$$
(8.34)

then

$$C_{t} = C_{0} \exp\left[(m\mu - r(m-1))t - \frac{1}{2}m^{2}\sigma^{2}t^{2H} + m\sigma B_{H}(t)\right] \times \left[\prod_{n=1}^{N_{t}} (1 + mY_{n})\right].$$
(8.35)

Therefore, we have (8.27).

155

# **Proposition 8.11.** The expected CPPI portfolio value at time t under the fractional Black-Scholes model with jumps is

$$\mathbb{E}_{\mu_{\phi}}[V_t] = C_0 \exp\{(r+m(\mu-r))t\} \sum_{k=1}^{\infty} \frac{e^{-\int_0^t \lambda_s ds} (\int_0^t \lambda_s ds)^k}{k!} \times \mathbb{E}_{\mu_{\phi}}\left[\prod_{n=1}^k (1+mY_n)\right] + F_t$$

*Proof.* Since

$$\begin{split} \mu_{\phi} \left[ \prod_{n=1}^{N_{t}} (1+mY_{n}) \leq x \right] &= \mu_{\phi} \left[ \bigcup_{k=1}^{\infty} \left[ \prod_{n=1}^{N_{t}} (1+mY_{n}) \leq x, N_{t} = k \right] \right] \\ &= \sum_{k=1}^{\infty} \mu_{\phi} \left[ \prod_{n=1}^{N_{t}} (1+mY_{n}) \leq x | N_{t} = k \right] \mu_{\phi} [N_{t} = k] \\ &= \sum_{k=1}^{\infty} \mu_{\phi} \left[ \prod_{n=1}^{k} (1+mY_{n}) \leq x, N_{t} = k \right] \mu_{\phi} [N_{t} = k] \\ &= \sum_{k=1}^{\infty} \mu_{\phi} \frac{\left[ \prod_{n=1}^{k} (1+mY_{n}) \leq x \right] \mu_{\phi} [N_{t} = k]}{\mu_{\phi} [N_{t} = k]} \mu_{\phi} [N_{t} = k] \\ &= \sum_{k=1}^{\infty} \mu_{\phi} \left[ \prod_{n=1}^{k} (1+mY_{n}) \leq x \right] \mu_{\phi} [N_{t} = k] \\ &= \sum_{k=1}^{\infty} \mu_{\phi} \left[ \prod_{n=1}^{k} (1+mY_{n}) \leq x \right] \mu_{\phi} [N_{t} = k] \\ &= \sum_{k=1}^{\infty} \mu_{\phi} \left[ \prod_{n=1}^{k} (1+mY_{n}) \leq x \right] \mu_{\phi} [N_{t} = k] \\ &= \sum_{k=1}^{\infty} \frac{e^{-\int_{0}^{t} \lambda_{s} ds} (\int_{0}^{t} \lambda_{s} ds)^{k}}{k!} \mu_{\phi} \left[ \prod_{n=1}^{k} (1+mY_{n}) \leq x \right], \end{split}$$

we get,

$$\mathbb{E}_{\mu_{\phi}}\left[\prod_{n=1}^{N_t} (1+mY_n)\right] = \sum_{k=1}^{\infty} \frac{e^{-\int_0^t \lambda_s ds} (\int_0^t \lambda_s ds)^k}{k!} \mathbb{E}_{\mu_{\phi}}\left[\prod_{n=1}^k (1+mY_n)\right].$$

From (3.50) in [35], we obtain

$$\mathbb{E}_{\mu_{\phi}}[V_t] = C_0 \mathbb{E}_{\mu_{\phi}} \left[ \exp\left\{ (m\mu - r(m-1))t - \frac{1}{2}m^2\sigma^2 t^{2H} + m\sigma B_H(t) \right\} \right]$$
156

$$\times \mathbb{E}_{\mu_{\phi}} \left[ \prod_{n=1}^{N_{t}} (1+mY_{n}) \right] + F_{t}$$

$$= C_{0} \exp\left\{ (r+m(\mu-r))t \right\} \sum_{k=1}^{\infty} \frac{e^{-\int_{0}^{t} \lambda_{s} ds} (\int_{0}^{t} \lambda_{s} ds)^{k}}{k!}$$

$$\times \mathbb{E}_{\mu_{\phi}} \left[ \prod_{n=1}^{k} (1+mY_{n}) \right] + F_{t}.$$

**Proposition 8.12.** The variance of the CPPI portfolio value at time t under the fractional Balck-Scholes model with jumps is

$$C_{0}^{2} \exp\left\{2((r+m(\mu-r)))t + m^{2}\sigma^{2}t^{2H}\right\} \times \sum_{k=1}^{\infty} \mathbb{E}_{\mu\phi} \left[\prod_{n=1}^{k} (1+mY_{n})\right]^{2} \frac{e^{-\int_{0}^{t} \lambda_{s} ds} (\int_{0}^{t} \lambda_{s} ds)^{k}}{k!} - C_{0}^{2} \left[\exp\{(r+m(\mu-r))t\sum_{k=1}^{\infty} \frac{e^{-\int_{0}^{t} \lambda_{s} ds} (\int_{0}^{t} \lambda_{s} ds)^{k}}{k!} \mathbb{E}_{\mu\phi} \left[\prod_{n=1}^{k} (1+mY_{n})\right]\right]^{2}.$$

*Proof.* Similar to the proof of the above proposition, we have

$$\mathbb{E}_{\mu_{\phi}}\left[\left[\prod_{n=1}^{N_{t}}(1+mY_{n})\right]^{2}\right] = \frac{e^{-\int_{0}^{t}\lambda_{s}ds}(\int_{0}^{t}\lambda_{s}ds)^{k}}{k!}\mathbb{E}_{\mu_{\phi}}\left[\left[\prod_{n=1}^{k}(1+mY_{n})\right]^{2}\right].$$

Thus,

$$\begin{aligned} \operatorname{Var}_{\mu_{\phi}}[V_{t}] &= \operatorname{Var}_{\mu_{\phi}}[C_{t}] \\ &= C_{0}^{2}\operatorname{Var}_{\mu_{\phi}}\left[\exp\left\{(r+m(\mu-r))t - \frac{1}{2}m^{2}\sigma^{2}t^{2H} + m\sigma B_{H}(t)\right\}\left[\prod_{n=1}^{N_{t}}(1+mY_{n})\right]\right] \\ &= C_{0}^{2}\mathbb{E}_{\mu_{\phi}}\left[\exp\left\{(r+m(\mu-r))t - \frac{1}{2}m^{2}\sigma^{2}t^{2H} + m\sigma B_{H}(t)\right\}\left[\prod_{n=1}^{N_{t}}(1+mY_{n})\right]\right]^{2} \\ &- C_{0}^{2}\left(\mathbb{E}_{\mu_{\phi}}\left[\exp\left\{(r+m(\mu-r))t - \frac{1}{2}m^{2}\sigma^{2}t^{2H} + m\sigma B_{H}(t)\right\}\left[\prod_{n=1}^{N_{t}}(1+mY_{n})\right]\right]\right)^{2} \end{aligned}$$

$$= C_{0}^{2}\mathbb{E}_{\mu\phi} \left[ \exp\left\{ (r+m(\mu-r))t - \frac{1}{2}m^{2}\sigma^{2}t^{2H} + m\sigma B_{H}(t) \right\} \left[ \prod_{n=1}^{N_{t}} (1+mY_{n}) \right] \right]^{2} \\ -C_{0}^{2} \left[ \exp\{ (r+m(\mu-r))t \} \sum_{k=1}^{\infty} \frac{e^{-\int_{0}^{t}\lambda_{s}ds}(\int_{0}^{t}\lambda_{s}ds)^{k}}{k!} \mathbb{E}_{\mu\phi} \left[ \prod_{n=1}^{k} (1+mY_{n}) \right] \right]^{2} \\ = C_{0}^{2}\mathbb{E}_{\mu\phi} \left[ \exp\left\{ 2(r+m(\mu-r))t - m^{2}\sigma^{2}t^{2H} + 2m\sigma B_{H}(t) \right\} \right] \mathbb{E}_{\mu\phi} \left[ \prod_{n=1}^{N_{t}} (1+mY_{n}) \right]^{2} \\ -C_{0}^{2} \left[ \exp\{ (r+m(\mu-r))t \} \sum_{k=1}^{\infty} \frac{e^{-\int_{0}^{t}\lambda_{s}ds}(\int_{0}^{t}\lambda_{s}ds)^{k}}{k!} \mathbb{E}_{\mu\phi} \left[ \prod_{n=1}^{k} (1+mY_{n}) \right] \right]^{2} \\ = C_{0}^{2} \exp\left\{ 2(r+m(\mu-r))t + m^{2}\sigma^{2}t^{2H} \right\} \\ \times \sum_{k=1}^{\infty} \mathbb{E}_{\mu\phi} \left[ \prod_{n=1}^{k} (1+mY_{n}) \right]^{2} \frac{e^{-\int_{0}^{t}\lambda_{s}ds}(\int_{0}^{t}\lambda_{s}ds)^{k}}{k!} \\ -C_{0}^{2} \left[ \exp\{ (r+m(\mu-r))t \right\} \sum_{k=1}^{\infty} \frac{e^{-\int_{0}^{t}\lambda_{s}ds}(\int_{0}^{t}\lambda_{s}ds)^{k}}{k!} \mathbb{E}_{\mu\phi} \left[ \prod_{n=1}^{k} (1+mY_{n}) \right] \right]^{2} .$$

The following lemma gives the density function of  $1 + mY_i$ .

**Lemma 8.13.** Let the density function of  $\ln(1 + Y_n)$  be  $f_Q(y)$ . Then the density function  $f'_Q$  of the random variable  $1 + mY_i$  is

$$f'_Q(z) = f_Q\left(\ln\left(1 + \frac{z-1}{m}\right)\right)\frac{1}{m+z-1}.$$

Proof. Since

$$\mu_{\phi}(1+mY_{i} \le z) = \mu_{\phi}\left(\ln(1+Y_{i}) \le \ln\left(1+\frac{z-1}{m}\right)\right) = \int_{-\infty}^{\ln\left(1+\frac{z-1}{m}\right)} f_{Q}(y)dy,$$

the density  $f'_Q$  of the random variable  $1 + mY_i$  is

$$f'_Q(z) = \frac{d(\mu_\phi(1+mY_i \le z))}{dz} = f_Q\left(\ln\left(1+\frac{z-1}{m}\right)\right)\frac{1}{m+z-1}.$$

. 6		

Now we can calculate

$$\mathbb{E}_{\mu_{\phi}}\left[\prod_{n=1}^{k}(1+mY_{n})\right] = \mathbb{E}_{\mu_{\phi}}\left[\exp\left\{\sum_{n=1}^{k}\ln(1+mY_{n})\right\}\right]$$
$$= \int_{\mathbb{R}}\exp\left\{\underbrace{f'_{Q}*f'_{Q}*\ldots*f'_{Q}(x)}_{\text{Convolved }k \text{ times}}\right\}dx$$

and

$$\mathbb{E}_{\mu_{\phi}}\left[\prod_{n=1}^{k} (1+mY_n)^2\right] = \mathbb{E}_{\mu_{\phi}}\left[\exp\left\{\sum_{n=1}^{k} 2\ln(1+mY_n)\right\}\right]$$
$$= \int_{\mathbb{R}} \exp\left\{2\underbrace{f'_Q * f'_Q * \dots * f'_Q(x)}_{\text{Convolved }k \text{ times}}\right\} dx.$$

## Bibliography

- H. Ben Ameur & J.L. Prigent(2006) Portfolio Insurance: determination of a dynamic CPPI multiple as function of state variables, Working paper.
- [2] H. Ben Ameur & J.L. Prigent(2009)CPPI Method with Conditional Floor. The Discrete Time Case, Working paper.
- [3] H. Ben Ameur(2010), Chapter 9 GARCH Models with CPPI Application, in Professor William Barnett (ed.) Nonlinear Modeling of Economic and Financial Time-Series (International Symposia in Economic Theory and Econometrics, Volume 20), Emerald Group Publishing Limited, 187-205.
- [4] T. Arai(2005), An extension of mean-variance hedging to the discontinuous case, Finance Stochastics 9, 129-139(2005).
- [5] S. Balder & M. Brandl & A. Mahayni(2009), Effectiveness of CPPI Strategies under Discrete-Time Trading. Journal of Economic Dynamics and Control 33, 204-220.
- [6] P. Bertrand & J. Prigent(2002), Portfolio Insurance Strategies: OBPI versus CPPI, Working paper.
- [7] F. Biagini, Y. Hu, B. Ksendal & T. Zhang(2008), Stochstic Calculus for Fractional Brownian Motion and Application, Springer, 2008.

- [8] D. Brigo & F. Mercurio(2006), Interest Rate Models Theory and Practice: With Smile, Inflation and Credit, 2nd edition, Springer, 2006.
- T. Bjork & H. Hult(2005), A note on Wick products and fractional Black-Scholes model, Finance and Stochastics, 9, 197-209(2005).
- [10] F. Black & A. Perold(1992), Theory of Constant Proportion Portfolio Insurance. The Journal of Economic Dynamics and Control 16, 403-426.
- [11] A.N. Borodin & P. Salminen(2002): Handbook of Brownian Motion (Facts and Formulae), 2nd ed., Birkhauser Basel, 2002.
- [12] J. Bouchaud & M. Potters(2003), Theory of financial risks: from statistical physics to risk management, 2nd ed., Cambridge University Press, 2003.
- [13], P. Bremaud(1981), Point Processes and Queues: Martingale Dynamics, Springer Verlag, New York, 1981.
- [14] P. Carr & D. Madan(1998), option valuation using the fast fourier transform,J. Computational Finance, 2(1998), 61-73.
- [15] C. Chiarella & A. Ziogas (2004), McKean's method applied to american call option on jump-diffusion processes, Quantitative Finance Research Centre, University of Technology Sydney, Research Paper, No. 117.
- [16] A. Cipollini(2008), Capital Protection modeling the CPPI portfolio, working paper.
- [17] R. Cont & P. Tankov(2009), Constant Proportion Portfolio Insurance in the Presence of Jumps in Asset Prices, Math finance, Vol. 19, No.3, 379-401.
- [18] R. Cont & P. Tankov(2004), Financial Modelling With Jump Processes, Chapman & Hall/CRC financial mathematics series, 2004.

- [19] R. Cont & E. Voltchkova(2005), Integro-differential equations for option prices in exponential Levy models, Finance and Stochastic. 9, 299-325(2005).
- [20] R. Cont,& P. Tankov, & E. Voltchkova(2007), Hedging with options in models with jumps, Stochastic Analysis and Applications: Proc. Abel Sympos., Springer, pp. 197-218, 2007.
- [21] R. Craine & L. Lochstoer & K. Syrtveit(2000), Estimation of a Stochastic-Volatility Jump-Diffusion Model, Working paper.
- [22] T. Duncan, Y. Hu & B. Pasik-Duncan(2000), Stochastic Calculus for Fractional Brownian Motion 1.Theory, SIAM J. Control Optim. Vol. 38, NO.2, pp. 582-612.
- [23] R. Elliott & J. Hoek(2003), A general fractional white noise theory and application to finance, Mathematical finance, Vol. 13 (April 2003), 301-330.
- [24] R. Elliott, Ito formulas for fractional Brownian motions, lecture note, http://www.norbertwiener.umd.edu/Madan/talks/Elliott.pdf
- [25] R. Elliott & J. Hoek(2007), Ito Formulas for Fractional Brownian Motion, Advances in Mathematical Finance Applied and Numerical Harmonic Analysis, 2007, Part I, 59-81.
- [26] M. Escobar & A. Kiechle & L. Seco & R. Zagst(2011), Option on a CPPI, International Mathematical Forum, Vol. 6, 2011, no. 5, 229-262.
- [27] H. Follmer & P. Leukert(1999), Quantile Hedging, Finance and Stochastics, 251-273.
- [28] H. Follmer and M. Schweizer (1991), Hedging of Contingent Claims under Incomplete Information, in: M. H. A. Davis and R. J. Elliott (eds.), Applied

Stochastic Analysis, Stochastics Monographs, Vol. 5, Gordon and Breach, 389-414.

- [29] J. Franke & W. Hardle & C. Hafner(2008), Statistics of Financial Markets, 2nd ed., Springer, 2008.
- [30] H. Gerber, and E. S. W. Shiu (1994), Option Pricing by Esscher Transform, Trans. Soc. Actuaries 46, 99-191.
- [31] Paul Glasserman(2004), Monte Carlo methods engineering in financial, Springer, 2004.
- [32] R.Hogg & A. Craig(1995), Introduction to Mathematical Statistics, 5th Edition, Pearson Education, 1995.
- [33] F. B. Hanson(2007), Applied Stochastic Processes and Control for Jump-Diffusions: Modeling, Analysis, and Computation, Adcances in Design and Control, 2007.
- [34] P. Honore(1998), Pitfalls in Estimating Jump-Diffusion Models, Working paper.
- [35] Y. Hu & B. Ksendal(2003), Fractional white noise calculus and application to finance, Infinite Dimensional Analysis, Quantum Probability and Related Topics Vol. 6, NO. 1 (2003)1-32.
- [36] Y. Huang & S. Chen(2002), Warrants pricing: Stochastic volatility vs. Black-Scholes, Pacific-Basin Finance Journal 10(2002), 393-409.
- [37] F. Hubalek &J. Kallsen, & L. Krawczyk(2006), Variance-optimal hedging for processes with stationary independent increments, The Annals of Applied Probability, 16 (2006), 853-885.

- [38] J. Hull(2000), Options, Futures, & Other Derivatives, 4th ed., Prentice-Hall, Englewood Cliffs, 2000.
- [39] J. Hull(2011), Options, Futures, & Other Derivatives, 8th ed., Prentice-Hall, 2011.
- [40] J. Jacod & A.N. Shiryaev(2003), Limit Theorems for Stochastic Processes, 2nd ed. 2003.
- [41] M.S. Joshi(2008), The Concepts and Practice of Mathematical Finance, 2nd ed., Cambridge, 2008.
- [42] M.S. Joshi(2011), More mathematical finance, Pilot Whale Press, 2011.
- [43] M.S. Joshi(2008), C++ Design Patterns and Derivatives Pricing, Cambridge, 2008.
- [44] J. Kallsen & M. Taqqu(1998), Option Pricing in ARCH-type Models, math finance, Vol.8, No.1, 13-26. W.J.
- [45] I. Karatzas & S.E. Shreve(1991), Brownian Motion and Stochastic Calculus, 2nd ed., Springer, 1991.
- [46] I. Karatzas & S.E. Shreve(1998), Methods of Mathematical Finance, Springer, 1999.
- [47] B. Ksendal & A. Sulem(2004), Applied Stochastic Control of Jump Diffusion, Springer, 2004.
- [48] B. Ksendal(2005), Stochastic Differential Equations: An Introduction with Applications, 6th ed, Springer, 2005.
- [49] N.M. Kiefer(1978) Discrete Parameter Variation: Efficient Estimation of a Switching Regression Model, Econometrica 46, 427-434.

- [50] S. G. Kou(1999), A Jump Diffusion Model for Option Pricing with Three Properties: Leptokurtic Feature, Volatility Smile, and Analytical Tractability, first draft, working paper, 1999.
- [51] S. G. Kou(2002), A Jump-Diffusion Model for option pricing, Management Science(2002) Vol.48, No. 8, 1086-1101.
- [52] S.G. Kou & Hui Wang(2004), Option Pricing Under a Double Exponential Jump Diffusion Model, Management Science, Vol. 50, Sep. 2004, 1178-1192.
- [53] S. G. Kou(2007), Chapter 2 Jump-Diffusion Models for Asset Pricing in Financial Engineering, J.R. Birge and V. Linetsky (Eds.), Handbooks in Operations Research and Management Science, Vol. 15, Introduction to the Handbook of Financial Engineering, 73-116, 2007.
- [54] A. Lewis(2001), a simple option formula for general jump-diffusion and other exponential levy processes, working paper, available from http://optioncity.net/pubs/ExpLevy.pdf, 2001
- [55] A. E. B. Lim(2005), Mean-Variance Hedging When There are Jumps, SIAM J. Control Optim. 44 1893-1922.
- [56] Yuh-Dauh Lyuu(2002), Financial engineering and computation: principles, mathematics, algorithms, Cambridge University Press, 2002.
- [57] K. Matsuda, Introduction to Merton Jump Diffusion Model, working paper.
- [58] R. C. Merton(1976), Option Pricing When Underlying Stock Returns are Discontinuous, Journal of Financial Economics 3(1976) 125-144.
- [59] F. Mkaouar & J. Prigent(2007), CPPI with Stochastic Floors, Working paper.

- [60] H. Pham(2000), On quadratic hedging in continuous time, Math. Methods Oper. Res. 51(2), 315-339, 2000.
- [61] A. Perold, A constant proportion portfolio insurance. Unpublished manuscript, Harvard Business School.
- [62] L. C. G. Rogers & David Williams(2000), Diffusions, Markov Processes, and Martingales: Volume 1, Foundations, 2nd ed., Cambridge, 2000.
- [63] L. C. G. Rogers & David Williams(2000), Diffusions, Markov Processes and Martingales: Volume 2, Ito Calculus, 2nd ed., Cambridge, 2000.
- [64] S. Rostek(2009), Option Pricing in Fractional Brownian Markets, Springer, 2009.
- [65] W. Runggaldier(2003), Jump Diffusion Models. In : Handbook of Heavy Tailed Distributions in Finance (S.T. Rachev, ed.), Handbooks in Finance, Book 1 (W.Ziemba Series Ed.), Elesevier/North-Holland 2003, 169-209.
- [66] P. Sattayatham, A. Intarasit & A. P. Chaiyasena(2007), A fractional Black-Scholes Model with Jumps, Vietnam Journal of Mathematics 35:3(2007) 1-15.
- [67] M. Schweizer(2001), A Guided Tour through Quadratic Hedging Approaches, in: E. Jouini, J. Cvitanic, M. Musiela (eds.), Option Pricing, Interest Rates and Risk Management, Cambridge University Press (2001), 538-574.
- [68] S. E. Shreve(2005), Stochastic Calculus for Finance I: The Binomial Asset Pricing Model, Springer, 2005.
- [69] S. E. Shreve(2004), Stochastic Calculus for Finance II: Continuous-Time Models, Springer, 2004.

- [70] R. H. Shumway & D. S. Stoffer(2006), Time series analysis and its applications with r examples, 2nd edition, Springer, 2006.
- [71] P. Tankov and E. Voltchkvoa(2009), Jump-diffusion models: a practitioner's guide, Banque et March'es, working paper.
- [72] P. Tankov(2010), Pricing and hedging in exponential Levy models: review of recent results, To appear in the Paris-Princeton Lecture Notes in Mathematical Finance, Springer, 2010.
- [73] P. Tankov(2010), Pricing and hedging gap risk, The journal of Computaational Finance, 33-59, Vol 13 NO. 3, Springer, 2010.
- [74] P. Tankov(2010), Financial modeling with levy processes, lecture notes, 2010.
- [75] J. Xia(2006), Mean-variance Hedging in the Discontinuous Case, eprint arXiv:math/0607775, 2006.

#### VITA

I was born on Jan. 14, 1983 in Quzhou, Zhejiang, China. I obtained my B.S. degree in mathematics in June 2006 from the Central South University in Changsha, Hunan, China. In August 2008 I joined the graduate program at the University of Missouri and pursued my Ph.D. degree in mathematics under the supervision of Professor Allanus Tsoi. I graduated in May 2012.